ON MOD p LOCAL-GLOBAL COMPATIBILITY FOR GL_3 IN THE ORDINARY CASE

FLORIAN HERZIG, DANIEL LE, AND STEFANO MORRA

ABSTRACT. Suppose that F/F^+ is a CM extension of number fields in which the prime p splits completely and every other prime is unramified. Fix a place w|p of F. Suppose that $\overline{r} : \operatorname{Gal}(\overline{F}/F) \to \operatorname{GL}_3(\overline{\mathbb{F}}_p)$ is a continuous irreducible Galois representation such that $\overline{r}|_{\operatorname{Gal}(\overline{F}_w/F_w)}$ is upper-triangular, maximally non-split, and generic. If \overline{r} is automorphic, and some suitable technical conditions hold, we show that $\overline{r}|_{\operatorname{Gal}(\overline{F}_w/F_w)}$ can be recovered from the $\operatorname{GL}_3(F_w)$ -action on a space of mod p automorphic forms on a compact unitary group. On the way we prove results about weights in Serre's conjecture for \overline{r} , show the existence of an ordinary lifting of \overline{r} , and prove the freeness of certain Taylor–Wiles patched modules in this context. We also show the existence of many Galois representations \overline{r} to which our main theorem applies.

Contents

1. Introduction	2
1.1. Motivation and statement of main results	2
1.2. Methods used	5
1.3. Acknowledgements	6
1.4. Notation	7
2. The local Galois side	7
2.1. Fontaine–Laffaille invariant	7
2.2. <i>p</i> -adic Hodge theory	10
2.3. Breuil submodules and Galois representations	14
2.4. On filtrations	15
2.5. From strongly divisible modules to Fontaine–Laffaille modules	23
2.6. Elimination of Galois types	28
3. The local automorphic side	34
3.1. Group algebra operators	35
3.2. Jacobi sums and characteristic zero principal series	40
4. Local-global compatibility	43
4.1. The space of automorphic forms on certain unitary groups	43
4.2. Serre weights	43
4.3. Potentially semistable lifts	45
4.4. Serre weights of \overline{r} in the maximally non-split case	46
4.5. Local-global compatibility	49
5. Freeness over the Hecke algebra	52
5.1. The setup	53
5.2. The Taylor–Wiles method	54

The first-named author was partially supported by a Sloan Fellowship and an NSERC grant.

5.3. The Diamond–Fujiwara trick	57
A. Some integral <i>p</i> -adic Hodge theory	60
A.1. On certain categories of torsion modules	60
A.2. Some basic lemmas on Breuil modules	62
A.3. Functors towards Galois representations	63
A.4. On dualities	67
A.5. Proofs of some results in Section 2.2	67
A.6. Proofs of some results in Section 2.3	68
References	70

1. INTRODUCTION

1.1. Motivation and statement of main results. Suppose that p is a prime and that $\overline{\rho} : \operatorname{Gal}(\overline{\mathbb{Q}}_p/\mathbb{Q}_p) \to \operatorname{GL}_n(\overline{\mathbb{F}}_p)$ is a continuous Galois representation. One would hope that there is a mod p Langlands correspondence that associates to $\overline{\rho}$ in a natural way a smooth representation $\Pi(\overline{\rho})$ of $\operatorname{GL}_n(\mathbb{Q}_p)$ over $\overline{\mathbb{F}}_p$ (or maybe a packet of such representations), and similarly for *p*-adic representations. Unfortunately, at this point, this is only known for $n \leq 2$ [Bre03], [Col10]. But suppose now that F/F^+ is a CM extension of number fields in which p splits completely, and fix a place w|p. Even in the absence of a local mod p Langlands correspondence for n > 2, given a global automorphic Galois representation $\overline{r} : \operatorname{Gal}(\overline{F}/F) \to \operatorname{GL}_n(\overline{\mathbb{F}}_p)$ we can define smooth representations $\Pi_{\text{glob}}(\overline{r})$ of $\operatorname{GL}_n(\mathbb{Q}_p)$ over $\overline{\mathbb{F}}_p$ on spaces of mod p automorphic forms on a definite unitary group, that serve as candidates for $\Pi(\overline{r}|_{\text{Gal}(\overline{F}_w/F_w)})$ (in the spirit of Emerton's local-global compatibility [Eme]; see also $[CEG^+16]$ in the *p*-adic setting). It is not known whether $\Pi_{glob}(\bar{r})$ depends only on $\overline{r}|_{\text{Gal}(\overline{F}_w/F_w)}$. The motivating question of this paper is opposite to this: do the candidate representations $\Pi_{\rm glob}(\bar{r})$ contain at least as much information as $\overline{r}|_{\text{Gal}(\overline{F}_w/F_w)}$? We answer this question in the affirmative in many cases when n = 3.

We fix a finite extension E/\mathbb{Q}_p with residue field \mathbb{F} , and consider absolutely irreducible Galois representations \overline{r} : $\operatorname{Gal}(\overline{F}/F) \to \operatorname{GL}_3(\mathbb{F})$. We assume moreover that $\overline{r}|_{\operatorname{Gal}(\overline{F}_w/F_w)}$ is upper-triangular, maximally non-split, and generic. This means that

(1.1.1)
$$\overline{r}|_{\text{Gal}(\overline{F}_w/F_w)} \sim \begin{pmatrix} \omega^{a+1} \mathrm{nr}_{\mu_2} & *_1 & * \\ & \omega^{b+1} \mathrm{nr}_{\mu_1} & *_2 \\ & & \omega^{c+1} \mathrm{nr}_{\mu_0} \end{pmatrix},$$

the extensions $*_1$, $*_2$ are non-split, and a-b > 2, b-c > 2, a-c < p-3. (Here, ω is the mod p cyclotomic character and nr_{μ} denotes the unramified character taking value $\mu \in \mathbb{F}^{\times}$ on geometric Frobenius elements.) It is not hard to see that once the diagonal characters are fixed, the isomorphism class of $\overline{r}|_{\operatorname{Gal}(\overline{F}_w/F_w)}$ is determined by an invariant $\operatorname{FL}(\overline{r}|_{\operatorname{Gal}(\overline{F}_w/F_w)})$ that can take any value in $\mathbb{P}^1(\mathbb{F}) \setminus \{\mu_1\}$. (We normalize this invariant using Fontaine–Laffaille theory, see Definition 2.1.10.)

To explain our results, we briefly describe our global setup, referring to Section 4 for details. Fix a unitary group $G_{/F^+}$ such that $G \times F \cong \operatorname{GL}_3$ and $G(F_v^+) \cong U_3(\mathbb{R})$ for all $v \mid \infty$. Choose a model $\mathcal{G}_{/\mathcal{O}_{F^+}}$ of G such that $\mathcal{G} \times \mathcal{O}_{F_{v'}^+}$ is reductive for all places v' of F^+ that split in F. Let $v = w \mid_{F^+}$. Choose a compact open subgroup $U = U^v \times \mathcal{G}(\mathcal{O}_{F_v^+}) \leq G(\mathbb{A}_{F^+}^\infty)$ that is sufficiently small, and unramified at all places

 $\mathbf{2}$

dividing p. Let V' denote an irreducible smooth representation of $\prod_{v'|p,v'\neq v} \mathcal{G}(\mathcal{O}_{F_{v'}^+})$ over \mathbb{F} determined by a highest weight in the lowest alcove, and let \widetilde{V}' denote a Weyl module of $\prod_{v'|p,v'\neq v} \mathcal{G}(\mathcal{O}_{F_{v'}^+})$ over \mathcal{O}_E of the same highest weight, so $\widetilde{V}' \otimes_{\mathcal{O}_E} \mathbb{F} \cong V'$. (Except for Theorem C below, the reader may assume for simplicity that $V' = \mathbb{F}$ and $\widetilde{V}' = \mathcal{O}_E$.) We can then define in the usual way spaces of mod p automorphic forms $S(U^v, V') = \varinjlim_{U_v \leq G(F_v^+)} S(U^v U_v, V')$ and similarly $S(U^v, \widetilde{V}')$ that are smooth representations of $G(F_v^+) \cong \operatorname{GL}_3(\mathbb{Q}_p)$ (where this isomorphism depends on our chosen place w|v).

We fix a cofinite subset \mathcal{P} of places $w' \nmid p$ of F that split over F^+ , such that U is unramified at $w'|_{F^+}$, and such that \overline{r} is unramified at w'. Then the abstract Hecke algebra $\mathbb{T}^{\mathcal{P}}$ generated over \mathcal{O}_E by Hecke operators at all places in \mathcal{P} acts on $S(U^v, V')$ and $S(U^v, \widetilde{V}')$, commuting with the $\mathrm{GL}_3(\mathbb{Q}_p)$ -actions. Moreover, \overline{r} determines a maximal ideal $\mathfrak{m}_{\overline{r}}$ of $\mathbb{T}^{\mathcal{P}}$. We assume that \overline{r} is automorphic in this setup, which means that $S(U^v, V')[\mathfrak{m}_{\overline{r}}] \neq 0$ (or equivalently, $S(U^v, V')_{\mathfrak{m}_{\overline{r}}} \neq 0$). These $\mathrm{GL}_3(\mathbb{Q}_p)$ -representations, $S(U^v, V')[\mathfrak{m}_{\overline{r}}]$, are the natural candidates $\Pi_{\mathrm{glob}}(\overline{r})$ that we mentioned above, at least if the level U^v is chosen optimally.

It is a consequence of earlier work of the first-named author and C. Breuil [BH15] that the $\operatorname{GL}_3(\mathbb{Q}_p)$ -representation $S(U^v, V')[\mathfrak{m}_{\overline{r}}]$ determines the ordered triple of diagonal characters $(\omega^{a+1}\mathrm{nr}_{\mu_2}, \omega^{b+1}\mathrm{nr}_{\mu_1}, \omega^{c+1}\mathrm{nr}_{\mu_0})$ of $\overline{r}|_{\operatorname{Gal}(\overline{F}_w/F_w)}$. In fact, the triple (a, b, c) can be recovered from the (ordinary part of the) $\operatorname{GL}_3(\mathbb{Z}_p)$ -socle – i.e., the Serre weights of \overline{r} – by [GG12] and the $\mu_i \in \mathbb{F}^{\times}$ are determined by the Hecke action at p on the $\operatorname{GL}_3(\mathbb{Z}_p)$ -socle. It therefore remains for us to show that $S(U^v, V')[\mathfrak{m}_{\overline{r}}]$ determines the invariant $\operatorname{FL}(\overline{r}|_{\operatorname{Gal}(\overline{F}_w/F_w)})$. (We note that the representation $\Pi(\overline{r}|_{\operatorname{Gal}(\overline{F}_w/F_w)})^{\operatorname{ord}}$ of [BH15] does not contain this information.)

Let I denote the Iwahori subgroup of $\operatorname{GL}_3(\mathbb{Z}_p)$, which is the preimage of the upper-triangular matrices $B(\mathbb{F}_p)$ in $\operatorname{GL}_3(\mathbb{F}_p)$. If V is a representation of $\operatorname{GL}_3(\mathbb{Z}_p)$ over \mathcal{O}_E and $a_i \in \mathbb{Z}$ we write

$$V^{I,(a_2,a_1,a_0)} \stackrel{\text{def}}{=} \operatorname{Hom}_I(\mathcal{O}_E(\widetilde{\omega}^{a_2} \otimes \widetilde{\omega}^{a_1} \otimes \widetilde{\omega}^{a_0}), V),$$

where the character in the domain denotes the inflation to I of the homomorphism $B(\mathbb{F}_p) \to \mathcal{O}_E^{\times}$, $\begin{pmatrix} x & y & z \\ u & w \end{pmatrix} \mapsto \widetilde{x}^{a_2}\widetilde{u}^{a_1}\widetilde{w}^{a_0}$. If V is even a representation of $\operatorname{GL}_3(\mathbb{Q}_p)$, then $V^{I,(a_2,a_1,a_0)}$ affords an action of U_p -operators U_1, U_2 (see (3.1.10)). Define also $\Pi \stackrel{\text{def}}{=} \begin{pmatrix} 1 & 1 \\ p & 1 \end{pmatrix}$, which sends $V^{I,(a_2,a_1,a_0)}$ to $V^{I,(a_1,a_0,a_2)}$.

Finally, and crucially, we define explicit group algebra operators $S, S' \in \mathbb{F}[\operatorname{GL}_3(\mathbb{F}_p)]$, see (4.5.1). We can now state our first main theorem.

Theorem A (Thm. 4.5.2). We make the following additional assumptions:

- (i) $\operatorname{FL}(\overline{r}|_{\operatorname{Gal}(\overline{F}_w/F_w)}) \notin \{0,\infty\}.$
- (ii) The \mathcal{O}_E -dual of $S(U^v, \widetilde{V}')_{\mathfrak{m}_{\overline{r}}}^{I,(-b,-c,-a)}$ is free over \mathbb{T} , where \mathbb{T} denotes the \mathcal{O}_E -subalgebra of End $\left(S(U^v, \widetilde{V}')_{\mathfrak{m}_{\overline{r}}}^{I,(-b,-c,-a)}\right)$ generated by $\mathbb{T}^{\mathcal{P}}$, U_1 , and U_2 .

Then we have the equality

$$S' \circ \Pi = (-1)^{a-b} \cdot \frac{b-c}{a-b} \cdot \operatorname{FL}(\overline{r}|_{\operatorname{Gal}(\overline{F}_w/F_w)}) \cdot S$$

of maps

$$S(U^{v}, V')[\mathfrak{m}_{\overline{r}}]^{I, (-b, -c, -a)}[U_{1}, U_{2}] \to S(U^{v}, V')[\mathfrak{m}_{\overline{r}}]^{I, (-c-1, -b, -a+1)}.$$

Moreover, these maps are injective with non-zero domain. In particular, $\operatorname{FL}(\overline{r}|_{\operatorname{Gal}(\overline{F}_w/F_w)})$ is determined by the smooth $\operatorname{GL}_3(\mathbb{Q}_p)$ -representation $S(U^v, V')[\mathfrak{m}_{\overline{r}}]$.

The first assumption is related to the surprising fact that the $\operatorname{GL}_3(\mathbb{Z}_p)$ -socle of $S(U^v, V')[\mathfrak{m}_{\overline{r}}]$ changes for the exceptional two invariants, see Theorem D below. (Incidentally, this means that even in the exceptional cases $\operatorname{FL}(\overline{r}|_{\operatorname{Gal}(\overline{F}_w/F_w)})$ is determined by $S(U^v, V')[\mathfrak{m}_{\overline{r}}]$.)

We show that the second assumption is often a consequence of the first assumption. The second assumption is an analogue of Mazur's "mod p multiplicity one" result, and thus our result may be of independent interest. We have not tried to optimize our hypotheses, the most stringent of which is that U may be taken to be unramified at all finite places.

Theorem B (Thm. 5.1.1). Assume hypotheses (i)–(ix) in Section 5.1. If we have $\operatorname{FL}(\overline{r}|_{\operatorname{Gal}(\overline{F}_w/F_w)}) \neq \infty$, then assumption (ii) in Theorem A holds.

In fact we show that for any value of $FL(\overline{r}|_{Gal}(\overline{F}_w/F_w))$ either the second assumption or its dual holds (see Remark 5.3.6).

We also show, using results of [EG14], that for any given local Galois representation as in (1.1.1) we can construct a globalization to which Theorem B applies.

Theorem C (Thm. 5.3.7). Suppose that $\overline{\rho}$: $\operatorname{Gal}(\overline{\mathbb{Q}}_p/\mathbb{Q}_p) \to \operatorname{GL}_3(\mathbb{F})$ is uppertriangular, maximally non-split, and generic. Then, after possibly replacing \mathbb{F} by a finite extension, there exist a CM field F, a Galois representation \overline{r} : $\operatorname{Gal}(\overline{F}/F) \to$ $\operatorname{GL}_3(\mathbb{F})$, a place w|p of F, groups $G_{/F^+}$ and $\operatorname{G}_{/\mathbb{O}_{F^+}}$, and a compact open subgroup U^v (where $v = w|_{F^+}$) satisfying all hypotheses of the setup in Section 5.1 such that $\overline{r}|_{\operatorname{Gal}(\overline{F}_w/F_w)} \cong \overline{\rho}$. In particular, if $\operatorname{FL}(\overline{\rho}) \notin \{0, \infty\}$, Theorem A applies to \overline{r} .

As a by-product of our methods we almost completely determine the set of Serre weights of \overline{r} . Here, the set $W_w(\overline{r})$ is defined to be the set of irreducible $\operatorname{GL}_3(\mathbb{Z}_p)$ -representations whose duals occur in the $\operatorname{GL}_3(\mathbb{Z}_p)$ -socle of $S(U^v, V')_{\mathfrak{m}_{\overline{r}}}$ (for some U^v and \mathcal{P} as above). See Section 4.2 for our notation for Serre weights.

Theorem D (Thm. 4.4.1). Keep the assumptions on \overline{r} that precede Theorem A above. (i) If $\operatorname{FL}(\overline{r}|_{C_{2}(\overline{r}_{-1}(\overline{r}_{-1})}) \notin \{0,\infty\}$ we have

1) If
$$\operatorname{FL}(r|_{\operatorname{Gal}(\overline{F}_w/F_w)}) \notin \{0, \infty\}$$
 we have
 $\{F(a-1, b, c+1)\} \subseteq W_w(\overline{r}) \subseteq$
 $\subseteq \{F(a-1, b, c+1), F(c+p-1, b, a-p+1)\}.$

(ii) If $\operatorname{FL}(\overline{r}|_{\operatorname{Gal}(\overline{F}_m/F_m)}) = \infty$ we have

$$\{F(a-1,b,c+1), F(a,c,b-p+1)\} \subseteq W_w(\bar{r}) \subseteq \subseteq \{F(a-1,b,c+1), F(c+p-1,b,a-p+1), F(a,c,b-p+1)\}.$$

(iii) If $\operatorname{FL}(\overline{r}|_{\operatorname{Gal}(\overline{F}_w/F_w)}) = 0$ we have $\{F(a-1,b,c+1), F(b+p-1,a,c)\} \subseteq W_w(\overline{r}) \subseteq$ $\subseteq \{F(a-1,b,c+1), F(c+p-1,b,a-p+1), F(b+p-1,a,c)\}.$ This is one of the first Serre weight results in dimension 3. It was completed in early 2014, before the recent progress of [LLHLM] (on Serre weights in dimension 3 in the generic semisimple case, using different methods). As we said above, the dependence on $FL(\bar{r}|_{Gal}(\bar{F}_w/F_w))$ in this theorem was unexpected, as there were no explicit Serre weight conjectures in the literature that apply to non-semisimple $\bar{r}|_{Gal}(\bar{F}_w/F_w)$.

As a consequence of this theorem we also show the existence of an automorphic lift r of \overline{r} such that $r|_{\text{Gal}(\overline{F}_w/F_w)}$ is upper-triangular. It is in the same spirit as the main results of [BLGG12] (which concerned two-dimensional representations).

Theorem E (Cor. 4.4.4). In the setting of Theorem D, \overline{r} has an automorphic lift $r : \operatorname{Gal}(\overline{F}/F) \to \operatorname{GL}_3(\mathcal{O}_E)$ (after possibly enlarging E) such that $r|_{\operatorname{Gal}(\overline{F}_w/F_w)}$ is crystalline and ordinary of Hodge–Tate weights $\{-a-1, -b-1, -c-1\}$.

1.2. Methods used. Theorems A and B generalize earlier work of Breuil–Diamond [BD14] which treated two-dimensional Galois representations of $\text{Gal}(\overline{F}/F)$, where F is totally real and p is unramified in F. We follow the same general strategy: we lift the Hecke eigenvalues of \overline{r} to a well-chosen type in characteristic zero, use classical local-global compatibility at p, and then study carefully how both the Galois-side and the GL₃-side reduce modulo p. However, it is significantly more difficult to carry out this strategy in the GL₃-setting.

We first prove the upper bound in Theorem D by lifting to various types in characteristic zero and using integral *p*-adic Hodge theory to reduce modulo *p*. This is more involved in dimension 3, since we are no longer in the potentially Barsotti–Tate setting. We crucially use results of Caruso to filter our Breuil module (corresponding to $\overline{r}|_{\text{Gal}(\overline{F}_w/F_w)}$) according to the socle filtration on $\overline{r}|_{\text{Gal}(\overline{F}_w/F_w)}$, see Proposition 2.3.5.

The following theorem is our key local result on the Galois-side. Our chosen type is a tame principal series that contains in its reduction mod p all elements of $W_w(\bar{r})$ (unlike in [BD14], where the intersection always consisted of one element). In contrast to [BD14] we get away with a rough classification of the strongly divisible module corresponding to ρ . (We do not need to determine Frobenius and monodromy operators.) We also note that the relevant information on the Galoisside is independent of the Hodge filtration, so that we can transfer this information to the GL₃-side using classical local-global compatibility.

Theorem F (Thm. 2.5.1). Let ρ : $\operatorname{Gal}(\overline{\mathbb{Q}}_p/\mathbb{Q}_p) \to \operatorname{GL}_3(\mathcal{O}_E)$ be a potentially semistable p-adic Galois representation of Hodge–Tate weights $\{-2, -1, 0\}$ and inertial type $\widetilde{\omega}^a \oplus \widetilde{\omega}^b \oplus \widetilde{\omega}^c$. Assume that the residual representation $\overline{\rho}$: $\operatorname{Gal}(\overline{\mathbb{Q}}_p/\mathbb{Q}_p) \to$ $\operatorname{GL}_3(\mathbb{F})$ is upper-triangular, maximally non-split, and generic as in (1.1.1). Let $\lambda \in \mathcal{O}_E$ be the Frobenius eigenvalue on $\operatorname{D}_{\operatorname{st}}^{\mathbb{Q}_p,2}(\rho)^{I_{\mathbb{Q}_p}=\widetilde{\omega}^b}$. Then the Fontaine–Laffaille invariant of $\overline{\rho}$ is given by:

$$\operatorname{FL}(\overline{\rho}) = \operatorname{red}\left(p\lambda^{-1}\right),$$

where red denotes the specialization map $\mathbb{P}^1(\mathcal{O}_E) \to \mathbb{P}^1(\mathbb{F})$.

On the GL₃-side our main innovation consists of the explicit group algebra operators S, S'. The analogues of these operators for GL₂ show up in various contexts (see, for example, [Pas07], Lemma 4.1, [BP12], Lemma 2.7, [Bre11], §4, and [BD14], Proposition 2.6.1). Our proof is specific to GL₃. It would be interesting to find a more conceptual explanation for them. See also Question 3.1.3 for a further discussion of such operators.

Proposition G (Prop. 3.1.2).

(i) There is a unique non-split extension of irreducible $GL_3(\mathbb{F}_p)$ -representations

$$0 \to F(-c-1, -b, -a+1) \to V \to F(-b+p-1, -c, -a) \to 0$$

and S induces an isomorphism $S: V^{I,(-b,-c,-a)} \xrightarrow{\sim} V^{I,(-c-1,-b,-a+1)}$ of one-dimensional vector spaces.

(ii) There is a unique non-split extension of irreducible $GL_3(\mathbb{F}_p)$ -representations

$$\begin{array}{l} 0 \rightarrow F(-c-1,-b,-a+1) \rightarrow V \rightarrow F(-c,-a,-b-p+1) \rightarrow 0 \\ and \ S' \ induces \ an \ isomorphism \ S' \ : \ V^{I,(-c,-a,-b)} \xrightarrow{\sim} V^{I,(-c-1,-b,-a+1)} \\ of \ one-dimensional \ vector \ spaces. \end{array}$$

The reduction mod p result on the GL₃-side is comparatively easier, see Proposition 3.2.2.

By combining the above results we deduce Theorem A. We note that assumption (ii) is needed for lifting elements of $S(U^v, V')[\mathfrak{m}_{\overline{r}}]^{I,(-b,-c,-a)}[U_1, U_2]$ to suitable Iwahori eigenvectors in characteristic zero. The U_i -operators allow us to deal with the possible presence of the shadow weight F(c+p-1,b,a-p+1) in Theorem D. (The term "shadow weight" is defined in [EGH13], §6 and more generally in [GHS], §1.5 and §7.2.) Namely, if $v \in S(U^v, V')[\mathfrak{m}_{\overline{r}}]^{I,(-b,-c,-a)}[U_1, U_2]$ is non-zero, we show that it generates the representation of Proposition G(i) under the $GL_3(\mathbb{Z}_p)$ -action. Similar comments apply to Πv . Proposition G then allows us to deduce that the maps in Theorem A are well-defined and injective. (We refer to Remark 4.5.8 for variations on assumption (ii). The stronger assumptions appearing in Remark 4.5.9 are analogous to the multiplicity one conditions appearing in [BD14].)

Interestingly, the argument proving Theorem A also lets us deduce the hardest part of Theorem D, namely the existence of the shadow weights F(a, c, b - p + 1), F(b + p - 1, a, c) in the two exceptional cases. After [EGH13], this is only the second result in the literature proving the existence of shadow weights. (Again this precedes [LLHLM].)

Finally, we establish Theorem B. As in [BD14] our method relies on the Taylor–Wiles method. However, as we do not know whether our local deformation ring at p is formally smooth (which in any case should be false if our chosen type intersects $W_w(\bar{r})$ in more than one element) we cannot directly apply Diamond's method [Dia97]. Instead we use the patched modules of [CEG⁺¹⁶] that live over the universal deformation space at p and use ideas of [EGS15] and [Le]. See Theorem 5.2.3 for our freeness result at infinite level from which we deduce Theorem B. Similarly to above, we add U_p -operators in order to deal with the possible presence of the shadow weight F(c + p - 1, b, a - p + 1).

1.3. Acknowledgements. The debt that this paper owes to the work of Breuil– Diamond [BD14] should be obvious to the reader. We thank Christophe Breuil for a number of very helpful conversations as well as for his encouragement. We thank Guy Henniart for providing a reference for the proof of Lemma 4.3.3, Xavier Caruso for helpful correspondence regarding Appendix A, and John Enns for helpful remarks on an earlier version of this paper. Finally we are thankful to the referee for useful comments that in particular helped us to improve the exposition. 1.4. Notation. Let $\overline{\mathbb{Q}}$ be an algebraic closure of \mathbb{Q} . All number fields F/\mathbb{Q} will be considered as subfields of $\overline{\mathbb{Q}}$ and we write $G_F \stackrel{\text{def}}{=} \operatorname{Gal}(\overline{\mathbb{Q}}/F)$ to denote the absolute Galois group of F. For any rational prime $\ell \in \mathbb{Q}$, we fix algebraic closures $\overline{\mathbb{Q}}_{\ell}$ of \mathbb{Q}_{ℓ} and embeddings $\overline{\mathbb{Q}} \hookrightarrow \overline{\mathbb{Q}}_{\ell}$ (hence inclusions $G_{\mathbb{Q}_{\ell}} \hookrightarrow G_{\mathbb{Q}}$). The residue field of $\overline{\mathbb{Q}}_{\ell}$, which is an algebraic closure of \mathbb{F}_{ℓ} , is denoted by $\overline{\mathbb{F}}_{\ell}$. As above, all algebraic extensions of \mathbb{Q}_{ℓ} , \mathbb{F}_{ℓ} will be considered as subfields of the fixed algebraic closures $\overline{\mathbb{Q}}_{\ell}$, $\overline{\mathbb{F}}_{\ell}$.

Let k/\mathbb{F}_p be a finite extension of degree $f \geq 1$, and let $K_0 \stackrel{\text{def}}{=} W(k)[\frac{1}{p}]$ be the unramified extension of \mathbb{Q}_p of degree f. Suppose that $e \geq 1$ is any divisor of $p^f - 1$. (Starting in Section 2.4 we will assume $e = p^f - 1$. However, in the appendix it will be convenient to allow more general e, in particular e = 1.) We consider the Eisenstein polynomial $E(u) \stackrel{\text{def}}{=} u^e + p \in \mathbb{Q}_p[u]$ and fix a root $\varpi = \sqrt[e]{-p} \in \overline{\mathbb{Q}}_p$. Let $K \stackrel{\text{def}}{=} K_0(\varpi)$, a tamely totally ramified cyclic extension of K_0 of degree e with uniformizer ϖ .

Let *E* be a finite extension of \mathbb{Q}_p . We write \mathcal{O}_E for its ring of integers, \mathbb{F} for its residue field and $\varpi_E \in \mathcal{O}_E$ to denote a uniformizer. We always assume that $K \subseteq E$.

The choice of $\varpi \in K$ provides us with a homomorphism

$$\widetilde{\omega}_{\varpi} : \operatorname{Gal}(K/\mathbb{Q}_p) \longrightarrow W(k)^{\times}$$
$$g \longmapsto \frac{g(\varpi)}{\varpi}$$

whose reduction mod p will be denoted by ω_{ϖ} . Note that the inclusion $k \subseteq \mathbb{F}$ induced by $K \subseteq E$ provides us with a niveau f fundamental character ω_f : $\operatorname{Gal}(K/K_0) \to \mathbb{F}^{\times}$, namely $\omega_f = \omega_{\varpi}|_{\operatorname{Gal}(K/K_0)}$.

We denote by $\omega: G_{\mathbb{Q}_p} \to \mathbb{F}_p^{\times}$ the mod p cyclotomic character, so $\omega = \omega_1$.

Write φ for the *p*-power Frobenius on *k*. We recall the standard idempotent elements $\epsilon_{\sigma} \in k \otimes_{\mathbb{F}_p} \mathbb{F}$ defined for $\sigma \in \text{Hom}(k,\mathbb{F})$, which verify $(\varphi \otimes 1)(\epsilon_{\sigma}) = \epsilon_{\sigma \circ \varphi^{-1}}$ and $(\lambda \otimes 1)\epsilon_{\sigma} = (1 \otimes \sigma(\lambda))\epsilon_{\sigma}$. We write $\hat{\epsilon}_{\sigma} \in W(k) \otimes_{\mathbb{Z}_p} \mathfrak{O}_E$ for the standard idempotent elements; they reduce to ϵ_{σ} modulo ϖ_E .

Our convention on Hodge–Tate weights is that the cyclotomic character ε : $G_{\mathbb{Q}_p} \to \mathbb{Q}_p^{\times}$ has Hodge–Tate weight -1.

Given a potentially semistable *p*-adic representation $\rho : G_{\mathbb{Q}_p} \to \mathrm{GL}_n(E)$, we write WD(ρ) to denote the associated Weil–Deligne representation as defined in [CDT99], Appendix B.1. Therefore, $\rho \mapsto \mathrm{WD}(\rho)$ is a *covariant* functor. We refer to WD(ρ)|_{$I_{\mathbb{Q}_p}$} as the *inertial type* associated to ρ .

2. The local Galois side

In this section we analyze the local Galois side. In particular, we establish Theorem F of the introduction.

2.1. Fontaine-Laffaille invariant. Let $\overline{\rho} : G_{\mathbb{Q}_p} \to \mathrm{GL}_3(\mathbb{F})$ be a continuous Galois representation. We assume that $\overline{\rho}$ is maximally non-split meaning that $\overline{\rho}$ is uniserial and the graded pieces in its socle filtration are at most one dimensional over \mathbb{F} . (Recall that a finite length module is uniserial if it has a unique composition series.)

In other words,

(2.1.1)
$$\overline{\rho} \sim \begin{pmatrix} \omega^{a_2+1} \mathrm{nr}_{\mu_2} & *_1 & * \\ & \omega^{a_1+1} \mathrm{nr}_{\mu_1} & *_2 \\ & & \omega^{a_0+1} \mathrm{nr}_{\mu_0} \end{pmatrix}$$

for some $a_i \in \mathbb{Z}$, $\mu_i \in \mathbb{F}^{\times}$ and where $*_1, *_2$ are non-split. Here, $\operatorname{nr}_{\mu} : G_{\mathbb{Q}_p} \to \mathbb{F}^{\times}$ denotes the unramified character taking the value μ on a geometric Frobenius element of $G_{\mathbb{Q}_p}$.

2.1.1. **Preliminaries on Fontaine–Laffaille theory**. We briefly recall the theory of Fontaine–Laffaille modules with \mathbb{F} -coefficients and its relation with mod-p Galois representations.

- A Fontaine-Laffaille module $(M, \operatorname{Fil}^{\bullet} M, \phi_{\bullet})$ over $k \otimes_{\mathbb{F}_p} \mathbb{F}$ is the datum of
 - (i) a finite free $k \otimes_{\mathbb{F}_p} \mathbb{F}$ -module M;
- (ii) a decreasing filtration $\{\operatorname{Fil}^{j} M\}_{j \in \mathbb{Z}}$ on M by $k \otimes_{\mathbb{F}_{p}} \mathbb{F}$ -submodules such that $\operatorname{Fil}^{0} M = M$ and $\operatorname{Fil}^{p-1} M = 0$;

(iii) a φ -semilinear isomorphism $\phi_{\bullet} : \operatorname{gr}^{\bullet} M \to M$.

Defining the morphisms in the obvious way, we obtain the abelian category \mathbb{F} - $\mathcal{FL}^{[0,p-2]}$ of Fontaine–Laffaille modules over $k \otimes_{\mathbb{F}_p} \mathbb{F}$. Given a Fontaine–Laffaille module M and $\sigma \in \text{Hom}(k, \mathbb{F})$, we define the Hodge–Tate weights of M with respect to σ :

$$\operatorname{HT}_{\sigma}(M) \stackrel{\text{def}}{=} \left\{ i \in \mathbb{N} : \dim_{\mathbb{F}} \left(\frac{\epsilon_{\sigma} \operatorname{Fil}^{i} M}{\epsilon_{\sigma} \operatorname{Fil}^{i+1} M} \right) \neq 0 \right\}.$$

In the remainder of this paper we focus on Fontaine–Laffaille modules with *parallel* Hodge–Tate weights, i.e. we assume that for all $i \in \mathbb{N}$, the submodules $\operatorname{Fil}^{i} M$ are free over $k \otimes_{\mathbb{F}_{n}} \mathbb{F}$.

Definition 2.1.2. Let M be a Fontaine–Laffaille module with parallel Hodge– Tate weights. A $k \otimes_{\mathbb{F}_p} \mathbb{F}$ -basis $\underline{e} = (e_0, \ldots, e_{n-1})$ on M is compatible with the Hodge filtration if for all $i \in \mathbb{N}$ there exists $j_i \in \mathbb{N}$ such that $\operatorname{Fil}^i M = \sum_{j=j_i}^n (k \otimes_{\mathbb{F}_p} \mathbb{F}) \cdot e_j$.

Note that if the graded pieces of the Hodge filtration have rank at most one, then any two compatible bases on M are related by a lower triangular matrix in $\operatorname{GL}_n(k \otimes_{\mathbb{F}_p} \mathbb{F}).$

Given a Fontaine–Laffaille module and a compatible basis \underline{e} , it is convenient to describe the Frobenius action via a matrix $\operatorname{Mat}_{\underline{e}}(\phi_{\bullet}) \in \operatorname{GL}_n(k \otimes_{\mathbb{F}_p} \mathbb{F})$, defined in the obvious way using the principal symbols $(\operatorname{gr}(e_0), \ldots, \operatorname{gr}(e_{n-1}))$ as a basis of $\operatorname{gr}^{\bullet} M$.

Theorem 2.1.3. There is an exact, fully faithful contravariant functor

$$\Gamma^*_{\operatorname{cris}}: \mathbb{F}\operatorname{-}\mathcal{FL}^{[0,p-2]} \to \operatorname{Rep}_{\mathbb{F}}(G_{K_0})$$

which is moreover compatible with base change: if K'_0/K_0 is finite unramified, with residue field k'/k, then

$$T^*_{cris}(M \otimes_k k') \cong T^*_{cris}(M)|_{G_{K'_*}}$$

Also, the essential image of T^*_{cris} is closed under subquotients.

Proof. The statement with \mathbb{F}_p -coefficients is in [FL82], Théorème 6.1; its analogue with \mathbb{F} -coefficient is a formal argument which is left to the reader (cf. also [GL14], Theorem 2.2.1).

Lemma 2.1.4. Let $\overline{\rho} : G_{\mathbb{Q}_p} \to \operatorname{GL}_3(\mathbb{F})$ be as in (2.1.1). If the integers a_i verify $a_1 - a_0 > 1$, $a_2 - a_1 > 1$ and $p - 2 > a_2 - a_0$ then $\overline{\rho} \otimes \omega^{-a_0}$ is Fontaine–Laffaille, i.e. it is in the essential image of $T^*_{\operatorname{cris}}$.

Proof. This follows, for example, from [GG12], Lemma 3.1.5.

In order to obtain the main results on Serre weights (§4.4) and local-global compatibility (§4.5), we must assume a stronger genericity condition on the integers a_i .

Definition 2.1.5. We say that a maximally non-split Galois representation $\overline{\rho}$: $G_{\mathbb{Q}_p} \to \mathrm{GL}_3(\mathbb{F})$ as in (2.1.1) is *generic* if the triple (a_2, a_1, a_0) satisfies the condition

$$(2.1.6) a_1 - a_0 > 2, \ a_2 - a_1 > 2, \ p - 3 > a_2 - a_0.$$

2.1.2. The Fontaine-Laffaille invariant. Let $\overline{\rho} : G_{\mathbb{Q}_p} \to \mathrm{GL}_3(\mathbb{F})$ be as in Definition 2.1.5. By Lemma 2.1.4 there is a Fontaine-Laffaille module M with Hodge-Tate weights $\{1, a_1 - a_0 + 1, a_2 - a_0 + 1\}$ such that $\mathrm{T}^*_{\mathrm{cris}}(M) \cong \overline{\rho} \otimes \omega^{-a_0}$ and which is moreover endowed with a filtration by Fontaine-Laffaille submodules $0 \subsetneq M_0 \subsetneq M_1 \subsetneq M_2 = M$ induced via $\mathrm{T}^*_{\mathrm{cris}}$ from the cosocle filtration on $\overline{\rho}$ (cf. Theorem 2.1.3).

Lemma 2.1.7. Let $M \in \mathbb{F}$ - $\mathcal{FL}^{[0,p-2]}$ be such that $T^*_{cris}(M) \cong \overline{\rho} \otimes \omega^{-a_0}$. There exists a basis $\underline{e} = (e_0, e_1, e_2)$ of M such that

$$M_i \cap \operatorname{Fil}^{a_i - a_0 + 1} M = \mathbb{F} \cdot e_i$$

for all $i \in \{0, 1, 2\}$.

Proof. This follows by noting that $M_i \cap \operatorname{Fil}^{a_i - a_0 + 1} M = \operatorname{Fil}^{a_i - a_0 + 1} M_i$ and that M_i has Hodge–Tate weights $a_j - a_0 + 1$ with $0 \le j \le i$.

Note that a basis \underline{e} as in Lemma 2.1.7 is compatible with the Hodge filtration. From Lemma 2.1.7 we deduce a useful observation on the Frobenius action on M.

Corollary 2.1.8. Let $\overline{\rho}$, M, \underline{e} be as in Lemma 2.1.7. Then

$$\operatorname{Mat}_{\underline{e}}(\phi_{\bullet}) = \begin{pmatrix} \mu_0 & \alpha_{01} & \alpha_{02} \\ & \mu_1 & \alpha_{12} \\ & & \mu_2 \end{pmatrix} \in \operatorname{GL}_3(\mathbb{F}),$$

where moreover $\alpha_{01}, \alpha_{12} \in \mathbb{F}^{\times}$.

Proof. This follows from the lemma, by recalling that the Fontaine–Laffaille module associated to a $G_{\mathbb{Q}_p}$ -character $\omega^r \operatorname{nr}_{\mu}$ has Hodge–Tate weight r and $\phi = \mu$. Note that $\alpha_{01}, \alpha_{12} \neq 0$ as $\overline{\rho}$ is maximally non-split.

Conversely, we also note that any such matrix defines a Fontaine–Laffaille module whose associated Galois representation is maximally non-split as in 2.1.1.

The Fontaine–Laffaille invariant $FL(\overline{\rho})$ associated to $\overline{\rho}$ is defined in terms of $Mat_e(\phi_{\bullet})$.

Lemma 2.1.9. Keep the hypotheses of Corollary 2.1.8. The element $\frac{\alpha_{02}}{\alpha_{01}\alpha_{12}}$ deduced from Mat_e(ϕ_{\bullet}) does not depend on the choice of the basis <u>e</u>.

Proof. By Lemma 2.1.7 any other such basis is of the form $\beta_i e_i$ for $\beta_i \in \mathbb{F}^{\times}$. The lemma follows.

Definition 2.1.10. Let $\overline{\rho}: G_{\mathbb{Q}_p} \to \mathrm{GL}_3(\mathbb{F})$ be maximally non-split and generic as in Definition 2.1.5. Let M be the Fontaine–Laffaille module associated to $\overline{\rho} \otimes \omega^{-a_0}$, $\underline{e} = (e_0, e_1, e_2)$ a basis of M as in Lemma 2.1.7 and let

$$\operatorname{Mat}_{\underline{e}}(\phi_{\bullet}) = \begin{pmatrix} \mu_0 & \alpha_{01} & \alpha_{02} \\ & \mu_1 & \alpha_{12} \\ & & \mu_2 \end{pmatrix}$$

be the matrix of the Frobenius action on M.

The Fontaine–Laffaille invariant associated to $\overline{\rho}$ is defined as

$$\operatorname{FL}(\overline{\rho}) \stackrel{\text{def}}{=} \frac{\operatorname{det} \begin{pmatrix} \alpha_{01} & \alpha_{02} \\ \mu_1 & \alpha_{12} \end{pmatrix}}{-\alpha_{02}} \in \mathbb{P}^1(\mathbb{F}) = \mathbb{F} \cup \{\infty\}.$$

By Lemma 2.1.9 it is well-defined.

Remark 2.1.11. Let $\overline{\rho}$ be maximally non-split as in (2.1.1). The isomorphism class of $\overline{\rho}$ is determined by the diagonal characters $\omega^{a_i+1} \operatorname{nr}_{\mu_i}$ and the Fontaine–Laffaille invariant $\operatorname{FL}(\overline{\rho})$. Note that $\operatorname{FL}(\overline{\rho})$ can take any value in $\mathbb{P}^1(\mathbb{F})$ except for μ_1 . (Similarly, a maximally non-split Galois representation $\overline{\rho}: G_{\mathbb{Q}_p} \to \operatorname{GL}_n(\mathbb{F})$ is determined by the diagonal characters and $\binom{n-1}{2}$ invariants.)

Remark 2.1.12. Note that in the situation of Definition 2.1.10, if \mathbb{F}'/\mathbb{F} is a finite field extension, then $FL(\overline{\rho} \otimes_{\mathbb{F}} \mathbb{F}') = FL(\overline{\rho})$.

Remark 2.1.13. We leave it to the reader to show that the Fontaine–Laffaille module associated to $\overline{\rho}^{\vee} \otimes \omega^{a_2+2}$ is described by

$$\begin{pmatrix} & 1 \\ & 1 & \\ 1 & & \end{pmatrix} \cdot {}^{t} \operatorname{Mat}_{\underline{e}}(\phi_{\bullet})^{-1} \cdot \begin{pmatrix} & 1 \\ & 1 & \\ 1 & & \end{pmatrix}.$$

As a consequence, $FL(\overline{\rho}^{\vee}) = FL(\overline{\rho})^{-1}$.

2.2. *p*-adic Hodge theory. This section mainly consists of a review of some integral *p*-adic Hodge theory, although many of the results are not available in the literature in the form or generality that we need.

In the first subsection (§2.2.1) we define the categories of mod-p objects we are going to work with (Breuil modules with descent data, étale φ -modules, etc.). Moreover we obtain a key result, Corollary 2.2.2, which provides a criterion for deciding when a given Breuil module with descent data and a Fontaine–Laffaille module have isomorphic Galois representations.

The second subsection (§2.2.2) is of a more technical nature. On the one hand we make two of the functors from §2.2.1 (relating Breuil, Fontaine–Laffaille and étale φ -modules) more explicit. We also provide a useful change-of-basis result for a Breuil module with descent data.

All missing proofs of this section are contained in §A.5. Our rationale is to state in this section all the results we need to prove our main results on the Galois side, and to relegate technical details to the appendix.

2.2.1. Breuil modules with tame descent data. Let $K' \subseteq K_0$ be a subfield containing \mathbb{Q}_p . The residual Breuil ring $\overline{S} \stackrel{\text{def}}{=} (k \otimes_{\mathbb{F}_p} \mathbb{F})[u]/(u^{ep})$ is equipped with

an action of $\operatorname{Gal}(K/K')$ by $k \otimes_{\mathbb{F}_p} \mathbb{F}$ -semilinear automorphisms. Explicitly if $g \in \operatorname{Gal}(K/K')$ and $a \in k \otimes_{\mathbb{F}_p} \mathbb{F}$, we have

$$\widehat{g}(au^i) \stackrel{\text{def}}{=} (g \otimes 1)(a) \cdot (\omega_{\varpi}(g) \otimes 1)^i u^i$$

If $\overline{\chi}$: $\operatorname{Gal}(K/K') \to \mathbb{F}^{\times}$ is a character, we write $\overline{S}_{\overline{\chi}}$ to denote the $\overline{\chi}$ -isotypical component of \overline{S} for the action of $\operatorname{Gal}(K/K')$.

We recall that \overline{S} is equipped with an $k \otimes_{\mathbb{F}_p} \mathbb{F}$ -linear derivation $N \stackrel{\text{def}}{=} -u \frac{\partial}{\partial u}$ and with a semilinear Frobenius φ defined by $u \mapsto u^p$ (semilinear with respect to the arithmetic Frobenius $\varphi \otimes 1$ on $k \otimes_{\mathbb{F}_p} \mathbb{F}$), which moreover commute with the action of $\operatorname{Gal}(K/K')$ on \overline{S} .

Fix $r \in \{0, \ldots, p-2\}$. A Breuil module of weight r with descent data from K to K' is a quintuple $(\mathcal{M}, \operatorname{Fil}^r \mathcal{M}, \varphi_r, N, \{\widehat{g}\})$, where

- (i) \mathcal{M} is a finite free \overline{S} -module;
- (ii) $\operatorname{Fil}^r \mathcal{M}$ is an \overline{S} -submodule of \mathcal{M} , verifying $u^{er}\mathcal{M} \subseteq \operatorname{Fil}^r \mathcal{M}$;
- (iii) a morphism φ_r : Fil^r $\mathcal{M} \to \mathcal{M}$ which is φ -semilinear and whose image generates \mathcal{M} ;
- (iv) the operator $N : \mathcal{M} \to \mathcal{M}$ is $k \otimes_{\mathbb{F}_p} \mathbb{F}$ -linear and satisfies certain axioms (see the beginning of Section 3.2 in [EGH13]);
- (v) an action of $\operatorname{Gal}(K/K')$ on \mathcal{M} by automorphisms \widehat{g} which are semilinear with respect to the Galois action on \overline{S} and which preserve $\operatorname{Fil}^r \mathcal{M}$ and commute with φ_r and N.

A morphism of Breuil modules is an \overline{S} -linear morphism which is compatible, in the evident sense, with the additional structures.

We write \mathbb{F} -BrMod^{*r*}_{dd} to denote the category of Breuil modules of weight *r* with descent data from *K* to *K'*; the field *K'* will always be clear from the context (and will be specified in case of ambiguities). As we did for the coefficient ring \overline{S} , given a character $\overline{\chi} : \operatorname{Gal}(K/K') \to \mathbb{F}^{\times}$ we write $\mathcal{M}_{\overline{\chi}}$, $(\operatorname{Fil}^r \mathcal{M})_{\overline{\chi}}$ to denote the $\overline{\chi}$ -isotypical component of \mathcal{M} , $\operatorname{Fil}^r \mathcal{M}$ respectively.

We remark that the category \mathbb{F} -BrMod^r_{dd} is additive and admits kernels and cokernels (cf. [Car11], Théorème 4.2.4 and the Remarque following it). In particular a complex

$$0 \to \mathcal{M}_1 \xrightarrow{f_1} \mathcal{M}_2 \xrightarrow{f_2} \mathcal{M}_3 \to 0$$

in \mathbb{F} -BrMod^r_{dd} is *exact* if the morphisms f_i induce exact sequences on the underlying \overline{S} -modules \mathcal{M}_j and Fil^r \mathcal{M}_j ($j \in \{0, 1, 2\}$). This endows \mathbb{F} -BrMod^r_{dd} with the structure of an exact category (see Proposition 2.3.4 below).

We recall that we have an exact, faithful, contravariant functor

$$\begin{aligned} \mathbf{T}^*_{\mathrm{st}} &: \mathbb{F}\text{-}\mathrm{BrMod}^r_{\mathrm{dd}} \to \mathrm{Rep}_{\mathbb{F}}(G_{K'}) \\ & \mathcal{M} \mapsto \mathbf{T}^*_{\mathrm{st}}(\mathcal{M}) \stackrel{\mathrm{def}}{=} \mathrm{Hom}(\mathcal{M}, \widehat{A}), \end{aligned}$$

where \widehat{A} is a certain period ring and homomorphisms respect all structures (cf. §A.3). We have dim_F $T_{st}^*(\mathcal{M}) = \operatorname{rank}_{\overline{S}} \mathcal{M}$ (cf. [Car11], Théorème 4.2.4 and the Remarque following it; see also [EGH13], Lemma 3.2.2).

We will be mainly concerned with the following covariant version of T_{st}^* : $T_{st}^r(\mathcal{M}) \stackrel{\text{def}}{=} (T_{st}^*(\mathcal{M}))^{\vee} \otimes \omega^r$. We remark that this is compatible with the notion of duality $\mathcal{M} \mapsto \mathcal{M}^*$ on \mathbb{F} -BrMod_{dd}^r recalled in §A.4, namely $T_{st}^r(\mathcal{M}) \cong T_{st}^*(\mathcal{M}^*)$, cf. the discussion before Corollary 3.2.9 in [EGH13].

We now move to the categories of étale φ -modules. In the remainder of this subsection we will assume $K' = K_0$, i.e. we will only consider descent data from K to K_0 .

We fix the field of norms $k((\underline{\omega}))$ associated to a suitable Kummer extension K_{∞} of K (see §A.1 for its precise definition). It is endowed with a *p*-power Frobenius endomorphism and with an action of $\text{Gal}(K/K_0)$.

An étale $(\phi, k \otimes_{\mathbb{F}_p} \mathbb{F}((\underline{\varpi})))$ -module with descent data is the datum of a finite free $k \otimes_{\mathbb{F}_p} \mathbb{F}((\underline{\varpi}))$ -module \mathfrak{M} endowed with a semilinear injective Frobenius endomorphism $\phi : \mathfrak{M} \to \mathfrak{M}$ and a semilinear action of $\operatorname{Gal}(K/K_0)$, commuting with ϕ . We write \mathbb{F} - $\mathfrak{Mod}_{k((\underline{\varpi})), dd}$ to denote the category of étale $(\phi, k \otimes_{\mathbb{F}_p} \mathbb{F}((\underline{\varpi})))$ -modules with descent data. In the special case when e = 1 this category is denoted by \mathbb{F} - $\mathfrak{Mod}_{k((\underline{p}))}$. (In this case $k((\underline{p}))$ is also the field of norms associated to a suitable Kummer extension $(K_0)_{\infty}/K_0$.) We remark that $k((\underline{\varpi}))/k((\underline{p}))$ is a cyclic extension of degree e with Galois group $\operatorname{Gal}(K/K_0)$, and we write $k((\underline{p}))^s$ to denote a choice of separable closure of k((p)) containing $k((\underline{\varpi}))$. For details, see §A.1.

Finally, recall the category of Fontaine–Laffaille modules $\mathbb{F}-\mathcal{FL}^{[0,p-2]}$ over $k \otimes_{\mathbb{F}_p} \mathbb{F}$ (defined in §2.1.1) and the category \mathbb{F} -BrMod^r_{dd} of Breuil modules of weight r with descent data from K to K_0 (defined in §2.2.1).

The relations between the categories introduced so far are summarized in the following proposition. Its proof, as well as the definition of the functors $M_{k((\underline{x}))}$, \mathcal{F}, \ldots can be found in Appendix A.

Proposition 2.2.1. Let $0 \le r \le p-2$. We have the following commutative diagram:



where the descent data is from K to K_0 . Moreover, the functor res $\circ T^*_{cris}$ is fully faithful.

We record the following immediate, yet crucial, corollary.

Corollary 2.2.2. Let $0 \le r \le p-2$ and let \mathfrak{M} , M be objects in \mathbb{F} -BrMod^r_{dd} and \mathbb{F} - $\mathfrak{FL}^{[0,p-2]}$ respectively. Assume that $T^*_{st}(\mathfrak{M})$ is Fontaine–Laffaille. If

$$M_{k((\varpi))}(\mathfrak{M}) \cong \mathfrak{F}(M) \otimes_{k((p))} k((\underline{\varpi}))$$

then one has an isomorphism of G_{K_0} -representations

$$T^*_{st}(\mathcal{M}) \cong T^*_{cris}(M).$$

Let us explain the role that Corollary 2.2.2 plays in the proof of our main theorem on the Galois side (Theorem 2.5.1). Thus suppose that ρ is potentially semistable of Hodge–Tate weights $\{-2, -1, 0\}$ with reduction $\overline{\rho}$ that is maximally non-split and generic as in Definition 2.1.5. Associated to ρ is a strongly divisible module $\widehat{\mathcal{M}}$, whose reduction $\mathcal{M} \in \mathbb{F}$ -BrMod^r_{dd} has Galois representation $\overline{\rho}$. Corollary 2.2.2 will allow us to compute the Fontaine–Laffaille module M associated to $\overline{\rho}$ – and hence the invariant $\operatorname{FL}(\overline{\rho})$ – in terms of $\widehat{\mathcal{M}}$, so in terms of ρ . In fact, we will start with \mathcal{M} in the top left corner of the diagram and then go around it in a clockwise sense: first computing $M_{k((\underline{\varpi}))}(\mathcal{M})$, then descending it to \mathbb{F} - $\mathfrak{Mod}_{k((\underline{p}))}$ and finally to \mathbb{F} - $\mathfrak{FL}^{[0,p-2]}$. To obtain the precise result, Theorem 2.5.1, we will moreover need more information about $\widehat{\mathcal{M}}$, and this will be obtained in §2.4.

2.2.2. Linear algebra with descent data. We continue to assume that $K' = K_0$. It will be convenient to introduce bases \underline{e} (resp. generating sets \underline{f}) of a Breuil module \mathcal{M} with descent data (resp. of Fil^{*r*} \mathcal{M}) that are compatible with the action of Gal(K/K_0), and to describe Fil^{*r*} \mathcal{M} (resp. φ_r) by matrices with respect to \underline{e} , \underline{f} . We then use this formalism to make the functors $M_{k((\underline{\varpi}))}$, \mathcal{F} appearing in the diagram of Proposition 2.2.1 more explicit, as well as we obtain a change-of-basis result for a Breuil module with descent data. Proofs can be found in §A.5.

Definition 2.2.3. We say that a Breuil module $\mathcal{M} \in \mathbb{F}$ -BrMod^{*r*}_{dd} is of type $\omega_{\varpi}^{a_0} \oplus \cdots \oplus \omega_{\varpi}^{a_{n-1}}$ (where $a_i \in \mathbb{Z}$) if $\mathcal{M}/u\mathcal{M} \cong \bigoplus_{i=0}^{n-1}(\omega_{\varpi}^{a_i} \otimes 1)$ as $(k \otimes_{\mathbb{F}_p} \mathbb{F})[\operatorname{Gal}(K/K_0)]$ modules. Equivalently, \mathcal{M} has an \overline{S} -basis (e_0, \ldots, e_{n-1}) such that $\widehat{g}e_i = (\omega_{\varpi}^{a_i}(g) \otimes 1)e_i$ for all *i* and all $g \in \operatorname{Gal}(K/K_0)$. We call such a basis a framed basis of \mathcal{M} .

1) e_i for all i and all $g \in \operatorname{Gal}(K/K_0)$. We call such a basis a framed basis of \mathcal{M} . If \mathcal{M} is of type $\omega_{\varpi}^{a_0} \oplus \cdots \oplus \omega_{\varpi}^{a_{n-1}}$ we say that (f_0, \ldots, f_{n-1}) is a framed system of generators of $\operatorname{Fil}^r \mathcal{M}$ if $\operatorname{Fil}^r \mathcal{M} = \sum_{i=0}^{n-1} \overline{S} \cdot f_i$ and $\widehat{g}f_i = (\omega_{\varpi}^{p^{-1}a_i}(g) \otimes 1)f_i$ for all i and all $g \in \operatorname{Gal}(K/K_0)$.

To justify the claim implicit in this definition, choose an \overline{S}/u -basis ($\overline{e}_1, \ldots, \overline{e}_{n-1}$) of $\mathcal{M}/u\mathcal{M}$ such that $\widehat{g} \cdot \overline{e}_i = (\omega_{\overline{\varpi}}^{a_i}(g) \otimes 1)\overline{e}_i$ for all i and all $g \in \operatorname{Gal}(K/K_0)$. Since $(k \otimes_{\mathbb{F}_p} \mathbb{F})[\operatorname{Gal}(K/K_0)]$ is a semisimple (commutative) ring we can pick a $(k \otimes_{\mathbb{F}_p} \mathbb{F})[\operatorname{Gal}(K/K_0)]$ -linear splitting of $\mathcal{M} \twoheadrightarrow \mathcal{M}/u\mathcal{M}$ and hence find $e_i \in \mathcal{M}$ lifting \overline{e}_i such that $\widehat{g}e_i = (\omega_{\overline{\varpi}}^{a_i}(g) \otimes 1)e_i$. By Nakayama's lemma, the e_i form an \overline{S} -basis of \mathcal{M} .

The notion of a framed basis (resp. a framed system of generators) depends on an ordering of the integers a_i . It will always be clear from the context which ordering we use.

Lemma 2.2.4. If $\mathcal{M} \in \mathbb{F}$ -BrMod^{*r*}_{dd} is of type $\bigoplus_{i=0}^{n-1} \omega_{\varpi}^{a_i}$, then Fil^{*r*} \mathcal{M} admits a framed system of generators.

Definition 2.2.5. Let $\mathcal{M} \in \mathbb{F}$ -BrMod^{*r*}_{dd} be of type $\omega_{\varpi}^{a_0} \oplus \cdots \oplus \omega_{\varpi}^{a_{n-1}}$. Let $\underline{e}, \underline{f}$ be a framed basis and a framed system of generators of \mathcal{M} , Fil^{*r*} \mathcal{M} respectively. The *matrix of the filtration*, with respect to $\underline{e}, \underline{f}$, is the element $\operatorname{Mat}_{\underline{e}, \underline{f}}(\operatorname{Fil}^r \mathcal{M}) \in \operatorname{M}_n(\overline{S})$ verifying

$$f = \underline{e} \cdot \operatorname{Mat}_{\underline{e},f} (\operatorname{Fil}^r \mathcal{M}).$$

We define the matrix of Frobenius with respect to $\underline{e}, \underline{f}$ as the element $\operatorname{Mat}_{\underline{e},\underline{f}}(\varphi_r) \in \operatorname{GL}_n(\overline{S})$ characterized by

$$\varphi_r(\underline{f}) = \underline{e} \cdot \operatorname{Mat}_{\underline{e},f}(\varphi_r).$$

As we require $\underline{e}, \underline{f}$ to be framed, the coefficients $\operatorname{Mat}_{\underline{e},\underline{f}}(\operatorname{Fil}^r \mathfrak{M})_{ij}$ verify the following conditions:

$$\operatorname{Mat}_{\underline{e},\underline{f}}(\operatorname{Fil}^{r} \mathcal{M})_{ij} \in \overline{S}_{\omega_{\overline{u}}^{p^{-1}a_{j}-a_{i}}}$$

Concretely $\operatorname{Mat}_{\underline{e},\underline{f}}(\operatorname{Fil}^{r} \mathfrak{M})_{ij} = u^{[p^{-1}a_{j}-a_{i}]}s_{ij}$, where $[x] \in \{0,\ldots,e-1\}$ is defined by $[x] \equiv x \mod e$ for $x \in \mathbb{Z}/e\mathbb{Z}$ and $s_{ij} \in \overline{S}_{\omega_{\varpi}^{0}} = k \otimes_{\mathbb{F}_{p}} \mathbb{F}[u^{e}]/(u^{ep})$.

On the other hand, $\operatorname{Mat}_{\underline{e},f}(\varphi_r) \in \operatorname{GL}_n^{\square}(\overline{S})$, where

$$\operatorname{GL}_{n}^{\Box}(\overline{S}) \stackrel{\text{def}}{=} \left\{ A \in \operatorname{GL}_{n}(\overline{S}) : A_{ij} \in \overline{S}_{\omega_{\varpi}^{a_{j}-a_{i}}} \text{ for all } 0 \le i, j \le n-1 \right\}.$$

Lemma 2.2.6. Let \mathcal{M} be a Breuil module of type $\omega_{\varpi}^{a_0} \oplus \cdots \oplus \omega_{\varpi}^{a_{n-1}}$, and let \underline{e} be a framed basis of \mathcal{M} and \underline{f} a framed system of generators of $\operatorname{Fil}^r \mathcal{M}$, respectively. Let $V \stackrel{\text{def}}{=} \operatorname{Mat}_{\underline{e},\underline{f}}(\operatorname{Fil}^r \mathcal{M}) \in \operatorname{M}_n(\overline{S})$ and $A \stackrel{\text{def}}{=} \operatorname{Mat}_{\underline{e},\underline{f}}(\varphi_r) \in \operatorname{GL}_n^{\square}(\overline{S})$. Then there exists a basis $\underline{\mathfrak{e}}$ of $M_{k((\underline{\varpi}))}(\mathcal{M}^*)$ with $\widehat{g} \cdot \mathbf{e}_i = (\omega_{\overline{\varpi}}^{-p^{-1}a_i}(g) \otimes 1)\mathbf{e}_i$ for all i and $g \in$ $\operatorname{Gal}(K/K_0)$, such that $\operatorname{Mat}_{\underline{\mathfrak{e}}}(\phi) \in \operatorname{M}_n(k \otimes_{\mathbb{F}_p} \mathbb{F}[[\underline{\varpi}]])$ is given by any chosen lift of ${}^tV \cdot {}^tA^{-1} \in \operatorname{M}_n(\overline{S})$ via the morphism $k \otimes_{\mathbb{F}_p} \mathbb{F}[[\underline{\varpi}]] \twoheadrightarrow \overline{S}$ sending $\sum \lambda_i \underline{\varpi}^i$ to $\sum \lambda_i u^i$ and such that $(\operatorname{Mat}_{\underline{\mathfrak{e}}}(\phi))_{ij} \in (k \otimes_{\mathbb{F}_p} \mathbb{F}[[\underline{\varpi}]])_{\omega_{\underline{m}}}^{-p^{-1}a_i-a_j}$ for all $0 \leq i, j \leq n-1$.

Lemma 2.2.7. Let $M \in \mathbb{F}$ - $\mathcal{FL}^{[0,p-2]}$ be a rank *n* Fontaine–Laffaille module with parallel Hodge–Tate weights $0 \leq m_0 \leq \cdots \leq m_{n-1} \leq p-2$ (counted with multiplicities).

Let $\underline{e} = (e_0, \ldots, e_{n-1})$ be a $k \otimes_{\mathbb{F}_p} \mathbb{F}$ -basis of M_i that is compatible with the Hodge filtration $\operatorname{Fil}^{\bullet} M$, and let $F \in \operatorname{GL}_n(k \otimes_{\mathbb{F}_p} \mathbb{F})$ be the associated matrix of the Frobenius $\phi_{\bullet} : \operatorname{gr}^{\bullet} M \xrightarrow{\sim} M$.

Then there exists a basis $\underline{\mathfrak{e}}$ of $\mathfrak{M} \stackrel{\text{def}}{=} \mathfrak{F}(M)$ in \mathbb{F} - $\mathfrak{Mod}_{k((\underline{p}))}$ such that the Frobenius ϕ on \mathfrak{M} is described by

$$\operatorname{Mat}_{\mathfrak{e}}(\phi) = \operatorname{Diag}(p^{m_0} \dots p^{m_{n-1}})F.$$

Lemma 2.2.8. Let $\mathcal{M} \in \mathbb{F}$ -BrMod^r_{dd} be a Breuil module of type $\omega_{\varpi}^{a_0} \oplus \cdots \oplus \omega_{\varpi}^{a_{n-1}}$ and let $\underline{e}, \underline{f}$ be a framed basis of \mathcal{M} and a framed system of generators of Fil^r \mathcal{M} respectively.

Write $V \stackrel{\text{def}}{=} \operatorname{Mat}_{\underline{e},\underline{f}}(\operatorname{Fil}^{r} \mathfrak{M}), A \stackrel{\text{def}}{=} \operatorname{Mat}_{\underline{e},\underline{f}}(\varphi_{r}).$ Assume that there exist elements $V' \in \operatorname{M}_{n}(\overline{S}), B \in \operatorname{GL}_{n}(\overline{S})$ such that $V'_{ij} \in \overline{\overline{S}}_{\omega_{\overline{\omega}}^{p^{-1}a_{j}-a_{i}}}, B_{ij} \in \overline{S}_{\omega_{\overline{\omega}}^{p^{-1}(a_{j}-a_{i})}}$ and

(2.2.9)
$$AV' \equiv VB \mod u^{e(r+1)}.$$

Then $\underline{e}' \stackrel{\text{def}}{=} \underline{e} \cdot A$ is a framed basis of \mathcal{M} , $\underline{f}' \stackrel{\text{def}}{=} \underline{e}' \cdot V'$ is a framed system of generators for Fil^r \mathcal{M} , and $\operatorname{Mat}_{e',f'}(\varphi_r) = \varphi(B)$.

2.3. Breuil submodules and Galois representations. In this subsection we discuss some preliminaries on subobjects and quotients in the category \mathbb{F} -BrMod^r_{dd}. Even though these notions are presumably well-known to the experts, we did not find a suitable reference in the literature. The main result, Proposition 2.3.5, is a slight improvement of a result of Caruso ([Car11]). All missing proofs of this subsection are contained in §A.6.

In what follows, we let $\overline{S}_k \stackrel{\text{def}}{=} k[u]/u^{ep}$.

Definition 2.3.1. Let \mathcal{M} be an object in \mathbb{F} -BrMod^{*r*}_{dd}. An \overline{S} -submodule $\mathcal{N} \subseteq \mathcal{M}$ is said to be a *Breuil submodule* if \mathcal{N} fulfills the following conditions:

- (i) \mathcal{N} is an \overline{S}_k -direct summand of \mathcal{M} ;
- (ii) $N(\mathcal{N}) \subseteq \mathcal{N}$ and $\widehat{g}(\mathcal{N}) \subseteq \mathcal{N}$ for all $g \in \operatorname{Gal}(K/K')$;
- (iii) $\varphi_r(\mathcal{N} \cap \operatorname{Fil}^r \mathcal{M}) \subseteq \mathcal{N}.$

Lemma 2.3.2. Let \mathcal{M} be an object in \mathbb{F} -BrMod^r_{dd} and let $\mathcal{N} \subseteq \mathcal{M}$ be a Breuil submodule. Then the \overline{S} -modules \mathcal{N} , \mathcal{M}/\mathcal{N} with their induced structures are objects of \mathbb{F} -BrMod^r_{dd} and the sequence

$$0 \to \mathbb{N} \to \mathbb{M} \to \mathbb{M}/\mathbb{N} \to 0$$

is exact in \mathbb{F} -BrMod^r_{dd}. Conversely, given an exact sequence

$$0 \to \mathcal{M}_1 \xrightarrow{J} \mathcal{M} \to \mathcal{M}_2 \to 0$$

in \mathbb{F} -BrMod^r_{dd}, then $f(\mathfrak{M}_1) \subseteq \mathfrak{M}$ is a Breuil submodule.

An immediate consequence of Lemma 2.3.2 is that the notion of Breuil submodule is transitive:

Lemma 2.3.3. Let \mathcal{M} be an object in \mathbb{F} -BrMod^r_{dd}.

- (i) If $\mathcal{M}_1 \subseteq \mathcal{M}$ and $\mathcal{M}_2 \subseteq \mathcal{M}_1$ are Breuil submodules, then so is $\mathcal{M}_2 \subseteq \mathcal{M}$.
- (ii) Let M₁, M₂ be Breuil submodules of M, and assume that M₂ ⊆ M₁. Then M₂ is a Breuil submodule of M₁, and the Breuil module structures on M₂ inherited from M₁ and M coincide. Similarly, M₁/M₂ ⊆ M/M₂ is a Breuil submodule, and the Breuil module structures on M₁/M₂, as a Breuil submodule of M/M₂ and as a quotient of M₁, coincide.

Proposition 2.3.4. With the above notion of exact sequence, the category \mathbb{F} -BrMod^r_{dd} is an exact category in the sense of [Kel90] and T^r_{st} is an exact functor.

We can now state a crucial result relating Breuil submodules of \mathcal{M} and subrepresentations of $T_{st}^r(\mathcal{M})$. It improves [Car11], Proposition 2.2.5 (cf. also [EGH13], Corollary 3.2.9).

Proposition 2.3.5. Let \mathcal{M} be an object in \mathbb{F} -BrMod^r_{dd}. There is a natural orderpreserving bijection

 $\Theta: \{Breuil submodules in \mathcal{M}\} \xrightarrow{\sim} \{G_{K'} \text{-subrepresentations of } T^r_{st}(\mathcal{M})\}$

sending $\mathbb{N} \subseteq \mathbb{M}$ to the image of $\mathrm{T}^r_{\mathrm{st}}(\mathbb{N}) \hookrightarrow \mathrm{T}^r_{\mathrm{st}}(\mathbb{M})$. Moreover, if $\mathbb{M}_2 \subseteq \mathbb{M}_1$ are Breuil submodules of \mathbb{M} , then $\Theta(\mathbb{M}_1)/\Theta(\mathbb{M}_2) \cong \mathrm{T}^r_{\mathrm{st}}(\mathbb{M}_1/\mathbb{M}_2)$.

2.4. On filtrations. Our goal is to give a rough classification of the filtration on certain strongly divisible modules $\widehat{\mathcal{M}}$ of rank 3 whose associated *p*-adic Galois representation ρ has maximally non-split and generic reduction $\overline{\rho}$. The idea is to start with a simpler analysis in characteristic *p* (§2.4.1), using that the mod *p* reduction \mathcal{M} of $\widehat{\mathcal{M}}$ (a Breuil module) has associated mod *p* Galois representation $\overline{\rho}$. The resulting Corollary 2.4.7 concerning Fil² \mathcal{M} helps us in our analysis of Fil² $\widehat{\mathcal{M}}$ in §2.4.2. We obtain our classification in Proposition 2.4.10. This will be a key input into our main local result on the Galois side, Theorem 2.5.1.

Our conventions regarding strongly divisible modules will be explained at the start of §2.4.2. For the remainder of Section 2 we will assume that $e = p^f - 1$.

2.4.1. *Filtration on Breuil modules*. We now obtain the first structure results for Breuil modules with descent data giving rise to maximally non-split $\overline{\rho}$.

Proposition 2.4.1. Let $K_0 = \mathbb{Q}_p$ and suppose that $\mathfrak{M} \in \mathbb{F}$ -BrMod²_{dd} is of type $\omega^{a_0} \oplus \omega^{a_1} \oplus \omega^{a_2}$. Assume that $\overline{\rho} \stackrel{\text{def}}{=} T^2_{\text{st}}(\mathfrak{M})$ is maximally non-split and generic as in Definition 2.1.5.

There is a framed basis $\underline{e} = (e_0, e_1, e_2)$ of \mathcal{M} and a framed system of generators $\underline{f} = (f_0, f_1, f_2)$ for Fil² \mathcal{M} such that the coordinates of the elements in \underline{f} with respect to \underline{e} are described as follows:

$$f_{0} = \begin{pmatrix} u^{e} \\ \mu u^{e-(a_{1}-a_{0})} \\ \nu u^{e-(a_{2}-a_{0})} \end{pmatrix}, \quad f_{1} = \begin{pmatrix} 0 \\ u^{e} \\ \lambda u^{e-(a_{2}-a_{1})} \end{pmatrix}, \quad f_{2} = \begin{pmatrix} 0 \\ 0 \\ u^{e} \end{pmatrix},$$

where $\lambda, \mu, \nu \in \mathbb{F}$ verify moreover $\lambda \mu \neq 0$.

The proof of Proposition 2.4.1 will occupy the remainder of this subsection. We start with the following lemma. It gives a concrete criterion for the Galois representation associated to a rank two Breuil module to split as a direct sum of characters.

Lemma 2.4.2. Assume $K_0 = \mathbb{Q}_p$ (so e = p - 1) and let

$$0 \to \mathcal{M}_1 \to \mathcal{M} \to \mathcal{M}_2 \to 0$$

be an extension of rank one objects in \mathbb{F} -BrMod²_{dd}. For each $i \in \{1, 2\}$, assume that \mathcal{M}_i is of type ω^{a_i} , with $\omega^{a_1} \not\cong \omega^{a_2}$ and suppose Fil² $\mathcal{M}_i = u^{\delta_i e} \mathcal{M}_i$ for $0 \leq \delta_1 \leq \delta_2 \leq 2$. Finally, assume that $\overline{\rho} \stackrel{\text{def}}{=} T^2_{\text{st}}(\mathcal{M})$ is Fontaine–Laffaille (possibly after a twist).

(i) If the extension of \overline{S} -modules

$$(2.4.3) 0 \to \operatorname{Fil}^2 \mathfrak{M}_1 \to \operatorname{Fil}^2 \mathfrak{M} \to \operatorname{Fil}^2 \mathfrak{M}_2 \to 0$$

splits, then $\overline{\rho}$ splits as a direct sum of two characters.

(ii) If $\delta_1 = 1$, $\delta_2 = 2$ then $\overline{\rho}$ splits as a direct sum of two characters.

Proof. By assumption, \mathcal{M} is of type $\omega^{a_1} \oplus \omega^{a_2}$. Let $\underline{e} \stackrel{\text{def}}{=} (e_1, e_2)$ be a framed basis of \mathcal{M} such that $\mathcal{M}_1 = \overline{S} \cdot e_1$. In what follows, we define $[a_2 - a_1] \in \{1, \ldots, e - 1\}$ by the requirement $[a_2 - a_1] \equiv a_2 - a_1$ modulo e.

From the exact sequence (2.4.3) we can find a framed system of generators of $\operatorname{Fil}^2 \mathcal{M}$ of the form $f_1 = u^{\delta_1 e} e_1$, $f_2 = u^{\delta_2 e} e_2 + x u^{[a_2 - a_1]} e_1$, where $x \in \overline{S}_{\omega^0}$. Moreover, we have

$$A \stackrel{\text{def}}{=} \operatorname{Mat}_{\underline{e}, \underline{f}}(\varphi_2) = \begin{pmatrix} \alpha & \gamma u^{\lfloor a_2 - a_1 \rfloor} \\ \beta \end{pmatrix} \in \operatorname{GL}_2(\overline{S})$$

for some $\alpha, \beta \in \overline{S}_{\omega^0}^{\times}$ and $\gamma \in \overline{S}_{\omega^0}$.

Assume that the sequence (2.4.3) splits. Then we can fix a splitting $s : \operatorname{Fil}^2 \mathfrak{M}_2 \to \operatorname{Fil}^2 \mathfrak{M}$ which we can moreover assume to be $\operatorname{Gal}(K/K_0)$ -equivariant (by averaging). Let $\overline{e}_2 \in (\mathfrak{M}_2)_{\omega^{a_2}}$ be a generator of \mathfrak{M}_2 . Then $s(u^{e\delta_2}\overline{e}_2)$ is killed by $u^{e(p-\delta_2)}$ and hence of the form $u^{\delta_2 e} e_2$ for some $e_2 \in \mathfrak{M}$. Without loss of generality, $e_2 \in \mathfrak{M}_{\omega^{a_2}}$. It is now easy to see that (e_1, e_2) is a framed basis of \mathfrak{M} and that $(f_1, f_2) = (u^{\delta_1 e} e_1, u^{\delta_2 e} e_2)$ is a framed system of generators of $\operatorname{Fil}^2 \mathfrak{M}$. In other words, we can take x = 0 above. From the obvious matrix equality

$$A\begin{pmatrix} u^{\delta_1 e} \\ u^{\delta_2 e} \end{pmatrix} = \begin{pmatrix} u^{\delta_1 e} \\ u^{\delta_2 e} \end{pmatrix} \underbrace{\begin{pmatrix} \alpha & \gamma u^{(\delta_2 - \delta_1)e + [a_2 - a_1]} \\ \beta \end{pmatrix}}_{\substack{\text{def}_B}}$$

and Lemma 2.2.8, we deduce that $\underline{e}' \stackrel{\text{def}}{=} \underline{e} \cdot A$ is a basis of \mathcal{M} such that $\mathcal{M}_1 = \overline{S} \cdot e'_1$, that $f' \stackrel{\text{def}}{=} (u^{\delta_1 e} e'_1, u^{\delta_2 e} e'_2)$ is a system of generators for Fil² \mathcal{M} and that

$$\operatorname{Mat}_{\underline{e}',\underline{f}'}(\varphi_2) = \varphi(B) = \begin{pmatrix} \varphi(\alpha) & \varphi(\gamma)u^{[a_2-a_1]p} \\ & \varphi(\beta) \end{pmatrix}.$$

As $[a_2 - a_1] > 0$ we can further assume that $\varphi(\gamma) = 0$, up to re-iterating the procedure above. Therefore $M_{\mathbb{F}_p((\underline{\varpi}))}(\mathcal{M})$ splits into a direct sum of rank one $(\varphi, \mathbb{F}((\underline{\varpi})))$ modules (as can be immediately checked by Lemma 2.2.6), hence $\overline{\rho}|_{G_{(\mathbb{Q}_p)\infty}}$ splits as a direct sum of two characters. As $\overline{\rho}$ is Fontaine–Laffaille, we deduce from Proposition 2.2.1 that $\overline{\rho}$ splits into a direct sum of two characters, completing the proof of case (i).

Let us assume that $\delta_1 = 1$, $\delta_2 = 2$. We then have $f_2 = u^{2e}e_2 + xu^{[a_2-a_1]}e_1$ and by adding a multiple of f_1 we may assume $x \in \mathbb{F}$. If $x \neq 0$ it follows that $\operatorname{Fil}^2 \mathcal{M} = \langle u^{[a_2-a_1]}e_1, u^{3e-[a_2-a_1]}e_2 \rangle$, so $u^{2e}e_2 \notin \operatorname{Fil}^2 \mathcal{M}$, contradiction. Therefore x = 0 and we are in the situation of (i). \Box

Remark 2.4.4. The analogous statement of Lemma 2.4.2 holds when $K_0 \neq \mathbb{Q}_p$. This will be described in detail in a particular case in the proof of Proposition 2.6.4, when $[K_0 : \mathbb{Q}_p] = 2$ and developed in further generality in [MP], Lemma 3.2.

We now make use of our knowledge of rank one Breuil modules with descent data and their associated Galois representations to start to understand $\text{Fil}^2 \mathcal{M}$ in Proposition 2.4.1.

Lemma 2.4.5. Assume $K_0 = \mathbb{Q}_p$ and let $\overline{\rho} : G_{\mathbb{Q}_p} \to \operatorname{GL}_3(\mathbb{F})$ be maximally non-split and generic as in Definition 2.1.5. Let $\mathfrak{M} \in \mathbb{F}\operatorname{-BrMod}_{\operatorname{dd}}^2$ be of type $\omega^{a_0} \oplus \omega^{a_1} \oplus \omega^{a_2}$ and such that $\operatorname{T}^2_{\operatorname{st}}(\mathfrak{M}) \cong \overline{\rho}$ and write $0 = \mathfrak{M}_3 \subsetneq \mathfrak{M}_2 \subsetneq \mathfrak{M}_1 \subsetneq \mathfrak{M}_0 \stackrel{\text{def}}{=} \mathfrak{M}$ to denote the Breuil submodule filtration on \mathfrak{M} deduced from the socle filtration on $\overline{\rho}$ (cf. Proposition 2.3.5).

Then for each $i \in \{0, 1, 2\}$ the rank one quotient $\mathfrak{M}_i/\mathfrak{M}_{i+1} \in \mathbb{F} ext{-BrMod}_{dd}^2$ is of type ω^{a_i} and its filtration is described by

$$\operatorname{Fil}^{2}\left(\mathcal{M}_{i}/\mathcal{M}_{i+1}\right) = u^{e}\left(\mathcal{M}_{i}/\mathcal{M}_{i+1}\right).$$

Proof. By Proposition 2.3.5, there exists a permutation $\sigma \in S_3$ such that for any $i \in \{0, 1, 2\}$ the rank one module $\mathcal{M}_i/\mathcal{M}_{i+1}$ is of type $\omega^{a_{\sigma(i)}}$. This implies, by means of [EGH13], Lemma 3.3.2, that there exists $\delta_i \in \{0, 1, 2\}$ such that $\operatorname{Fil}^2(\mathcal{M}_i/\mathcal{M}_{i+1}) = u^{e\delta_i}(\mathcal{M}_i/\mathcal{M}_{i+1})$ and $\operatorname{T}^2_{\operatorname{st}}(\mathcal{M}_i/\mathcal{M}_{i+1})|_{I_{\mathbb{Q}_p}} \cong \omega^{a_{\sigma(i)}+\delta_i}$.

On the other hand we have $T^2_{st}(\mathcal{M}_i/\mathcal{M}_{i+1})|_{I_{\mathbb{Q}_p}} \cong \omega^{a_i+1}$ for all i, by the definition of the filtration $\{\mathcal{M}_i\}_i$. As $\overline{\rho}$ is generic (cf. Definition 2.1.5) we conclude that $\sigma = \mathrm{id}$ and that $\delta_i = 1$ for all i.

Proof of Proposition 2.4.1. Let $0 \subsetneq \mathcal{M}_2 \subsetneq \mathcal{M}_1 \subsetneq \mathcal{M}_0 \stackrel{\text{def}}{=} \mathcal{M}$ be the filtration by Breuil submodules on \mathcal{M} , obtained from the socle filtration on $\overline{\rho}$ via Proposition

2.3.5. From Lemma 2.4.5 we obtain a framed basis (e_0, e_1, e_2) of \mathcal{M} with $e_i \in \mathcal{M}_i$ such that

(2.4.6)
$$\operatorname{Fil}^{2}\left(\mathcal{M}_{i}/\mathcal{M}_{i+1}\right) = \langle u^{e}\overline{e}_{i}\rangle_{\overline{S}}$$

for $i \in \{0, 1, 2\}$ (with the obvious notation for the elements \overline{e}_i).

As the descent data acts semisimply on \mathcal{M}_1 , from the exact sequence

 $0 \rightarrow \operatorname{Fil}^2 \mathcal{M}_2 \rightarrow \operatorname{Fil}^2 \mathcal{M}_1 \rightarrow \operatorname{Fil}^2 (\mathcal{M}_1/\mathcal{M}_2) \rightarrow 0$

we see that

$$\operatorname{Fil}^2 \mathfrak{M}_1 = \langle u^e e_2, u^e e_1 + s e_2 \rangle_{\overline{S}} \quad \text{for some} \quad s \in \overline{S}_{\omega^{a_1 - a_2}} = u^{e - (a_2 - a_1)} \overline{S}_{\omega^0}.$$

Without loss of generality, we can assume $s = \lambda u^{e-(a_2-a_1)}$ for some $\lambda \in \mathbb{F}$ and, by Lemma 2.4.2 and the non-splitness assumption, we moreover have $\lambda \neq 0$.

In a completely similar fashion we obtain

$$\operatorname{Fil}^{2} \mathcal{M} = \langle u^{e} e_{2}, \ u^{e} e_{1} + \lambda u^{e - (a_{2} - a_{1})} e_{2}, \ u^{e} e_{0} + s_{1} e_{1} + s_{2} e_{2} \rangle_{\overline{S}},$$

where $s_1 \in \overline{S}_{\omega^{a_0-a_1}}$, $s_2 \in \overline{S}_{\omega^{a_0-a_2}}$. As above, we can assume without loss of generality that $s_1 = \mu u^{e-(a_1-a_0)}$ and $s_2 = \nu u^{e-(a_2-a_0)}$ and we furthermore deduce from Lemma 2.4.2 (applied to $\mathcal{M}/\mathcal{M}_2$) that $\nu \neq 0$.

The following immediate corollary of Proposition 2.4.1 will play an important role in describing the filtration on certain strongly divisible \mathcal{O}_E -modules, see §2.4.2.

Corollary 2.4.7. Let \mathcal{M} be a Breuil module and $\lambda, \nu, \mu \in \mathbb{F}$ as in the statement of Proposition 2.4.1.

The elementary divisors for $\mathcal{M}/\operatorname{Fil}^2\mathcal{M}$ as an $\mathbb{F}[u]$ -module are described by one of the following possibilities:

- (i) if $\nu(\nu \lambda \mu) \neq 0$, by $(u^{e-(a_2-a_0)}, u^e, u^{e+(a_2-a_0)})$;
- (ii) if $\nu \lambda \mu = 0$, by $(u^{e^{-(a_2 a_0)}}, u^{e^{+(a_2 a_1)}}, u^{e^{+(a_1 a_0)}})$;
- (iii) if $\nu = 0$, by $(u^{e^{-(a_2-a_1)}}, u^{e^{-(a_1-a_0)}}, u^{e^{+(a_2-a_0)}})$.

We also have:

- (iv) Fil² $\mathcal{M} \subseteq u^{e-(a_2-a_0)}\mathcal{M};$
- (v) $(\operatorname{Fil}^2 \mathfrak{M})_{\omega^{a_0}} \not\subseteq u^e \mathfrak{M};$ (vi) $(\operatorname{Fil}^2 \mathfrak{M} \cap u^e \mathfrak{M})_{\omega^{a_0}} \subseteq u^{2e (a_2 a_0)} \mathfrak{M};$
- (vii) $(\operatorname{Fil}^2 \mathfrak{M})_{u^{a_2}} \subseteq u^e \mathfrak{M}.$

2.4.2. Filtration on strongly divisible modules. In this section we pursue the analysis started in $\S2.4.1$. The main result of this section is Proposition 2.4.10.

As in §2.2.1, we let $K' \subseteq K_0$ be a subfield containing \mathbb{Q}_p . The ring $S_{W(k)}$ (cf. [Bre97], §4.1) is defined as the *p*-adic completion of $W(k)[u, \frac{u^{ie}}{i!}]_{i\in\mathbb{N}}$. The ring $S_{W(k)}$ is endowed with a descending filtration ${\operatorname{Fil}^{i} S_{W(k)}}_{i \in \mathbb{N}}$, a semilinear Frobenius φ , a W(k)-linear derivation N, and with a Galois action by W(k)-algebra endomorphisms defined by $\widehat{g}(u) = \widetilde{\omega}_{\varpi}(g)u$ for any $g \in \operatorname{Gal}(K/K')$. In particular, the action of any $g \in \operatorname{Gal}(K/K')$ preserves the filtration and commutes with the Frobenius and the monodromy on $S_{W(k)}$. By extension of scalars, the ring $S \stackrel{\text{def}}{=} S_{W(k)} \otimes_{\mathbb{Z}_p} \mathbb{O}_E$ is endowed with the additional structures inherited from $S_{W(k)}$. Note in particular that we have a natural map $S \twoheadrightarrow \overline{S}$, defined as the reduction modulo $(\varpi_E, \operatorname{Fil}^p S)$. For more details, see [EGH13], §3.1.

Fix $r \in \{0, \ldots, p-2\}$. We refer to [EGH13], §3.1 for the definition of the category \mathcal{O}_E -Mod^r_{dd} of strongly divisible \mathcal{O}_E -modules of weight r and with descent data from K to K'. The objects are certain quintuples $(\mathcal{M}, \operatorname{Fil}^r \mathcal{M}, \varphi, N, \{\widehat{g}\})$, where \mathcal{M} is a finite free S-module. There is a contravariant functor $T_{st}^{*,K'}$: \mathcal{O}_E -Mod $_{dd}^r \rightarrow$ $\operatorname{Rep}_{\mathcal{O}_{E}}(G_{K'})$, which by a theorem of T. Liu ([Liu08], Theorem 2.3.5, cf. also [EGH13], Proposition 3.1.4) provides an equivalence of categories of \mathcal{O}_E -Mod^r_{dd} with the category of $G_{K'}$ -stable \mathcal{O}_E -lattices in finite dimensional *E*-representations of $G_{K'}$ that become semistable over K and have Hodge–Tate weights in [-r, 0]. As in the case of Breuil modules, we consider its covariant version defined by $\mathbf{T}_{\mathrm{st}}^{K',r}(\widehat{\mathcal{M}}) \stackrel{\mathrm{def}}{=} \left(\mathbf{T}_{\mathrm{st}}^{*,K'}(\widehat{\mathcal{M}})\right)^{\vee} \otimes \varepsilon^{r}.$

We also recall that if $\widehat{\mathcal{M}} \in \mathcal{O}_E$ -Mod^r_{dd} the base change $\widehat{\mathcal{M}} \otimes_S \overline{S}$ is naturally an object of \mathbb{F} -BrMod^r_{dd} and one has a natural isomorphism $T^{*,K'}_{st}(\widehat{\mathcal{M}}) \otimes_{\mathcal{O}_E} \mathbb{F} \cong$ $T^*_{st}(\mathcal{M} \otimes_S \overline{S})$ of $\mathbb{F}[G_{K'}]$ -modules.

We write $\operatorname{Mod}_{E}^{w.a.}(\varphi, N, K/K')$ for the category of weakly admissible filtered $(\varphi, N, K/K', E)$ -modules (see e.g. [EGH13], §3.1). In particular, the underlying objects are finite free $K_0 \otimes_{\mathbb{Q}_p} E$ -modules. We recall the contravariant equivalence of categories $D_{st}^{*,K'}$: $\operatorname{Rep}_{E}^{K-st}(G_{K'}) \to \operatorname{Mod}_{E}^{w.a.}(\varphi, N, K/K')$, where $\operatorname{Rep}_{E}^{K-st}(G_{K'})$ denotes the category of finite dimensional *E*-representations of $G_{K'}$ that become semistable over K. If $\rho \in \operatorname{Rep}_E^{K-\operatorname{cris}}(G_{K'})$ has Hodge–Tate weights in [-r, 0], we define $\operatorname{D}_{\operatorname{st}}^{K',r}(\rho) \stackrel{\text{def}}{=} \operatorname{D}_{\operatorname{st}}^{*,K'}(\rho^{\vee} \otimes \varepsilon^r)$.

As in the mod p setting, given an $\mathcal{O}_E[\operatorname{Gal}(K/K')]$ -module X and a character $\chi: \mathrm{Gal}(K/K') \to \mathbb{O}_E^{\times}$ we let X_{χ} denote its $\chi\text{-isotypical component.}$

We first require two lemmas.

Lemma 2.4.8. Assume that $K' = \mathbb{Q}_p$ and that the p-adic Galois representation $\mathbf{T}^{\mathbb{Q}_p,r}_{\mathrm{st}}(\widehat{\mathfrak{M}}) \otimes_{\mathbb{O}_E} E \text{ has inertial type } \oplus_{i=0}^{n-1} \widetilde{\omega}^{a_i}_f. \text{ Then } \widehat{\mathfrak{M}} \otimes_S \overline{S} \in \mathbb{F}\text{-}\mathrm{Br}\mathrm{Mod}^r_{\mathrm{dd}} \text{ is of type}$ $\oplus_{i=0}^{n-1}\omega_{\varpi}^{a_i}.$

Proof. The assumption that $K' = \mathbb{Q}_p$ implies that the multi-set $\{\widetilde{\omega}_f^{a_i}\}_{i=0}^{n-1}$ (and hence the multi-set $\{\omega_{\varpi}^{a_i}\}_{i=0}^{n-1}$ is stable under the action of the *p*-power Frobenius. Together with [EGH13], Proposition 3.3.1, this is all we need to use the argument at the beginning of [EGH13], proof of Theorem 3.3.13 to construct the required basis for $\mathcal{M} \otimes_S \overline{S}$.

Lemma 2.4.9. Assume $K_0 = \mathbb{Q}_p$ and let $\rho : G_{\mathbb{Q}_p} \to \mathrm{GL}_3(E)$ be a Galois representation, becoming semistable over K, with Hodge–Tate weights $\{-2, -1, 0\}$. Let $\widehat{\mathcal{M}} \in \mathcal{O}_E$ -Mod²_{dd} be such that $T^{\mathbb{Q}_p, 2}_{st}(\widehat{\mathcal{M}}) \otimes_{\mathcal{O}_E} E \cong \rho$. Let $X \stackrel{\text{def}}{=} \left(\frac{\operatorname{Fil}^2 \widehat{\mathcal{M}}}{\operatorname{Fil}^2 S \cdot \widehat{\mathcal{M}}}\right) \otimes_{\mathcal{O}_E} E$ and $Y \stackrel{\text{def}}{=} \left(\frac{\operatorname{Fil}^1 S \cdot \widehat{\mathcal{M}}}{\operatorname{Fil}^2 S \cdot \widehat{\mathcal{M}}}\right) \otimes_{\mathcal{O}_E} E$. Then for any character

 $\chi: \operatorname{Gal}(K/\mathbb{Q}_p) \to E^{\times}$ we have

- (i) $\dim_E X_{\chi} = 3;$
- (ii) $\dim_E (X \cap Y)_{\chi} = 2.$

Moreover, multiplication by $u \in S$ induces an isomorphism $X_{\chi} \xrightarrow{\sim} X_{\chi \widetilde{\omega}}$.

Proof. Let $D \stackrel{\text{def}}{=} D_{\text{st}}^{*,\mathbb{Q}_p}(\rho)$. As $\rho|_{G_K}$ is semistable, with Hodge–Tate weights $\{-2, -1, 0\}$, the *E*-linear spaces $\operatorname{gr}^{i}(\operatorname{Fil}^{\bullet} D_{K})$ are at most one dimensional and they are non-zero if and only if $i \in \{0, 1, 2\}$.

Let $\mathcal{D} \stackrel{\text{def}}{=} D \otimes_E S_E$. We then have $X = \operatorname{Fil}^2 \mathcal{D} / (\operatorname{Fil}^2 S_E \cdot \mathcal{D}), Y = (\operatorname{Fil}^1 S_E \cdot \mathcal{D}) / (\operatorname{Fil}^2 S_E \cdot \mathcal{D})$ and by the analogue with *E*-coefficients of [Bre97], Proposition A.4, we deduce that

$$\operatorname{Fil}^{1} \mathcal{D} = (\operatorname{Fil}^{1} S_{E}) \widehat{f}_{0} \oplus S_{E} \widehat{f}_{1} \oplus S_{E} \widehat{f}_{2}, \quad \operatorname{Fil}^{2} \mathcal{D} = (\operatorname{Fil}^{2} S_{E}) \widehat{f}_{0} \oplus (\operatorname{Fil}^{1} S_{E}) \widehat{f}_{1} \oplus S_{E} \widehat{f}_{2},$$

for some S_E -basis \hat{f}_i of \mathcal{D} .

From the E-linear isomorphism

$$\frac{S}{\operatorname{Fil}^m S} \cong \bigoplus_{i=0}^{m-1} \bigoplus_{j=0}^{e-1} \langle u^j E(u)^i \rangle_E$$

for $m \leq p$, we deduce that $\dim_E X = 3e$, $\dim_E(X \cap Y) = 2e$. We now note that $\operatorname{Gal}(K/\mathbb{Q}_p)$ acts semisimply and that multiplication by u defines an E-linear automorphism on $S_E/\operatorname{Fil}^p S_E$ which cyclically permutes isotypical components. The result follows.

We can now state the main result of this section:

Proposition 2.4.10. Assume that $K_0 = \mathbb{Q}_p$ and let $\rho : G_{\mathbb{Q}_p} \to \operatorname{GL}_3(\mathbb{O}_E)$ be a semistable Galois representation, becoming crystalline over K, with inertial type $\widetilde{\omega}^{a_0} \oplus \widetilde{\omega}^{a_1} \oplus \widetilde{\omega}^{a_2}$ and Hodge–Tate weights $\{-2, -1, 0\}$. Let $\widehat{\mathcal{M}} \in \mathcal{O}_E$ -Mod²_{dd} be a strongly divisible \mathcal{O}_E -module such that $\operatorname{T}^{\mathbb{Q}_p,2}_{\operatorname{st}}(\widehat{\mathcal{M}}) \cong \rho$. Assume that $\overline{\rho} : G_{\mathbb{Q}_p} \to \operatorname{GL}_3(\mathbb{F})$ is maximally non-split and generic as in Definition 2.1.5.

There exist a framed basis $\hat{\underline{e}} = (\hat{e}_0, \hat{e}_1, \hat{e}_2)$ of $\widehat{\mathcal{M}}$, with respect to $\widetilde{\omega}^{a_0} \oplus \widetilde{\omega}^{a_1} \oplus \widetilde{\omega}^{a_2}$, and a framed system of generators $\underline{\widehat{f}} = (\widehat{f}_0, \widehat{f}_1, \widehat{f}_2)$ for $\operatorname{Fil}^2 \widehat{\mathcal{M}} / (\operatorname{Fil}^p S \cdot \widehat{\mathcal{M}})$, whose coordinates with respect to $\underline{\widehat{e}}$ are described by one of the following possibilities. Case A:

$$\widehat{f}_0 = \begin{pmatrix} \alpha \\ 0 \\ u^{e-(a_2-a_0)} \end{pmatrix}, \ \widehat{f}_1 = E(u) \begin{pmatrix} 0 \\ 1 \\ 0 \end{pmatrix}, \ \widehat{f}_2 = \begin{pmatrix} (p+E(u))u^{a_2-a_0} \\ 0 \\ -\frac{p^2}{\alpha} \end{pmatrix},$$

where $\alpha \in \mathcal{O}_E$ with $0 < \operatorname{ord}_p(\alpha) < 2$.

Case B:

$$\widehat{f}_0 = \begin{pmatrix} \alpha \\ 0 \\ u^{e-(a_2-a_0)} \end{pmatrix}, \ \widehat{f}_1 = E(u) \begin{pmatrix} u^{a_1-a_0} \\ -\frac{p}{\beta} \\ 0 \end{pmatrix}, \ \widehat{f}_2 = \begin{pmatrix} \beta u^{a_2-a_0} \\ E(u)u^{a_2-a_1} \\ -\frac{p\beta}{\alpha} \end{pmatrix},$$

where $\alpha, \beta \in \mathcal{O}_E$ with $0 < \operatorname{ord}_p(\beta) < 1$ and $0 < \operatorname{ord}_p(\alpha) < \operatorname{ord}_p(\beta) + 1$. Case C:

$$\widehat{f}_0 = \begin{pmatrix} -\frac{p\alpha}{\beta} \\ u^{e-(a_1-a_0)} \\ \frac{p}{\beta}u^{e-(a_2-a_0)} \end{pmatrix}, \ \widehat{f}_1 = \begin{pmatrix} -\alpha u^{a_1-a_0} \\ -\beta \\ u^{e-(a_2-a_1)} \end{pmatrix}, \ \widehat{f}_2 = \begin{pmatrix} (p+E(u))u^{a_2-a_0} \\ \frac{p\beta}{\alpha}u^{a_2-a_1} \\ \frac{p^2}{\alpha} \end{pmatrix},$$

where $\alpha, \beta \in \mathcal{O}_E$ with $0 < \operatorname{ord}_p(\beta) < 1$ and $0 < \operatorname{ord}_p(\alpha) < \operatorname{ord}_p(\beta) + 1$.

Here, by a framed basis $\hat{\underline{e}}$ of $\widehat{\mathcal{M}}$ we mean an S-basis such that $\widehat{g} \cdot \widehat{e}_i = \widetilde{\omega}(g)^{a_i} \widehat{e}_i$ for all $0 \leq i \leq 2, g \in \operatorname{Gal}(K/K_0)$. Similarly, we mean that $\underline{\widehat{f}}$ consists of elements of Fil² $\widehat{\mathcal{M}}$ that generate Fil² $\widehat{\mathcal{M}}/(\operatorname{Fil}^p S \cdot \widehat{\mathcal{M}})$ as S-module and such that $\widehat{g} \cdot \widehat{f}_i = \widetilde{\omega}(g)^{a_i} \widehat{f}_i$ for all $0 \leq i \leq 2, g \in \operatorname{Gal}(K/K_0)$. Finally, ord_p is the valuation of E normalized by $\operatorname{ord}_p(p) = 1$. *Proof.* Let $\underline{e} \stackrel{\text{def}}{=} (e_0, e_1, e_2)$ be a framed basis of $\widehat{\mathcal{M}}$. We write the elements of $\widehat{\mathcal{M}}$ in terms of coordinates with respect to \underline{e} . Moreover, we let $\mathcal{M} \stackrel{\text{def}}{=} \widehat{\mathcal{M}} \otimes_S \overline{S}$, define $\mathcal{D} \stackrel{\text{def}}{=} \widehat{\mathcal{M}} \otimes_{\mathcal{O}_E} S_E$, and set $X = (\operatorname{Fil}^2 \mathcal{M} / \operatorname{Fil}^2 S \cdot \mathcal{M}) \otimes_{\mathcal{O}_E} E$ as in Lemma 2.4.9. In particular $\mathcal{M} \in \mathbb{F}$ -BrMod²_{dd} is of type $\omega^{a_0} \oplus \omega^{a_1} \oplus \omega^{a_2}$ by Lemma 2.4.8.

By Lemma 2.4.9 we have a non-zero element $f_0 \in (\operatorname{Fil}^2 \widehat{\mathcal{M}} / \operatorname{Fil}^2 S \cdot \widehat{\mathcal{M}})_{\widetilde{\omega}^{a_0}}$ of the form

$$f_0 = \begin{pmatrix} x \\ yu^{e-(a_1-a_0)} \\ zu^{e-(a_2-a_0)} \end{pmatrix} + E(u) \begin{pmatrix} x' \\ y'u^{e-(a_1-a_0)} \\ z'u^{e-(a_2-a_0)} \end{pmatrix},$$

where $x, y, z, x', y', z' \in \mathcal{O}_E$. As $\operatorname{Fil}^2 \widehat{\mathcal{M}} / \operatorname{Fil}^2 S \cdot \widehat{\mathcal{M}}$ is ϖ_E -torsion free, we may assume that one of x, y, z, x', y', z' is in \mathcal{O}_E^{\times} . If $x, y, z \in \varpi_E \mathcal{O}_E$ for all choices of f_0 , we get a contradiction from Corollary 2.4.7(v). On the other hand by Corollary 2.4.7(iv) we necessarily have $x \in \varpi_E \mathcal{O}_E$, hence we may assume that $y \in \mathcal{O}_E^{\times}$ or $z \in \mathcal{O}_E^{\times}$.

Case 1: assume $z \in \mathcal{O}_E^{\times}$. Let us define the element $e'_2 \in \widehat{\mathcal{M}}$ by

$$e_{2}' = \begin{pmatrix} x'u^{a_{2}-a_{0}} \\ (y+y'E(u))u^{a_{2}-a_{1}} \\ z+z'E(u) \end{pmatrix}.$$

As $z \in \mathcal{O}_E^{\times}$ we deduce by Nakayama's lemma that $\underline{e}' \stackrel{\text{def}}{=} (e_0, e_1, e_2')$ is again a framed basis of $\widehat{\mathcal{M}}$. By letting $\alpha \stackrel{\text{def}}{=} x + px'$ we therefore have the following coordinates for f_0 in the basis \underline{e}' :

$$f_0 = \begin{pmatrix} \alpha \\ 0 \\ u^{e-(a_2-a_0)} \end{pmatrix},$$

where $\operatorname{ord}_p(\alpha) > 0$. (Recall that $E(u) = u^e + p$.) From now on we use the basis \underline{e}' to write the coordinates of the elements in $\widehat{\mathcal{M}}$.

By Lemma 2.4.9 we easily deduce that $\left(\operatorname{Fil}^{2}\widehat{\mathcal{M}}/\operatorname{Fil}^{2}S\cdot\widehat{\mathcal{M}}\right)_{\widetilde{\omega}^{a_{0}}}$ equals

$$\left\langle \left(\begin{array}{c} \alpha\\ 0\\ u^{e-(a_2-a_0)} \end{array}\right), E(u) \left(\begin{array}{c} \alpha\\ 0\\ u^{e-(a_2-a_0)} \end{array}\right), E(u) \left(\begin{array}{c} \beta\\ \gamma u^{e-(a_1-a_0)}\\ 0 \end{array}\right) \right\rangle_{\mathcal{O}_E},$$

where $\beta, \gamma \in \mathcal{O}_E$ are such that either $\beta \neq 0$ or $\gamma \neq 0$ modulo ϖ_E . Moreover, by Corollary 2.4.7(vi) we necessarily have $\beta \equiv 0$ so that, without loss of generality, we may assume $\gamma = 1$. By Lemma 2.4.9 we obtain

$$X_{\widetilde{\omega}^{a_1}} = \left\langle \left(\begin{array}{c} \alpha u^{a_1 - a_0} \\ 0 \\ u^{e - (a_2 - a_1)} \end{array} \right), E(u) \left(\begin{array}{c} \alpha u^{a_1 - a_0} \\ 0 \\ u^{e - (a_2 - a_1)} \end{array} \right), E(u) \left(\begin{array}{c} \beta u^{a_1 - a_0} \\ -p \\ 0 \end{array} \right) \right\rangle_E$$

and we need to further distinguish two subcases according to the valuation of $\beta \in \mathcal{O}_E$.

Case 1a: assume $\beta \in p\mathcal{O}_E$. Then the element e'_1 defined by

$$e_1' = \begin{pmatrix} -\frac{\beta}{p}u^{a_1-a_0} \\ 1 \\ 0 \end{pmatrix}$$

is in $\left(\operatorname{Fil}^2 \widehat{\mathcal{M}}/\operatorname{Fil}^2 S \cdot \widehat{\mathcal{M}}\right)_{\widetilde{\omega}^{a_1}}$ and the family $\underline{e}'' \stackrel{\text{def}}{=} (e_0, e_1', e_2')$ is again a framed basis of $\widehat{\mathcal{M}}$. Until the end of Case 1a we use the basis \underline{e}'' to write the coordinates of the elements in $\widehat{\mathcal{M}}$.

Note that

$$\frac{1}{\alpha}u^{a_2-a_0}(p+E(u))f_0 = \begin{pmatrix} (p+E(u))u^{a_2-a_0}\\ 0\\ -\frac{p^2}{\alpha} \end{pmatrix} \in X_{\widetilde{\omega}^{a_2}}.$$

By Corollary 2.4.7(vii) we necessarily have $\frac{p^2}{\alpha} \in \varpi_E \mathcal{O}_E$ and therefore

$$\left\langle \left(\begin{array}{c} \alpha\\ 0\\ u^{e-(a_2-a_0)} \end{array}\right), E(u) \left(\begin{array}{c} 0\\ 1\\ 0 \end{array}\right), \left(\begin{array}{c} (p+E(u))u^{a_2-a_0}\\ 0\\ -\frac{p^2}{\alpha} \end{array}\right) \right\rangle_S \subseteq \operatorname{Fil}^2 \widehat{\mathcal{M}}/\operatorname{Fil}^2 S \cdot \widehat{\mathcal{M}}.$$

By Nakayama's lemma, noticing that the left-hand side surjects onto Fil² $\mathcal{M}/u^{2e}\mathcal{M}$ (e.g. by Corollary 2.4.7) we conclude that the inclusion is indeed an equality. We also see that the elementary divisors of $\mathcal{M}/\operatorname{Fil}^2 \mathcal{M}$ are those described by Corollary 2.4.7(i). In fact, we can repeat the Nakayama argument to deduce that the same three elements generate the S-module Fil² $\widehat{\mathcal{M}}/\operatorname{Fil}^p S \cdot \widehat{\mathcal{M}}$, since their images generate Fil² \mathcal{M} , so they form a framed system of generators of this module. (Alternatively there is a direct argument using Fil² $S = (E(u)^2, \operatorname{Fil}^p S)$.) We therefore land in Case A.

Case 1b: assume $p \in \beta \varpi_E \mathcal{O}_E$. As above we deduce

$$E(u) \begin{pmatrix} u^{a_1-a_0} \\ -\frac{p}{\beta} \\ 0 \end{pmatrix} \in \left(\operatorname{Fil}^2 \widehat{\mathcal{M}}/\operatorname{Fil}^2 S \cdot \widehat{\mathcal{M}}\right)_{\widetilde{\omega}^{a_1}}$$

and

$$\left\langle \left(\begin{array}{c} \alpha u^{a_2 - a_0} \\ 0 \\ -p \end{array} \right) + E(u) \left(\begin{array}{c} 0 \\ 0 \\ 1 \end{array} \right), E(u) \left(\begin{array}{c} \alpha u^{a_2 - a_0} \\ 0 \\ -p \end{array} \right), E(u) \left(\begin{array}{c} u^{a_2 - a_0} \\ -\frac{p}{\beta} u^{a_2 - a_1} \\ 0 \end{array} \right) \right\rangle_{\mathcal{O}_E} \subseteq \left(\operatorname{Fil}^2 \widehat{\mathcal{M}} / \operatorname{Fil}^2 S \cdot \widehat{\mathcal{M}} \right)_{\widetilde{\omega}^{a_2}}.$$

In particular

$$\begin{pmatrix} \beta u^{a_2-a_0} \\ u^{a_2-a_1} E(u) \\ -\frac{p\beta}{\alpha} \end{pmatrix} \in X_{\widetilde{\omega}^{a_2}}$$

and, by Corollary 2.4.7(vii) we necessarily have $\beta \in \varpi_E \mathcal{O}_E$ and $p\beta \in \alpha \varpi_E \mathcal{O}_E$. We thus obtain

$$\left\langle \begin{pmatrix} \alpha \\ 0 \\ u^{e-(a_2-a_0)} \end{pmatrix}, E(u) \begin{pmatrix} u^{a_1-a_0} \\ -\frac{p}{\beta} \\ 0 \end{pmatrix}, \begin{pmatrix} \beta u^{a_2-a_0} \\ u^{a_2-a_1}E(u) \\ -\frac{p\beta}{\alpha} \end{pmatrix} \right\rangle_S \subseteq \operatorname{Fil}^2 \widehat{\mathcal{M}} / \operatorname{Fil}^2 S \cdot \widehat{\mathcal{M}}.$$

It follows, as in Case 1a, that the inclusion is actually an equality and we land in Case B. We also see that the elementary divisors of $M/Fil^2 M$ are those described by Corollary 2.4.7(ii).

Case 2: assume $y \in \mathcal{O}_E^{\times}$ and $z \in \varpi_E \mathcal{O}_E$. We note that the image of f_0 in $(\operatorname{Fil}^2 \mathcal{M})_{\omega^{a_0}}$ is divisible by $u^{e-(a_1-a_0)}$ but not by u^e . From Proposition 2.4.1 we deduce that the elementary divisors of $\mathcal{M}/\operatorname{Fil}^2 \mathcal{M}$ are those described by Corollary 2.4.7(iii). It is then not hard to see that Case B of the proposition applies to

the dual strongly divisible module $\widehat{\mathcal{M}}^*$ ([Car05], Chapitre V) upon interchanging (a_0, a_1, a_2) by $(-a_2, -a_1, -a_0)$. We arrive at Case C by dualizing. Alternatively, Case 2 can be treated directly with some effort.

Corollary 2.4.11. Let ρ and $\widetilde{\mathcal{M}}$ be as in Proposition 2.4.10. Write λ_i for the Frobenius eigenvalue on $\mathbb{D}_{\mathrm{st}}^{\mathbb{Q}_p,2}(\rho)^{I_{\mathbb{Q}_p}=\widetilde{\omega}^{a_i}}$. Then

$$(\operatorname{ord}_p(\lambda_0), \operatorname{ord}_p(\lambda_1), \operatorname{ord}_p(\lambda_2)) = \begin{cases} (2 - \operatorname{ord}_p(\alpha), 1, \operatorname{ord}_p(\alpha)) \\ (2 - \operatorname{ord}_p(\alpha), \operatorname{ord}_p(\beta), 1 + \operatorname{ord}_p(\alpha) - \operatorname{ord}_p(\beta)) \\ (1 + \operatorname{ord}_p(\beta) - \operatorname{ord}_p(\alpha), 2 - \operatorname{ord}_p(\beta), \operatorname{ord}_p(\alpha)) \end{cases}$$

according to whether we are in Case A, B, or C of Proposition 2.4.10. In particular, $\operatorname{ord}_p(\lambda_i) \in (0,2)$ for all i and $\sum_{i=0}^{2} \operatorname{ord}_p(\lambda_i) = 3$.

Proof. We first note that the elements $\varphi_2(\widehat{f}_i)$ span $\widehat{\mathcal{M}}$ by Nakayama's lemma, as $\varphi_2(\operatorname{Fil}^p S \cdot \widehat{\mathcal{M}}) \subset p^{p-2} \widehat{\mathcal{M}} \subset \mathfrak{m}_S \widehat{\mathcal{M}}$ and p > 2.

From the proof of [EGH13], Proposition 3.1.4 we have $D_{st}^{\mathbb{Q}_p,2}(\rho) \cong \widehat{\mathcal{M}}[\frac{1}{p}] \otimes_{S_{\mathbb{Q}_p},s_0} \mathbb{Q}_p$ where $s_0: S_{\mathbb{Q}_p} \to \mathbb{Q}_p$ is defined by " $u \mapsto 0$ ". We have $\varphi_2(\widehat{f}_j) = \sum_{i=0}^2 \widehat{\alpha}_{ij} \widehat{e}_i$ where $\widehat{\alpha}_{ij} \in S_{\widetilde{\omega}^{a_j-a_i}}$. Hence, as the elements $\varphi_2(\widehat{f}_i)$ span $\widehat{\mathcal{M}}$, we see that $\widehat{\alpha}_{ii} \in S^{\times}$ for all $0 \leq i \leq 2$. On the other hand note that $S_{\widetilde{\omega}^i} \subseteq \ker(s_0)$ for all $i \not\equiv 0 \mod p - 1$. Let e_i, f_i denote the images of $\widehat{e}_i, \widehat{f}_i$ in $D_{st}^{\mathbb{Q}_p,2}(\rho)$. In Case A we now deduce that $\frac{\varphi}{p^2}(f_0) = \mu e_0$ where $\mu \stackrel{\text{def}}{=} s_0(\widehat{\alpha}_{00}) \in \mathcal{O}_E^{\times}$. In other words, $\alpha \varphi(e_0) = p^2 \mu e_0$ so that $\lambda_0 = p^2 \mu \alpha^{-1}$ with $\operatorname{ord}_p(\lambda_0) = 2 - \operatorname{ord}_p(\alpha)$. The other cases are deduced similarly. \Box

2.5. From strongly divisible modules to Fontaine–Laffaille modules. The aim of this section is to explicitly determine the Fontaine–Laffaille module associated to the mod-p reduction of a potentially semistable lift of a maximally non-split and generic Galois representation $\bar{\rho}$. Using Proposition 2.4.10 we compute the étale φ -modules associated to the mod-p reduction of strongly divisible modules, letting us compare the Frobenius eigenvalues on the weakly admissible module of certain potentially semistable lifts of $\bar{\rho}$ with the Fontaine–Laffaille invariant FL($\bar{\rho}$).

Let red : $\mathbb{P}^1(\mathcal{O}_E) \to \mathbb{P}^1(\mathbb{F})$ denote the specialization map. The main local result on the Galois side is the following theorem. It will be an important input for proving our global Theorem 4.5.2.

Theorem 2.5.1. Let $\rho: G_{\mathbb{Q}_p} \to \operatorname{GL}_3(\mathbb{O}_E)$ be a potentially semistable *p*-adic Galois representation of Hodge–Tate weights $\{-2, -1, 0\}$ and inertial type $\widetilde{\omega}^{a_0} \oplus \widetilde{\omega}^{a_1} \oplus \widetilde{\omega}^{a_2}$. Assume that the residual representation $\overline{\rho}: G_{\mathbb{Q}_p} \to \operatorname{GL}_3(\mathbb{F})$ is maximally non-split and generic as in Definition 2.1.5. Let $\lambda_i \in \mathbb{O}_E$ be the Frobenius eigenvalue on $\operatorname{D}_{\operatorname{st}}^{\mathbb{Q}_p,2}(\rho)^{I_{\mathbb{Q}_p}=\widetilde{\omega}^{a_i}}$. Then the Fontaine–Laffaille invariant of $\overline{\rho}$ is given by:

$$\operatorname{FL}(\overline{\rho}) = \operatorname{red}\left(p\lambda_1^{-1}\right).$$

We first determine the filtration and the Frobenius action on the Breuil modules obtained as the base change of strongly divisible \mathcal{O}_E -modules corresponding to \mathcal{O}_E -lattices inside ρ .

Proposition 2.5.2. Let $\widehat{\mathcal{M}}$, $\underline{\widehat{e}}$, $\underline{\widehat{f}}$ be as in the statement of Proposition 2.4.10. Define $\alpha_i \in \mathbb{F}^{\times}$ via the condition $[\varphi_2(\widehat{f}_i)] = \alpha_i[\widehat{e}_i]$ in $\widehat{\mathcal{M}}/\mathfrak{m}_S \widehat{\mathcal{M}}$ and let $\mathcal{M} \in \mathbb{F}$ -BrMod²_{dd} be the base change of $\widehat{\mathcal{M}}$ via $S \twoheadrightarrow \overline{S}$. Then \mathcal{M} is of type $\omega^{a_0} \oplus \omega^{a_1} \oplus \omega^{a_2}$ and there

exist a framed basis \underline{e} of \mathcal{M} and a framed system of generators \underline{f} for $\operatorname{Fil}^2 \mathcal{M}$ such that one of the following hold:

(i) Fil² $\widehat{\mathcal{M}}$ is as in Case A of Proposition 2.4.10 and

$$\operatorname{Mat}_{\underline{e},\underline{f}}\left(\operatorname{Fil}^{2} \mathcal{M}\right) = \begin{pmatrix} c_{00}u^{e} & c_{01}u^{e+(a_{1}-a_{0})} & u^{e+(a_{2}-a_{0})} \\ c_{10}u^{e-(a_{1}-a_{0})} & u^{e} \\ u^{e-(a_{2}-a_{0})} & & \end{pmatrix}; \operatorname{Mat}_{\underline{e},\underline{f}}(\varphi_{2}) = \operatorname{Diag}(\alpha_{0},\alpha_{1},\alpha_{2})$$

for some $c_{ij} \in \mathbb{F}$.

(ii) $\operatorname{Fil}^2 \widehat{\mathcal{M}}$ is as in Case B of Proposition 2.4.10 and

$$\operatorname{Mat}_{\underline{e},\underline{f}}\left(\operatorname{Fil}^{2} \mathcal{M}\right) = \begin{pmatrix} c_{00}u^{e} & u^{e+(a_{1}-a_{0})} \\ c_{10}u^{e-(a_{1}-a_{0})} & u^{e+(a_{2}-a_{1})} \\ u^{e-(a_{2}-a_{0})} & & \end{pmatrix}; \operatorname{Mat}_{\underline{e},\underline{f}}(\varphi_{2}) = \operatorname{Diag}(\alpha_{1},\alpha_{2},\alpha_{0})$$

for some $c_{ij} \in \mathbb{F}$.

(iii) $\operatorname{Fil}^2 \widehat{\mathcal{M}}$ is as in Case C of Proposition 2.4.10 and

$$\operatorname{Mat}_{\underline{e},\underline{f}}\left(\operatorname{Fil}^{2} \mathcal{M}\right) = \begin{pmatrix} c_{00}u^{e} & c_{01}u^{e+(a_{1}-a_{0})} & u^{e+(a_{2}-a_{0})} \\ u^{e-(a_{1}-a_{0})} & & \\ & u^{e-(a_{2}-a_{1})} \end{pmatrix}; \operatorname{Mat}_{\underline{e},\underline{f}}(\varphi_{2}) = \operatorname{Diag}(\alpha_{2},\alpha_{0},\alpha_{1})$$

for some $c_{ij} \in \mathbb{F}$.

Proof. We let $a \stackrel{\text{def}}{=} a_2 - a_0$, $b \stackrel{\text{def}}{=} a_1 - a_0$ and define $\underline{e}_0 \stackrel{\text{def}}{=} \widehat{e} \otimes 1$, $\underline{f}_0 \stackrel{\text{def}}{=} \widehat{f} \otimes 1$. Then \mathcal{M} is of type $\omega^{a_0} \oplus \omega^{a_1} \oplus \omega^{a_2}$, \underline{e}_0 is a framed basis of \mathcal{M} and \underline{f}_0 is a framed system of generators of Fil² \mathcal{M} . We let

$$V_0 \stackrel{\mathrm{def}}{=} \mathrm{Mat}_{\underline{e}_0, \underline{f}_0}(\mathrm{Fil}^2\, \mathbb{M}), \qquad \qquad A_0 \stackrel{\mathrm{def}}{=} \mathrm{Mat}_{\underline{e}_0, \underline{f}_0}(\varphi_2).$$

Note in particular that $\alpha_i \equiv (A_0)_{ii} \mod u^e$ for i = 0, 1, 2, by construction.

We now treat separately each of the three cases (i), (ii) and (iii).

(i) From Case A of Proposition 2.4.10 we deduce that

$$V_0 = \operatorname{Mat}_{\underline{e}_0, \underline{f}_0}(\operatorname{Fil}^2 \mathfrak{M}) = \begin{pmatrix} & u^{e+a} \\ & u^e & \\ & u^{e-a} & \end{pmatrix}.$$

We can write $V_0^{\text{adj}} = -u^e W_0$ for some $W_0 \in M_3(\overline{S})$, which is uniquely defined modulo $u^{e(p-1)}$; in fact, we can and will take $W_0 = V_0$. We claim that there exist $b_{00}, b_{01}, b_{10} \in \mathbb{F}$ such that

(2.5.3)
$$W_0 \cdot A_0 \cdot \underbrace{\begin{pmatrix} b_{00}u^e & b_{01}u^{e+b} & u^{e+a} \\ b_{10}u^{e-b} & u^e \\ u^{e-a} & \\ & &$$

for some $B_0 \in \operatorname{GL}_3^{\square}(\overline{S})$. Indeed, if (2.5.3) holds, one obtains

$$V_0 B_0 \equiv A_0 V_1 \bmod u^{3e}$$

(as $p \geq 5$) and by Lemma 2.2.8 we deduce that $V_1 = \operatorname{Mat}_{\underline{e}_1, \underline{f}_1}(\operatorname{Fil}^2 \mathcal{M})$ is a matrix for Fil² \mathcal{M} with respect to the basis $\underline{e}_1 \stackrel{\text{def}}{=} \underline{e}_0 \cdot A_0$ and a system of generators \underline{f}_1 of Fil² \mathcal{M} , and that $A_1 \stackrel{\text{def}}{=} \operatorname{Mat}_{\underline{e}_1, \underline{f}_1}(\varphi_2) = \varphi(B_0)$ describes the Frobenius action on \mathcal{M} with respect to $(\underline{e}_1, \underline{f}_1)$. To prove the claim, for $0 \leq i, j \leq 2$ let us write $(A_0)_{ij} = u^{[a_j - a_i]} \alpha_{ij}$ with $\alpha_{ij} \in \mathbb{F}[u^e]/u^{ep}$. A computation, using the explicit description of the element $W_0 \in \mathrm{M}_3^{\square}(\overline{S})$, gives:

$$W_0 \cdot A_0 \cdot V_1 \equiv \begin{pmatrix} \alpha_{22} u^{2e} & & \\ x u^{e-b} & \alpha_{11} u^{2e} & \\ y u^{e-a} & z u^{e-(a-b)} & \alpha_{00} u^{2e} \end{pmatrix} \mod u^{3e},$$

where $x, y, z \in \mathbb{F}[u^e]/u^{ep}$ are defined by

$$x = (\alpha_{11}b_{10} + \alpha_{12})u^e + \alpha_{10}b_{00}u^{2e}, \qquad y = (\alpha_{00}b_{00} + \alpha_{01}b_{10} + \alpha_{02})u^e$$
$$z = (\alpha_{00}b_{01} + \alpha_{01})u^e.$$

As $\alpha_{ij} \in \overline{S}_{\omega^0}$ and $\alpha_{ii} \in \overline{S}^{\times}$ we deduce the existence of unique $b_{00}, b_{10}, b_{01} \in \mathbb{F}$ such that (2.5.3) holds and, what is more important,

$$B_0 \equiv \begin{pmatrix} \alpha_2 & & \\ \beta_{10}u^{e-b} & \alpha_1 & \\ \beta_{20}u^{e-a} & \beta_{21}u^{e-(a-b)} & \alpha_0 \end{pmatrix} \mod u^e$$

for some $\beta_{ij} \in \mathbb{F}$. In particular, we obtain

$$A_1 \stackrel{\text{def}}{=} \operatorname{Mat}_{\underline{e}_1, \underline{f}_1}(\varphi_2) = \begin{pmatrix} \alpha_2 & & \\ \beta_{10} u^{p(e-b)} & \alpha_1 \\ \beta_{20} u^{p(e-a)} & \beta_{21} u^{p(e-(a-b))} & \alpha_0 \end{pmatrix}$$
$$\equiv \operatorname{Diag}(\alpha_2, \alpha_1, \alpha_0) \bmod u^{3e}$$

by the genericity assumption (2.1.6).

Writing again $V_1^{\text{adj}} = -u^e W_1$ for some $W_1 \in M_3(\overline{S})$ (uniquely defined modulo $u^{e(p-1)}$) we obtain

$$W_1 \cdot A_1 \equiv \begin{pmatrix} \alpha_0 u^{e+a} \\ u^e \alpha_1 & -\alpha_0 b_{10} u^{e+(a-b)} \\ \alpha_2 u^{e-a} & -\alpha_1 b_{01} u^{e-(a-b)} & \alpha_0 (b_{10} b_{01} - b_{00}) u^e \end{pmatrix} \mod u^{3e}$$

and a computation shows that there exist uniquely determined elements $c_{00}, c_{10}, c_{01} \in \mathbb{F}$ such that

(2.5.4)
$$W_1 \cdot A_1 \cdot \underbrace{\begin{pmatrix} c_{00}u^e & c_{01}u^{e+b} & u^{e+a} \\ c_{10}u^{e-b} & u^e \\ u^{e-a} & \\ \end{bmatrix}}_{\stackrel{\text{def}}{=} V_2} = u^{2e}B_1,$$

where now

$$B_1 \equiv \operatorname{Diag}(\alpha_0, \alpha_1, \alpha_2) \mod u^e.$$

By Lemma 2.2.8 one has $V_2 = \operatorname{Mat}_{\underline{e}_2,\underline{f}_2}(\operatorname{Fil}^2 \mathfrak{M})$, where $\underline{e}_2 \stackrel{\text{def}}{=} \underline{e}_1 \cdot A_1$, \underline{f}_2 is a system of generators for $\operatorname{Fil}^2 \mathfrak{M}$ and the Frobenius action is now given by:

$$A_2 \stackrel{\text{def}}{=} \operatorname{Mat}_{\underline{e}_2, \underline{f}_2}(\varphi_2) = \varphi(B_1) = \operatorname{Diag}(\alpha_0, \alpha_1, \alpha_2).$$

(ii) The proof is similar to (i) and we content ourselves with giving the general argument, leaving the computational details to the scrupulous reader.

From Proposition 2.4.10 the filtration for $\operatorname{Fil}^2 \mathcal{M}$ now has the form

$$V_0 \stackrel{\text{def}}{=} \left(\begin{array}{cc} u^{e+b} \\ & u^{e+(a-b)} \\ u^{e-a} \end{array} \right).$$

Writing $V_0^{\text{adj}} = u^e W_0$ for some $W_0 \in M_3(\overline{S})$ we claim that there exist $b_{00}, b_{10} \in \mathbb{F}$ and $B_0 \in \operatorname{GL}_3^{\square}(\overline{S})$ such that

Indeed, using the same notation as in case (i), an easy computation now gives:

$$W_0 \cdot A_0 \cdot V_1 \equiv \begin{pmatrix} \alpha_{22} u^{2e} & & \\ x u^{e-b} & \alpha_{00} u^{2e} & \alpha_{01} u^{2e+(a-b)} \\ y u^{e-a} & \alpha_{10} u^{3e-(a-b)} & \alpha_{11} u^{2e} \end{pmatrix} \mod u^{3e},$$

where $x, y \in \mathbb{F}[u^e]/u^{ep}$ are defined by

$$x = (\alpha_{00}b_{00} + \alpha_{01}b_{10} + \alpha_{02})u^e, \qquad y = (\alpha_{11}b_{10} + \alpha_{12})u^e + \alpha_{10}b_{00}u^{2e}$$

As $\alpha_{ii} \in \overline{S}^{\times}$ we find unique $b_{00}, b_{10} \in \mathbb{F}$ such that (2.5.5) holds and moreover

$$B_0 \equiv \begin{pmatrix} \alpha_2 & & \\ \beta_{10}u^{e-b} & \alpha_0 & \beta_{12}u^{a-b} \\ \beta_{20}u^{e-a} & \beta_{21}u^{e-(a-b)} & \alpha_1 \end{pmatrix} \mod u^e$$

for some $\beta_{ij} \in \mathbb{F}$. Note that

$$A_1 \stackrel{\text{def}}{=} \operatorname{Mat}_{\underline{e}_1, \underline{f}_1}(\varphi_2) \equiv \operatorname{Diag}(\alpha_2, \alpha_0, \alpha_1) \mod u^{3e}$$

by the genericity assumption.

By means of Lemma 2.2.8 we can iterate the procedure: one finds an explicit element $V_2 = \operatorname{Mat}_{\underline{e}_2, \underline{f}_2}(\operatorname{Fil}^2 \mathfrak{M}) \in \operatorname{M}_3(\overline{S})$, as in the statement of case (ii), which provides the filtration $\operatorname{Fil}^2 \mathfrak{M}$ with respect to the basis $\underline{e}_2 \stackrel{\text{def}}{=} \underline{e}_1 \cdot A_1$ and which verifies

$$\operatorname{Mat}_{e_2,f_2}(\varphi_2) = \operatorname{Diag}(\alpha_1, \alpha_2, \alpha_0).$$

(iii) This is analogous to (ii). More succinctly, it can be obtained from (ii) by duality. $\hfill \Box$

Thanks to Proposition 2.5.2 it is now straightforward to compute the Fontaine–Laffaille invariant of $\overline{\rho}$.

Proposition 2.5.6. Let $\widehat{\mathcal{M}}$, $\alpha_i \in \mathbb{F}^{\times}$ be as in Proposition 2.5.2. Let $\rho \stackrel{\text{def}}{=} T^{\mathbb{Q}_p,2}_{\text{st}}(\widehat{\mathcal{M}})$. The one of the following possibilities holds.

- (i) Fil² $\widehat{\mathcal{M}}$ is as in Case A of Proposition 2.4.10 and FL($\overline{\rho}$) = α_1^{-1} ;
- (ii) $\operatorname{Fil}^2 \widehat{\mathcal{M}}$ is as in Case B of Proposition 2.4.10 and $\operatorname{FL}(\overline{\rho}) = 0$;
- (iii) $\operatorname{Fil}^2 \widehat{\mathcal{M}}$ is as in Case C of Proposition 2.4.10 and $\operatorname{FL}(\overline{\rho}) = \infty$.

26

Proof. By twisting ρ by $\tilde{\omega}^{-a_0}$ we may assume that $a_0 = 0$. Let $a \stackrel{\text{def}}{=} a_2, b \stackrel{\text{def}}{=} a_1$.

(i) Let $\mathcal{M} \in \mathbb{F}\text{-BrMod}_{dd}^2$ be the base change of $\widehat{\mathcal{M}}$ via $S \twoheadrightarrow \overline{S}$. By Proposition 2.5.2 and Lemma 2.2.6, the ϕ -action on the $(\phi, \mathbb{F}((\underline{\varpi})))$ -module $\mathfrak{M} \stackrel{\text{def}}{=} M_{\mathbb{F}_p((\underline{\varpi}))}(\mathcal{M}^*)$ is given by

$$\operatorname{Mat}_{\underline{\mathfrak{c}}}(\phi) = \begin{pmatrix} c_{00}\underline{\varpi}^{e} & c_{01}\underline{\varpi}^{e-b} & \alpha_{2}^{-1}\underline{\varpi}^{e-a} \\ c_{10}\underline{\varpi}^{e+b} & \alpha_{1}^{-1}\underline{\varpi}^{e} \\ \alpha_{0}^{-1}\underline{\varpi}^{e+a} & & \end{pmatrix}$$

for a framed basis $\underline{\mathfrak{e}}$ (for the dual type $\omega^0 \oplus \omega^{-b} \oplus \omega^{-a}$) of \mathfrak{M} and some $c_{ij} \in \mathbb{F}$ (not those of Lemma 2.2.6). In the basis $\underline{\mathfrak{e}}' \stackrel{\text{def}}{=} \underline{\mathfrak{e}} \cdot \text{Diag}(1, \underline{\varpi}^b, \underline{\varpi}^a)$, which is $\text{Gal}(K/K_0)$ -invariant, one obtains

$$\operatorname{Mat}_{\underline{\mathfrak{e}}'}(\phi) = \begin{pmatrix} c_{00}\underline{\varpi}^e & c_{01}\underline{\varpi}^{e(b+1)} & \alpha_2^{-1}\underline{\varpi}^{e(a+1)} \\ c_{10}\underline{\varpi}^e & \alpha_1^{-1}\underline{\varpi}^{e(b+1)} \\ \alpha_0^{-1}\underline{\varpi}^e & & \end{pmatrix}$$

proving that \mathfrak{M} is the base change to $\mathbb{F}((\underline{\varpi}))$ of the $(\phi, \mathbb{F}((\underline{p})))$ -module \mathfrak{M}_0 described by

$$\operatorname{Mat}_{\underline{\mathfrak{c}}_{0}}(\phi) = \underbrace{\begin{pmatrix} c_{00} & c_{01} & \alpha_{2}^{-1} \\ c_{10} & \alpha_{1}^{-1} \\ \alpha_{0}^{-1} & & \\$$

for some basis $\underline{\mathfrak{e}}_0$ of \mathfrak{M}_0 . In the basis $\underline{\mathfrak{e}}'_0 \stackrel{\text{def}}{=} \underline{\mathfrak{e}}_0 \cdot F$ the matrix of ϕ becomes $\operatorname{Diag}(\underline{p}, \underline{p}^{b+1}, \underline{p}^{a+1})F$. We can now construct a Fontaine–Laffaille module $M \in \mathbb{F}$ - $\mathcal{FL}^{[0,p-2]}$ with Hodge–Tate weights $\{1, b+1, a+1\}$ endowed with a basis \underline{e} which is compatible with the Hodge filtration and satisfies $\operatorname{Mat}_{\underline{e}}(\phi_{\bullet}) = F$. By Lemma 2.2.7 we deduce that $\mathfrak{M}_0 \cong \mathcal{F}(M)$.

On the other hand $T^*_{cris}(M) \cong T^*_{st}(\mathcal{M}^*) \cong \overline{\rho}$ by Corollary 2.2.2. As $\overline{\rho}$ is maximally non-split, there exists a basis \underline{e}' of M compatible with the Hodge filtration and such that $\operatorname{Mat}_{\underline{e}'}(\phi_{\bullet})$ is described as in Corollary 2.1.8. In other words $\underline{e}, \underline{e}'$ are related via a lower triangular unipotent matrix $A \in \operatorname{GL}_3(\mathbb{F})$ in such a way that

$$\operatorname{Mat}_{\underline{e}'}(\phi_{\bullet}) = A \cdot F = \begin{pmatrix} \mu_0 & x & z \\ & \mu_1 & y \\ & & \mu_2 \end{pmatrix}$$

for some $x, y, \mu_i \in \mathbb{F}^{\times}$, $z \in \mathbb{F}$. It follows that

$$FL(\bar{\rho}) = \frac{\det \begin{pmatrix} x & z \\ \mu_1 & y \end{pmatrix}}{-z} = \frac{\det \begin{pmatrix} c_{01} & \alpha_2^{-1} \\ \alpha_1^{-1} \end{pmatrix}}{-\alpha_2^{-1}} = \alpha_1^{-1}.$$

(ii) The proof is similar to case (i). One checks that $\mathfrak{M} \stackrel{\text{def}}{=} M_{\mathbb{F}_p((\underline{\varpi}))}(\mathfrak{M}^*) \cong \mathfrak{M}_0 \otimes_{\mathbb{F}_p((\underline{p}))} \mathbb{F}_p((\underline{\varpi}))$, where the $(\phi, \mathbb{F}((\underline{p})))$ -module \mathfrak{M}_0 is described by

$$\underbrace{\begin{pmatrix} c_{00} & c_{01} & \alpha_0^{-1} \\ \alpha_1^{-1} & & \\ & \alpha_2^{-1} & \end{pmatrix}}_{\substack{\underline{\mathrm{def}}_F}} \mathrm{Diag}(\underline{p}, \underline{p}^{b+1}, \underline{p}^{a+1}).$$

We deduce by Lemma 2.2.7 that $\mathfrak{M}_0 \cong \mathfrak{F}(M)$ for the rank 3 Fontaine–Laffaille module $M \in \mathbb{F}$ - $\mathfrak{FL}^{[0,p-2]}$ with Hodge–Tate weights $\{1, b+1, a+1\}$ and $\operatorname{Mat}_{\underline{e}}(\phi_{\bullet}) = F$ for some basis \underline{e} of M that is compatible with the Hodge filtration. As in (i) we deduce that $\operatorname{FL}(\overline{\rho}) = 0$.

(iii) Follows by a similar argument.

Proof of Theorem 2.5.1. Let $\widehat{\mathcal{M}} \in \mathcal{O}_E$ -Mod²_{dd} be such that $T^{\mathbb{Q}_p,2}_{st}(\widehat{\mathcal{M}}) \cong \rho$ and let $\underline{\widehat{e}}$, \widehat{f} be as in the statement of Proposition 2.4.10.

In Case A, from the proof of Corollary 2.4.11 we see that $\lambda_1 = ps_0(\widehat{\alpha}_{11})$, where $\widehat{\alpha}_{ij} \in S$ are defined by $\varphi_2(\widehat{f}_j) = \sum_{i=0}^2 \widehat{\alpha}_{ij} \widehat{e}_i$, and $p\lambda_1^{-1} \in \mathcal{O}_E^{\times}$. Hence $\operatorname{red}(p\lambda_1^{-1}) = \overline{s_0(\widehat{\alpha}_{11})}^{-1} \in \mathbb{F}^{\times}$. We conclude by Proposition 2.5.6, as $\overline{s_0(\widehat{\alpha}_{11})} = \alpha_1$ in the notation of that proposition. In Case B we conclude by Proposition 2.5.6, noting that $\operatorname{red}(p\lambda_1^{-1}) = 0$, as $\operatorname{ord}_p(\lambda_1) < 1$. Case C is similar. \Box

2.6. Elimination of Galois types. The aim of this section is to perform a type elimination for a maximally non-split and generic representation $\overline{\rho}: G_{\mathbb{Q}_p} \to \mathrm{GL}_3(\mathbb{F})$ (cf. Definition 2.1.5), by means of integral *p*-adic Hodge theory. In other words we show for most tame inertial types $\tau: I_{\mathbb{Q}_p} \to \mathrm{GL}_3(E)$ (of niveau 1 or 2) that $\overline{\rho}$ cannot have a potentially crystalline lift with Hodge–Tate weights $\{-2, -1, 0\}$ and type τ .

The main results of this section (Propositions 2.6.1, 2.6.3 and 2.6.4) will be used to prove results about Serre weights in $\S4.4$.

2.6.1. **Niveau 1 types.** We first eliminate niveau 1 types. In particular, we have $K_0 = \mathbb{Q}_p, e = p - 1$, and $K = \mathbb{Q}_p(\zeta_p)$. The result is the following:

Proposition 2.6.1. Let $\rho : G_{\mathbb{Q}_p} \to \mathrm{GL}_3(\mathbb{O}_E)$ be a p-adic Galois representation, becoming semistable over K with Hodge-Tate weights $\{-2, -1, 0\}$. Assume that $\overline{\rho}$ is maximally non-split and generic as in Definition 2.1.5. Then

$$\mathrm{WD}(\rho)|_{I_{\mathbb{Q}_p}} \cong \widetilde{\omega}^i \oplus \widetilde{\omega}^j \oplus \widetilde{\omega}^k$$

where either $(i, j, k) \equiv (a_2, a_1, a_0) \mod p - 1$ or $(i, j, k) \equiv (a_2 - 1, a_1, a_0 + 1) \mod p - 1$.

Here and in the following the notation $x \equiv y \mod p - 1$ for $x, y \in \mathbb{Z}^3$ is shorthand for $x_i \equiv y_i \mod p - 1$ for all i.

Proof. After twisting we may assume $a_0 = 0$ and set $a \stackrel{\text{def}}{=} a_2$, $b \stackrel{\text{def}}{=} a_1$. Let $\widehat{\mathcal{M}}$ be a strongly divisible \mathcal{O}_E -module such that $\operatorname{T}_{\mathrm{st}}^{\mathbb{Q}_p,2}(\widehat{\mathcal{M}}) = \rho$ and let \mathcal{M} be the Breuil module obtained by base change to \overline{S} . By Proposition 2.3.5 \mathcal{M} admits a unique filtration by Breuil submodules, with graded pieces described by

$$\mathcal{M}: \mathcal{M}_2 \longrightarrow \mathcal{M}_1 \longrightarrow \mathcal{M}_0$$

(where the notation means that \mathcal{M}_2 is a submodule and \mathcal{M}_0 a quotient) and the rank one Breuil submodules \mathcal{M}_i verify $T^2_{st}(\mathcal{M}_i)|_{I_{\mathbb{Q}_p}} \cong \omega^{a_i+1}$ for i = 0, 1, 2. It follows from [EGH13], Lemma 3.3.2 that \mathcal{M}_i is of type ω^{k_i} , where $k_2 = a + 1 - \delta_2$, $k_1 = b + 1 - \delta_1$, $k_0 = 1 - \delta_0$ for some $\delta_i \in \{0, 1, 2\}$ and

$$\operatorname{Fil}^2 \mathfrak{M}_2 = u^{\delta_2 e} \mathfrak{M}_2, \ \operatorname{Fil}^2 \mathfrak{M}_1 = u^{\delta_1 e} \mathfrak{M}_1, \ \operatorname{Fil}^2 \mathfrak{M}_0 = u^{\delta_0 e} \mathfrak{M}_0.$$

On the other hand we see that $WD(\rho)|_{I_{\mathbb{Q}_p}} \cong \widetilde{\omega}^{k_0} \oplus \widetilde{\omega}^{k_1} \oplus \widetilde{\omega}^{k_2}$ (by Lemma 2.4.8) and hence det $\rho = \widetilde{\omega}^{k_0+k_1+k_2}\varepsilon^3$, as ρ has Hodge–Tate weights $\{-2, -1, 0\}$. It follows that $\delta_0 + \delta_1 + \delta_2 = 3$.

We now use Lemma 2.4.2 to conclude the proof. In what follows, we write $\overline{\rho}_{21}$ to denote the unique 2-dimensional sub-representation of $\overline{\rho}$, and, similarly, $\overline{\rho}_{10}$ to denote its unique 2-dimensional quotient. If $(\delta_2, \delta_1) = (1, 2)$ then $\overline{\rho}_{21}$ splits, by Lemma 2.4.2(ii). If $\delta_1 = 0$ then Fil² ($\mathcal{M}_1 - \mathcal{M}_0$) splits as an \overline{S} -module (as Fil² \mathcal{M}_1 is an injective \overline{S} -module, cf. Lemma A.2.1), hence $\overline{\rho}_{10}$ splits by Lemma 2.4.2(i). For the same reason, if $\delta_2 = 0$ then $\overline{\rho}_{21}$ splits.

Therefore $(\delta_0, \delta_1, \delta_2) = (1, 1, 1)$ or $(\delta_0, \delta_1, \delta_2) = (0, 1, 2)$, as claimed.

2.6.2. Niveau 2 types. We now eliminate niveau 2 types. In particular, we have $k = \mathbb{F}_{p^2}$, $K_0 = \mathbb{Q}_{p^2}$ is the degree two unramified extension of \mathbb{Q}_p , $e = p^2 - 1$, and $K = \mathbb{Q}_{p^2}(\sqrt[e]{-p})$.

Lemma 2.6.2. Let \mathcal{M} be a rank one object in \mathbb{F} -BrMod²_{dd}, with descent data from K to \mathbb{Q}_p . Then there exists a generator $m \in \mathcal{M}$ such that:

- (i) $\operatorname{Fil}^2 \mathfrak{M} = u^{r(p-1)} \mathfrak{M}$, where $0 \le r \le 2(p+1)$;
- (ii) there exists $k \in \mathbb{Z}$ verifying $\overline{k} + pr \equiv 0 \mod p + 1$ and such that \mathcal{M} is of type ω_{ϖ}^k and m is a framed basis for \mathcal{M} ;
- (iii) N(m) = 0.

Moreover, one has

$$\mathrm{T}^{2}_{\mathrm{st}}(\mathcal{M})|_{I_{\mathbb{Q}_{p}}} = \omega_{2}^{k+pr}.$$

Proof. Let $\sigma \in \operatorname{Gal}(K/\mathbb{Q}_p)$ be the lift of the arithmetic Frobenius in $\operatorname{Gal}(k/\mathbb{F}_p)$ that fixes ϖ , hence $\sigma^2 = 1$. Then σ interchanges the two primitive idempotents ϵ_0 , ϵ_1 , and one has $\sigma \circ g = g^p \circ \sigma$ for all $g \in \operatorname{Gal}(K/K_0)$.

By [EGH13], Lemma 3.3.2 there is a generator $m \in \mathcal{M}$ such that

- (i) $\epsilon_i \operatorname{Fil}^2 \mathfrak{M} = u^{r_i e} \epsilon_i \mathfrak{M}$ for i = 0, 1;
- (ii) $\widehat{g}(m) = (\omega_{\varpi}(g)^{k_0} \otimes 1)\epsilon_0 m + (\omega_{\varpi}(g)^{k_1} \otimes 1)\epsilon_1 m$ for all $g \in \operatorname{Gal}(K/K_0)$,

where the integers r_i, k_i verify $0 \le r_i \le 2e, k_i \equiv p(k_{i-1} + r_{i-1}) \mod e$.

As σ exchanges the idempotents $\epsilon_i \in k \otimes_{\mathbb{F}_p} \mathbb{F}$ we deduce that $r \stackrel{\text{def}}{=} r_0 = r_1$. Let us consider $\mathcal{M}/(u \cdot \mathcal{M})$. It is a rank one module over $k \otimes_{\mathbb{F}_p} \mathbb{F}$, endowed with a semilinear action of $\operatorname{Gal}(K/\mathbb{Q}_p)$. In particular $\sigma \overline{m} = \mu \overline{m}$ for some $\mu \in (k \otimes_{\mathbb{F}_p} \mathbb{F})^{\times}$ verifying $\mu \sigma(\mu) = 1$. We therefore deduce, by (ii) above, that for all $g \in \operatorname{Gal}(K/K_0)$,

$$\begin{aligned} \widehat{\sigma} \circ \widehat{g}(\overline{m}) &= \widehat{\sigma} \left((\omega_{\overline{\omega}}^{k_0}(g) \otimes 1) \epsilon_0 \overline{m} + (\omega_{\overline{\omega}}^{k_1}(g) \otimes 1) \epsilon_1 \overline{m} \right) \\ &= \left[(\omega_{\overline{\omega}}^{pk_0}(g) \otimes 1) \epsilon_1 \mu \overline{m} + (\omega_{\overline{\omega}}^{pk_1}(g) \otimes 1) \epsilon_0 \mu \overline{m} \right] \end{aligned}$$

and

$$\widehat{g^p} \circ \widehat{\sigma}(\overline{m}) = \mu \left[(\omega_{\varpi}^{pk_0}(g) \otimes 1) \epsilon_0 \overline{m} + (\omega_{\varpi}^{pk_1}(g) \otimes 1) \epsilon_1 \overline{m} \right]$$

As $\sigma \circ g = g^p \circ \sigma$ we conclude that $k \stackrel{\text{def}}{=} k_0 \equiv k_1 \mod e$. We finally obtain $k \equiv p(k+r) \mod e$, so that $r \equiv 0 \mod p - 1$ and $k + \frac{pr}{p-1} \equiv 0 \mod p + 1$. The last statement on $T^2_{\text{st}}(\mathcal{M})$ follows again from [EGH13], Lemma 3.3.2 and the previous results on k_i, r_i .

For the remainder of this section, all Breuil modules $\mathcal{M} \in \mathbb{F}\text{-BrMod}_{dd}^2$ have descent data from K to \mathbb{Q}_p .

Proposition 2.6.3. Let $\rho : G_{\mathbb{Q}_p} \to \operatorname{GL}_3(\mathbb{O}_E)$ be a p-adic Galois representation, becoming semistable over K with Hodge–Tate weights $\{-2, -1, 0\}$. Assume that $\overline{\rho}$ is maximally non-split and generic as in Definition 2.1.5 and that the inertial type of ρ has niveau two, i.e. ρ does not become semistable over $\mathbb{Q}_p(\zeta_p)$. Then

$$\mathrm{WD}(\rho)|_{I_{\mathbb{Q}_p}} \cong \widetilde{\omega}^x \oplus \widetilde{\omega}_2^y \oplus \widetilde{\omega}_2^{py}$$

where $x, y \in \mathbb{Z}, y \not\equiv 0 \mod (p+1)$ and the pair $(x, y) \in \mathbb{Z}^2$ verifies one of the following possibilities:

- (i) $x \equiv a_0 \delta \mod p 1$ and $y \equiv a_2 + pa_1 + \delta \epsilon(p 1) \mod e$;
- (ii) $x \equiv a_1 \delta \mod p 1$ and $y \equiv a_2 + pa_0 + \delta \epsilon(p 1) \mod e$;
- (iii) $x \equiv a_2 \delta \mod p 1$ and $y \equiv a_1 + pa_0 + \delta \epsilon(p 1) \mod e$,

where $\epsilon \in \{0, 1\}$, $\delta \in \mathbb{Z}$ are such that $\delta + \epsilon \in \{0, 1\}$.

Proof. As in the proof of Proposition 2.6.1 we may assume $a_0 = 0$ and set $a \stackrel{\text{def}}{=} a_2$, $b \stackrel{\text{def}}{=} a_1$. Let $\widehat{\mathcal{M}}$ be a strongly divisible \mathcal{O}_E -module such that $\operatorname{T}_{\operatorname{st}}^{\mathbb{Q}_p,2}(\widehat{\mathcal{M}}) \cong \rho$; let \mathcal{M} be the Breuil module obtained by base change to \overline{S} . Then \mathcal{M} is of type $\omega_{\varpi}^{x(p+1)} \oplus \omega_{\varpi}^y \oplus \omega_{\varpi}^{py}$ for some $x, y \in \mathbb{Z}, y \not\equiv 0 \mod p + 1$ (by Lemma 2.4.8). As ρ has Hodge–Tate weights $\{-2, -1, 0\}$ we moreover have det $\rho = \widetilde{\omega}^{x+y} \varepsilon^3$.

By Proposition 2.3.5, ${\mathcal M}$ has a unique filtration by Breuil submodules with graded pieces described by

$$\mathcal{M}: \mathcal{M}_2 - \mathcal{M}_1 - \mathcal{M}_0,$$

where $T_{st}^2(\mathcal{M}_i)|_{I_{\mathbb{Q}_p}} = \omega^{a_i+1}$. We conclude by Lemma 2.6.2 that one of the following three possibilities holds (up to interchanging y and py):

- (i) the Breuil modules $(\mathcal{M}_2, \mathcal{M}_1, \mathcal{M}_0)$ are respectively of type $(\omega_{\varpi}^y, \omega_{\varpi}^{py}, \omega_{\varpi}^{x(p+1)})$, where $x \equiv 1 - \delta \mod (p-1)$, $y \equiv (a+1)(p+1) - rp \mod e$ and $py \equiv (p+1)(b+1) - sp \mod e$;
- (ii) the Breuil modules $(\mathcal{M}_2, \mathcal{M}_1, \mathcal{M}_0)$ are respectively of type $(\omega_{\varpi}^y, \omega_{\varpi}^{x(p+1)}, \omega_{\varpi}^{py})$, where $x \equiv b + 1 - \delta \mod (p-1)$, $y \equiv (a+1)(p+1) - rp \mod e$ and $py \equiv (p+1) - sp \mod (p-1)$;
- (iii) the Breuil modules $(\mathcal{M}_2, \mathcal{M}_1, \mathcal{M}_0)$ are respectively of type $(\omega_{\varpi}^{x(p+1)}, \omega_{\varpi}^y, \omega_{\varpi}^{py})$, where $x \equiv a + 1 - \delta \mod (p-1)$, $y \equiv (b+1)(p+1) - rp \mod e$ and $py \equiv (p+1) - sp \mod e$,

where the integers δ, r, s verify $\delta \in \{0, 1, 2\}, 0 \le r, s \le 2(p+1) \text{ and } y+pr, y+s \equiv 0 \mod p+1$. Moreover, $r \not\equiv 0 \mod p+1$, as ρ does not become semistable over $\mathbb{Q}_p(\zeta_p)$.

Assume we are in case (i). We deduce that $pr - s \equiv (a - b)(p + 1) \mod e$ hence $r + s \equiv 0 \mod p + 1$. Provided the restrictions on r, s, we obtain $r + s = \alpha(p+1)$ for some $\alpha \in \{1, 2, 3\}$ and hence $r \equiv \alpha + (a - b) \mod p - 1$. Considering the condition det $\rho|_{I_{\mathbb{Q}_p}} \equiv \omega^{x+y}\omega^3$, we see that the integers α , δ verify moreover the relation $\delta + \alpha = 3$ (as p > 3).

As 2 < a - b < p - 3, the condition 0 < r, s < 2(p + 1) implies that $r \in \{1 + (a - b), 2 + (a - b), p + 1 + (a - b), p + 2 + (a - b)\}$ and hence $(x, y) \in \{(-1, a + pb + 1), (0, a + pb - (p - 1)), (0, a + pb), (1, a + pb - p)\}$. This concludes the analysis of case (i).

Cases (ii) and (iii) are strictly analogous. We remark that in case (ii), (iii) we obtain the condition $r \equiv \alpha + \sigma$ for $\sigma \in \{a, b\}$; as $2 < \sigma < p - 3$ this provides us with $r \in \{\sigma + 1 + \epsilon + \zeta p : \epsilon, \zeta \in \{0, 1\}\}$.

Proposition 2.6.4. Let $\rho : G_{\mathbb{Q}_p} \to \operatorname{GL}_3(\mathbb{O}_E)$ be a potentially semistable p-adic Galois representation with Hodge–Tate weights $\{-2, -1, 0\}$. Assume that $\overline{\rho}$ is maximally non-split and generic as in Definition 2.1.5.

(i) If
$$\operatorname{WD}(\rho)|_{I_{\mathbb{Q}_p}} \cong \widetilde{\omega}^{a_0} \oplus \widetilde{\omega}_2^{a_2+1+p(a_1-1)} \oplus \widetilde{\omega}_2^{a_1-1+p(a_2+1)}$$
, then $\operatorname{FL}(\overline{\rho}) = \infty$.
(ii) If $\operatorname{WD}(\rho)|_{I_{\mathbb{Q}_p}} \cong \widetilde{\omega}^{a_2} \oplus \widetilde{\omega}_2^{a_0-1+p(a_1+1)} \oplus \widetilde{\omega}_2^{a_1+1+p(a_0-1)}$, then $\operatorname{FL}(\overline{\rho}) = 0$.

In order to prove Proposition 2.6.4 we follow the procedure used in Section 2.5: we first diagonalize the Frobenius action on the Breuil module \mathcal{M} and, subsequently, we compute $FL(\overline{\rho})$ by extracting the Fontaine–Laffaille module from the datum of \mathcal{M} (via the results of Section 2.2).

The proof will occupy the remainder of this section.

Proof. We first note that (ii) follows from (i), replacing ρ by ρ^{\vee} and (a_0, a_1, a_2) by $(-a_2, -a_1, -a_0)$ and keeping in mind Remark 2.1.13.

We will now explain the proof of (i). By twisting we may assume that $a_0 = 0$. We set $a \stackrel{\text{def}}{=} a_2$, $b \stackrel{\text{def}}{=} a_1$ and define:

$$k_1 \stackrel{\text{def}}{=} a + 1 + p(b-1), \qquad k_2 \stackrel{\text{def}}{=} b - 1 + p(a+1),$$

$$r_1 \stackrel{\text{def}}{=} a - b + 2, \qquad r_2 \stackrel{\text{def}}{=} 2p - (a-b),$$

so that $\tau \stackrel{\text{def}}{=} \text{WD}(\rho)|_{I_{\mathbb{Q}_p}} = \widetilde{\omega}^0 \oplus \widetilde{\omega}_2^{k_1} \oplus \widetilde{\omega}_2^{k_2}$. Note in particular that $0 < k_1, k_2 < e$ as well as the following relations, which will be useful for the computations below:

$$r_1(p-1) < e,$$
 $(p-1)(r_1+r_2) = 2e,$
 $k_1 + r_1(p-1) = k_2,$ $k_2 + r_2(p-1) = k_1 + 2e.$

Let $\widehat{\mathcal{M}} \in \mathcal{O}_E$ - $\mathrm{Mod}_{\mathrm{dd}}^2$ be a strongly divisible \mathcal{O}_E -module such that $\mathrm{T}_{\mathrm{st}}^{\mathbb{Q}_P,2}(\widehat{\mathcal{M}}) = \rho$ and let $\mathcal{M} \in \mathbb{F}$ - $\mathrm{Br}\mathrm{Mod}_{\mathrm{dd}}^2$ be the base change of $\widehat{\mathcal{M}}$ via $S \twoheadrightarrow \overline{S}$. In particular by Lemma 2.4.8, \mathcal{M} is of type $\omega_{\varpi}^0 \oplus \omega_{\varpi}^{k_1} \oplus \omega_{\varpi}^{k_2}$. The first step of the proof is to describe the filtration and the Frobenius action on \mathcal{M} .

Claim 2.6.5. There is a framed basis \underline{e} of M and a framed system of generators \underline{f} for Fil² M such that

$$\operatorname{Mat}_{\underline{e},\underline{f}}(\operatorname{Fil}^{2} \mathfrak{M}) = \begin{pmatrix} u^{e} & & \\ & u^{r_{1}(p-1)} & z \\ & yu^{2e-k_{2}} & & u^{r_{2}(p-1)} \end{pmatrix}, \quad \operatorname{Mat}_{\underline{e},\underline{f}}(\varphi_{2}) = \begin{pmatrix} \lambda & & \\ & \mu & \\ & & \nu \end{pmatrix}$$

where $y, z \in k \otimes_{\mathbb{F}_p} \mathbb{F}$ and $\lambda, \mu, \nu \in (k \otimes_{\mathbb{F}_p} \mathbb{F})^{\times}$.

Proof. From Proposition 2.3.5, the proof of Proposition 2.6.3, and genericity, we see that \mathcal{M} has a unique filtration by Breuil submodules

$$\mathcal{M}: \mathcal{M}_1 \longrightarrow \mathcal{M}_2 \longrightarrow \mathcal{M}_0,$$

where \mathcal{M}_1 (resp. \mathcal{M}_2 , resp. \mathcal{M}_0) is a rank one Breuil module of type $\omega_{\varpi}^{k_1}$ (resp. $\omega_{\varpi}^{k_2}$, resp. ω_{ϖ}^0), and filtration described by $\operatorname{Fil}^2 \mathcal{M}_1 = u^{r_1(p-1)} \mathcal{M}_1$ (resp. $\operatorname{Fil}^2 \mathcal{M}_2 = u^{r_2(p-1)} \mathcal{M}_1$, resp. $\operatorname{Fil}^2 \mathcal{M}_0 = u^e \mathcal{M}_0$).

We therefore deduce the existence of a framed basis $\underline{e}_0 \stackrel{\text{def}}{=} (e_0, e_1, e_2)$, and a framed system of generators $\underline{f}_0 = (f_0, f_1, f_2)$ of Fil² \mathcal{M} such that

$$V_{0} \stackrel{\text{def}}{=} \operatorname{Mat}_{\underline{e}_{0}, \underline{f}_{0}} \left(\operatorname{Fil}^{2} \mathcal{M} \right) = \begin{pmatrix} u^{e} \\ x_{0} u^{e-k_{1}} & u^{r_{1}(p-1)} & z_{0} \\ y'_{0} u^{e-k_{2}} & u^{r_{2}(p-1)} \end{pmatrix},$$
$$A_{0} \stackrel{\text{def}}{=} \operatorname{Mat}_{\underline{e}_{0}, \underline{f}_{0}} (\varphi_{2}) = \begin{pmatrix} \lambda_{0} & \\ \alpha_{0} u^{e-k_{1}} & \mu_{0} & \gamma_{0} u^{r_{1}(p-1)} \\ \beta_{0} u^{e-k_{2}} & \nu_{0} \end{pmatrix},$$

where $x_0, y'_0, z_0, \alpha_0, \beta_0, \gamma_0 \in \overline{S}_{\omega_{\varpi}^0}$ and $\lambda_0, \mu_0, \nu_0 \in \overline{S}_{\omega_{\varpi}^0}^{\times}$.

Claim 2.6.6. We have $y'_0 = u^e y_0$ for some $y_0 \in \overline{S}_{\omega_{\omega}^0}$.

Proof. Let $\mathbb{N} : \mathbb{M}_2 \longrightarrow \mathbb{M}_0$ denote the rank two quotient of \mathbb{M} . By the previous analysis of \mathbb{M} , we deduce that there is a basis $(\overline{e}_0, \overline{e}_2)$ of \mathbb{N} and a pair $(\overline{f}_0, \overline{f}_1)$ of elements of Fil² \mathbb{N} such that

$$\operatorname{Fil}^{2} \mathbb{N} = \left\langle \overline{f}_{0} \stackrel{\text{def}}{=} \left(\begin{array}{c} u^{e} \\ y_{0}^{\prime} u^{e-k_{2}} \end{array} \right), \ \overline{f}_{1} \stackrel{\text{def}}{=} \left(\begin{array}{c} 0 \\ u^{r_{2}(p-1)} \end{array} \right) \right\rangle_{\overline{S}}$$

(in terms of the basis $(\overline{e}_0, \overline{e}_2)$). If $y'_0 \neq 0$ modulo u, then it is easy to see that $u^{2e}\overline{e}_0 \notin$ Fil² N, contradicting the Breuil module axioms. Therefore the claim follows. \Box

We now complete the procedure to diagonalize the Frobenius action. An elementary computation shows that there exist $y_1, z_1, \zeta_1 \in \overline{S}_{\omega_{\varpi}^0}$ such that the following equality holds:

$$A_0 \begin{pmatrix} u^e & & \\ & u^{r_1(p-1)} & z_1 \\ y_1 u^{2e-k_2} & & u^{r_2(p-1)} \end{pmatrix} \equiv V_0 \underbrace{\begin{pmatrix} \lambda_0 & & \\ & \zeta_1 u^{e-k_2} & \mu_0 \\ & & \nu_0 \end{pmatrix}}_{\substack{\text{def} \\ = B_0}} \text{ modulo } u^{3e}.$$

Indeed, it suffices to take $y_1 \equiv \nu_0^{-1}(\lambda_0 y_0 - \beta_0) \mod u^{2e}$, $z_1 \equiv \mu_0^{-1}(\nu_0 z_0 - \gamma_0 u^{2e}) \mod u^{3e}$ and $\zeta_1 \equiv -\lambda_0 x_0 + (\alpha_0 + y_1 \gamma_0) u^e \mod u^{3e}$.

We use now Lemma 2.2.8 to deduce that, in the basis $\underline{e}_1 \stackrel{\text{def}}{=} \underline{e}_0 \cdot A_0$, the filtration and the Frobenius action can respectively be described as:

$$V_1 \stackrel{\text{def}}{=} \begin{pmatrix} u^e & & \\ & u^{r_1(p-1)} & z_1 \\ y_1 u^{2e-k_2} & & u^{r_2(p-1)} \end{pmatrix}, \quad A_1 \stackrel{\text{def}}{=} \begin{pmatrix} \varphi(\lambda_0) & & \\ \varphi(\zeta_1) u^{p(e-k_2)} & \varphi(\mu_0) & \\ & & \varphi(\nu_0) \end{pmatrix},$$

where $y_1, z_1 \in \overline{S}_{\omega_{\varpi}^0}$ and $A_1 = \varphi(B_0)$.

We write $\underline{f_1} \stackrel{\text{def}}{=} \underline{e_1} \cdot V_1$. As $r_1(p-1) < e$, $(3e-k_2) - r_2(p-1) = e-k_1 > r_1(p-1)$, we can find a matrix in $\operatorname{GL}_3^{\Box}(\overline{S})$ of the form

$$C \stackrel{\text{def}}{=} \begin{pmatrix} 1 \\ c_{10}u^{e-k_1-r_1(p-1)} & 1 & c_{12}u^{e-r_1(p-1)} \\ c_{20}u^{3e-k_2-r_2(p-1)} & 1 \end{pmatrix}$$

32

such that, relative to \underline{e}_1 and $\underline{f}'_1 \stackrel{\text{def}}{=} \underline{f}_1 \cdot C$, the filtration and the Frobenius action can be described by

$$V_1' \stackrel{\text{def}}{=} \begin{pmatrix} u^e & & \\ & u^{r_1(p-1)} & z_1' \\ y_1' u^{2e-k_2} & & u^{r_2(p-1)} \end{pmatrix}, \ A_1' \stackrel{\text{def}}{=} \begin{pmatrix} \varphi(\lambda_0) & & \\ a_{10}' u^{p(e-k_2)} & \varphi(\mu_0) & a_{12}' u^{p(e-r_1(p-1))} \\ a_{20}' u^{p(e-k_1)} & & \varphi(\nu_0) \end{pmatrix},$$

where $y'_1, z'_1 \in k \otimes_{\mathbb{F}_p} \mathbb{F}$ and $a'_{ij} \in \overline{S}_{\omega_{\infty}^0}$. We can write $a'_{10}u^{p(e-k_2)} = \alpha'_1 u^{e-k_1}$, $a'_{20}u^{p(e-k_1)} = \beta'_1 u^{e-k_2}$, $a'_{12}u^{p(e-r_1(p-1))} = \gamma'_1 u^{r_1(p-1)}$ with $\alpha'_1, \beta'_1, \gamma'_1 \in u^{2e}\overline{S}_{\omega_{\infty}^0}$ by genericity. Writing $\lambda'_1 \stackrel{\text{def}}{=} \varphi(\lambda_0)$, $\mu'_1 \stackrel{\text{def}}{=} \varphi(\mu_0)$, $\nu'_1 \stackrel{\text{def}}{=} \varphi(\nu_0)$, which all lie in $k \otimes_{\mathbb{F}_p} \mathbb{F}$, we see that (V'_1, A'_1) is of the same form as (V_0, A_0) , so we can apply the above diagonalization procedure to obtain (V_2, A_2) as in the statement of the claim. \Box

We now compute the Fontaine–Laffaille invariant of the Galois representation associated to \mathcal{M} .

Claim 2.6.7. With the hypotheses and notation of Claim 2.6.5 we have

$$\operatorname{FL}(\operatorname{T}^2_{\operatorname{st}}(\mathcal{M})) = \infty.$$

Proof. Consider the $(\phi, k \otimes_{\mathbb{F}_p} \mathbb{F}((\underline{\varpi})))$ -module $\mathfrak{M} \stackrel{\text{def}}{=} M_{k((\underline{\varpi}))}(\mathcal{M}^*)$. From Claim 2.6.5 and Lemma 2.2.6 \mathfrak{M} is endowed with a framed basis $\underline{\mathfrak{e}}$ (for the type $\omega_{\overline{\omega}}^0 \oplus \omega_{\overline{\omega}}^{-k_1} \oplus \omega_{\overline{\omega}}^{-k_2}$) such that

$$\operatorname{Mat}_{\underline{\mathfrak{c}}}(\phi) = \begin{pmatrix} \lambda^{-1} \underline{\varpi}^{e} & y\nu^{-1} \underline{\varpi}^{2e-k_{2}} \\ \nu^{-1} \underline{\varpi}^{r_{2}(p-1)} & z\mu^{-1} \\ & \mu^{-1} \underline{\varpi}^{r_{1}(p-1)} \end{pmatrix},$$

where $y, z \in k \otimes_{\mathbb{F}_p} \mathbb{F}, \lambda, \mu, \nu \in (k \otimes_{\mathbb{F}_p} \mathbb{F})^{\times}$.

By considering the Gal(K/K_0)-invariant basis $\underline{\mathfrak{e}}' \stackrel{\text{def}}{=} \underline{\mathfrak{e}} \cdot \text{Diag}(1, \underline{\varpi}^{k_1}, \underline{\varpi}^{k_2})$ one deduces that $\mathfrak{M} \cong \mathfrak{M}_0 \otimes_{\mathbb{F}(\underline{p})} \mathbb{F}((\underline{\varpi}))$, where the $(\phi, k \otimes_{\mathbb{F}_p} \mathbb{F}(\underline{p}))$ -module \mathfrak{M}_0 is described (in some basis $\underline{\mathfrak{e}}_0$) by

$$(2.6.8) \qquad \operatorname{Mat}_{\underline{\mathfrak{c}}_{0}}(\phi) = \left(\begin{array}{cc} \lambda^{-1} & y\nu^{-1} \\ & \nu^{-1} & z\mu^{-1} \\ & & \mu^{-1} \end{array}\right) \operatorname{Diag}(\underline{p}, \underline{p}^{b+1}, \underline{p}^{a+1}).$$

From (2.6.8), Lemma 2.2.7 (and a change of basis over $k \otimes_{\mathbb{F}_p} \mathbb{F}$ as in the proof of Proposition 2.5.6) we deduce that the Frobenius action on the Fontaine–Laffaille module M_k associated to $\overline{\rho}|_{G_{K_0}}$ is given by

$$F_k \stackrel{\text{def}}{=} \operatorname{Mat}_{\underline{e}}(\phi_{\bullet}) = \varphi \begin{pmatrix} \lambda^{-1} & y\nu^{-1} \\ & \nu^{-1} & z\mu^{-1} \\ & & \mu^{-1} \end{pmatrix} \in \operatorname{GL}_3(k \otimes_{\mathbb{F}_p} \mathbb{F})$$

for some basis \underline{e} of M_k compatible with the Hodge filtration on M_k (and φ denotes the relative Frobenius on $k \otimes_{\mathbb{F}_p} \mathbb{F}$). By Corollary 2.2.2 we have $T^*_{\operatorname{cris}}(M_k) \cong \overline{\rho}|_{G_{K_0}}$. In particular, we have $M_k \cong M \otimes_{\mathbb{F}_p} k$ for some Fontaine–Laffaille module $M \in$ \mathbb{F} - $\mathcal{FL}^{[p-2]}$ such that $T^*_{\operatorname{cris}}(M) \cong \overline{\rho}$ as $G_{\mathbb{Q}_p}$ -representations.

As p > 2, $\overline{\rho}|_{G_{K_0}}$ is still maximally non-split, hence M_k is endowed with a unique filtration $0 \subseteq M_{k,0} \subseteq M_{k,1} \subseteq M_{k,2} = M_k$, obtained from the cosocle filtration on $\overline{\rho}|_{G_{K_0}}$. By unicity, we have $M_{k,i} = M_i \otimes_{\mathbb{F}_p} k$ for all i = 0, 1, 2, where $0 \subseteq M_0 \subseteq M_1 \subseteq M_2 \subseteq M_1 \subseteq M_2 = M$ is the filtration on M obtained via T^*_{cris} from the cosocle filtration on $\overline{\rho}$.

As Fil¹ $M_k \cap M_{k,0}$, Fil^{b+1} $M_k \cap M_{k,1}$, Fil^{a+1} $M_k \cap M_{k,2}$ are all free of rank one over $k \otimes_{\mathbb{F}_p} \mathbb{F}$ it follows that any change of basis of M_k preserving both the Hodge and the submodule filtration needs to be diagonal.

Therefore, by letting $F \in \operatorname{GL}_3(\mathbb{F})$ denote the matrix of the Frobenius action on M, we deduce that $F_{02} = 0$ if and only if $(F_k)_{02} = 0$, which shows that $\operatorname{FL}(\overline{\rho}) = \infty$, as claimed.

From the Claims 2.6.5, 2.6.7, the proof of Proposition 2.6.4(i) follows. \Box

3. The local automorphic side

In this section we analyze the GL_3 -side. In particular, we establish Proposition G of the introduction. We have [Jan03] as a main reference for the notation and terminology.

Let $G \stackrel{\text{def}}{=} \operatorname{GL}_{n/\mathbb{Z}_p}$ and let T be the maximal split torus consisting of diagonal matrices. We let Φ denote the set of roots with respect to T and $B \supseteq T$ the Borel subgroup of upper-triangular matrices. We write $\Delta \subseteq \Phi$ for the simple roots associated to the pair (B,T) and $\langle \cdot, \cdot \rangle$ for the natural pairing on $X^*(T) \times X_*(T)$. The character and cocharacter groups are identified with \mathbb{Z}^n in the usual way. We let \leq denote the dominance order on $X^*(T)$. Finally, we let $\overline{G}, \overline{B}, \ldots$ denote the base change of G, B, \ldots via $\mathbb{Z}_p \twoheadrightarrow \mathbb{F}_p$.

The Weyl group W_G of G is canonically isomorphic to the Weyl group of \overline{G} . We write $w_0 \in W_G$ for the longest element.

For any dominant character $\lambda \in X^*(\overline{T})$ let

$$H^{0}(\lambda) \stackrel{\mathrm{def}}{=} \left(\mathrm{Ind}_{\overline{B}}^{\overline{G}} w_{0} \lambda \right)^{\mathrm{alg}} \otimes_{\mathbb{F}_{p}} \mathbb{F}$$

be the associated dual Weyl module. It is an algebraic representation of \overline{G} (or more precisely of $\overline{G}_{/\mathbb{F}}$) and we write $F(\lambda) \stackrel{\text{def}}{=} \operatorname{soc}_{\overline{G}} (H^0(\lambda))$ for its irreducible socle. If the weight λ is moreover *p*-restricted, i.e. if $0 \leq \langle \lambda, \alpha^{\vee} \rangle \leq p-1$ for all $\alpha \in \Delta$, then $F(\lambda)$ is irreducible as $\overline{G}(\mathbb{F}_p)$ -representation (see for example [Her09], Corollary 3.17). If Q is an algebraic representation of \overline{G} and $\nu \in X^*(T)$ we denote by Q_{ν} the ν -weight space for the action of \overline{T} on Q, as usual.

Note that if n = 1 and $a \in \mathbb{Z}$ then $F(a) \cong \omega^a$, where ω is seen as a character of $\operatorname{GL}_1(\mathbb{F}_p)$ via the Artin reciprocity map (so ω is the inclusion $\mathbb{F}_p^{\times} \to \mathbb{F}^{\times}$).

Let us specialize to the case n = 3. We set $\alpha_1 \stackrel{\text{def}}{=} (1, -1, 0), \alpha_2 \stackrel{\text{def}}{=} (0, 1, -1)$, so that $\Delta = \{\alpha_1, \alpha_2\}$, and we fix the following lifts $\dot{s}_1, \dot{s}_2 \in G(\mathbb{Z}_p)$ of the simple reflections $s_1, s_2 \in W_G$ corresponding to α_1, α_2 :

$$\dot{s}_1 \stackrel{\text{def}}{=} \left(\begin{array}{cc} 1 \\ 1 \\ & 1 \end{array} \right), \quad \dot{s}_2 \stackrel{\text{def}}{=} \left(\begin{array}{cc} 1 \\ & 1 \end{array} \right).$$

Note that $\dot{w}_0 \stackrel{\text{def}}{=} \dot{s}_1 \dot{s}_2 \dot{s}_1$ is then a lift of $w_0 \in W_G$.

Finally, for each $\alpha \in \Phi$ we denote by U_{α} the associated root subgroup and by $u_{\alpha} : \mathbb{G}_a \xrightarrow{\sim} U_{\alpha}$ an isomorphism as in [Jan03], II.1.2. Explicitly we will take:

$$u_{\alpha_1}(x) = \begin{pmatrix} 1 & x \\ & 1 & \\ & & 1 \end{pmatrix}, \ u_{\alpha_2}(x) = \begin{pmatrix} 1 & 1 & x \\ & 1 & x \\ & & & 1 \end{pmatrix}, \ u_{\alpha_1 + \alpha_2}(x) = \begin{pmatrix} 1 & x & x \\ & 1 & x \\ & & & 1 \end{pmatrix}.$$

Let $K \stackrel{\text{def}}{=} G(\mathbb{Z}_p)$. (This should not cause any confusion with the field K of Sections 2 and A.) We write I for the Iwahori subgroup of K which is the preimage of $\overline{B}(\mathbb{F}_p)$ under the reduction map $K \to \overline{G}(\mathbb{F}_p)$ and similarly $I_1 \leq I$ for the pro-p

34

Iwahori subgroup which is the preimage of $\overline{U}(\mathbb{F}_p)$. If V is a representation of K over \mathcal{O}_E and $a_i \in \mathbb{Z}$ we write

$$V^{I,(a_2,a_1,a_0)} \stackrel{\text{def}}{=} \operatorname{Hom}_I(\mathcal{O}_E(\widetilde{\omega}^{a_2} \otimes \widetilde{\omega}^{a_1} \otimes \widetilde{\omega}^{a_0}), V),$$

the largest \mathcal{O}_E -submodule of V on which I acts via $\widetilde{\omega}^{a_2} \otimes \widetilde{\omega}^{a_1} \otimes \widetilde{\omega}^{a_0}$. In particular, this makes sense when V is killed by ϖ_E . Also, if V is a representation of $\overline{G}(\mathbb{F}_p)$ over \mathbb{F} we write $V^{\overline{T}(\mathbb{F}_p),(a_2,a_1,a_0)}$ to denote the largest subspace on which $\overline{T}(\mathbb{F}_p)$ acts as $F(a_2) \otimes F(a_1) \otimes F(a_0)$, and we note that $V^{I,(a_2,a_1,a_0)} \subseteq V^{\overline{T}(\mathbb{F}_p),(a_2,a_1,a_0)}$.

3.1. Group algebra operators. We introduce certain mod-p group algebra operators for $\overline{G}(\mathbb{F}_p)$ and study their effect on extensions of $\overline{G}(\mathbb{F}_p)$ -representations.

Let $(a_2, a_1, a_0) \in \mathbb{Z}^3$ be a triple verifying the genericity condition (2.1.6). Note that in this case the weight $(a_2, a_1, a_0) \in \mathbb{Z}^3$ is in particular *p*-restricted. We define the following elements of $\mathbb{F}[\overline{G}(\mathbb{F}_p)]$:

(3.1.1)
$$S \stackrel{\text{def}}{=} \sum_{x,y,z \in \mathbb{F}_p} x^{p-(a_2-a_0)} z^{p-(a_1-a_0)} \begin{pmatrix} 1 & x & y \\ & 1 & z \\ & & 1 \end{pmatrix} \dot{w}_0,$$
$$S' \stackrel{\text{def}}{=} \sum_{x,y,z \in \mathbb{F}_p} x^{p-(a_2-a_1)} z^{p-(a_2-a_0)} \begin{pmatrix} 1 & x & y \\ & 1 & z \\ & & 1 \end{pmatrix} \dot{w}_0.$$

In other words, $S = \sum_{x,y,z \in \mathbb{F}_p} x^{p-(a_2-a_0)} z^{p-(a_1-a_0)} u_{\alpha_2}(z) u_{\alpha_1+\alpha_2}(y) u_{\alpha_1}(x) \dot{w}_0$ and similarly for S'

ilarly for S'.

The following property of the operators S, S' will be crucial for us.

Proposition 3.1.2. Let $(a_2, a_1, a_0) \in \mathbb{Z}^3$ be a triple satisfying (2.1.6). Consider the associated operators $S, S' \in \mathbb{F}[\overline{G}(\mathbb{F}_p)]$.

(i) There is a unique non-split extension of irreducible $\overline{G}(\mathbb{F}_p)$ -representations

 $0 \to F(a_2 - 1, a_1, a_0 + 1) \to V \to F(a_1 + p - 1, a_2, a_0) \to 0$

and S induces an isomorphism $S: V^{I,(a_1,a_2,a_0)} \xrightarrow{\sim} V^{I,(a_2-1,a_1,a_0+1)}$ of one-dimensional vector spaces.

(ii) There is a unique non-split extension of irreducible $\overline{G}(\mathbb{F}_p)$ -representations

$$0 \to F(a_2 - 1, a_1, a_0 + 1) \to V \to F(a_2, a_0, a_1 - p + 1) \to 0$$

and S' induces an isomorphism $S': V^{I,(a_2,a_0,a_1)} \xrightarrow{\sim} V^{I,(a_2-1,a_1,a_0+1)}$ of one-dimensional vector spaces.

More generally, one could ask (also for more general groups):

Question 3.1.3. Suppose that V is an $\overline{G}(\mathbb{F}_p)$ -representation that is generated by an I-eigenvector $v \in V^{I,(b_2,b_1,b_0)}$. Given an I-eigenvector $v' \in V^{I,(c_2,c_1,c_0)}$, find an explicit element of $\mathbb{F}[\overline{G}(\mathbb{F}_p)]$ sending v to v'.

By Frobenius reciprocity, V is a quotient of the principal series $\operatorname{Ind}_{\overline{B}(\mathbb{F}_p)}^{\overline{G}(\mathbb{F}_p)}(F(b_2) \otimes F(b_1) \otimes F(b_0))$, and an interesting special case of the above question is when V has irreducible socle that is moreover generated by v'. Proposition 3.1.2 answers this in some instances. (See also the proof of [BD14], Proposition 2.6.1 for GL₂.)

The proof of Proposition 3.1.2 requires a certain amount of preliminaries and will occupy the rest of this section.

Let $\overline{P} \supseteq \overline{B}$ denote the standard parabolic associated to the simple root α_1 . We consider the equivariant monomorphism

$$\tau \stackrel{\text{def}}{=} \operatorname{Ind}_{\overline{P}(\mathbb{F}_p)}^{\overline{G}(\mathbb{F}_p)} \left(F(a_1 + p - 1, a_2) \otimes F(a_0) \right) \hookrightarrow \operatorname{Ind}_{\overline{B}(\mathbb{F}_p)}^{\overline{G}(\mathbb{F}_p)} \left(F(a_2) \otimes F(a_1) \otimes F(a_0) \right) \stackrel{\text{def}}{=} \sigma$$

obtained by parabolically inducing the injection $F(a_1+p-1, a_2) \hookrightarrow \operatorname{Ind}_{B_2(\mathbb{F}_p)}^{\operatorname{GL}_2(\mathbb{F}_p)}(F(a_2) \otimes F(a_1))$ coming from Frobenius reciprocity (where $B_2 \subseteq \operatorname{GL}_2$ denotes the upper-triangular Borel subgroup).

As in [EGH13], proof of Lemma 6.1.1, we check that

$$\operatorname{soc}(\tau) \cong F(a_0 + p - 1, a_1, a_2 - p + 1), \operatorname{cosoc}(\tau) \cong F(a_1 + p - 1, a_2, a_0)$$

and, again by [EGH13], Lemma 6.1.1, we see that $rad(\tau)/soc(\tau)$ has length 4, with constituents

$$F(a_1, a_0, a_2 - p + 1), F(a_0 + p - 2, a_2, a_1 + 1),$$

 $F(a_2 - 1, a_1, a_0 + 1), F(a_1 - 1, a_0, a_2 - p + 2).$

By [And87], §4 we deduce that $rad(\tau)/soc(\tau)$ is semisimple, in particular the extension V exists and we have a surjection $\tau \twoheadrightarrow V$. Also, by *loc. cit.*, the extension V is unique, as

$$\operatorname{Ext}_{\overline{G}(\mathbb{F}_p)}^{1}(F(a_1+p-1,a_2,a_0),F(a_2-1,a_1,a_0+1))$$
$$\xrightarrow{\sim} \left(\operatorname{Ext}_{\operatorname{SL}_3(\mathbb{F}_p)}^{1}(F(a_1+p-1,a_2,a_0),F(a_2-1,a_1,a_0+1))\right)^{\mathbb{F}_p^{\times}}$$

has dimension at most one.

We also see by [Her11], Lemma 2.3 that $V^{I,(a_2-1,a_1,a_0+1)}$ and $V^{I,(a_1,a_2,a_0)} \cong \tau^{I,(a_1,a_2,a_0)} \cong \sigma^{I,(a_1,a_2,a_0)}$ are one-dimensional.

 $\tau^{I,(a_1,a_2,a_0)} \cong \sigma^{I,(a_1,a_2,a_0)}$ are one-dimensional. Let $W \stackrel{\text{def}}{=} H^0(\lambda)$, where $\lambda \stackrel{\text{def}}{=} (a_0 + p - 1, a_1, a_2 - p + 1)$. We recall that $0 \to F(\lambda) \to W \to F(a_2 - 1, a_1, a_0 + 1) \to 0$ ([Her09], Proposition 3.18) and moreover that this sequence is non-split even on the level of $\overline{G}(\mathbb{F}_p)$ -representations ([Hum06], Theorem 5.9). The natural $\overline{G}(\mathbb{F}_p)$ -linear evaluation map

$$f: W = \left(\operatorname{Ind}_{\overline{B}}^{\overline{G}}(w_0 \lambda) \right)^{\operatorname{alg}} \longrightarrow \operatorname{Ind}_{\overline{B}(\mathbb{F}_p)}^{\overline{G}(\mathbb{F}_p)}(w_0 \lambda) = \sigma$$

is injective. (To see this, it is enough to check that the restriction of f to $\operatorname{soc}_{\overline{G}(\mathbb{F}_p)}(W) \cong F(\lambda)$ is injective, and hence enough to check that the composite of f followed by evaluation at $w_0 \in \overline{G}(\mathbb{F}_p)$ is injective on the highest weight space of $\operatorname{soc}_{\overline{G}(\mathbb{F}_p)}(W)$. The last statement is true by the proof of [Jan03], Proposition II.2.2(a).) As σ is multiplicity free we obtain an injection $W \hookrightarrow \tau$.

Lemma 3.1.4. We have $S(\tau^{I,(a_1,a_2,a_0)}) \subseteq rad(\tau)$.

Proof. It is equivalent to show that the operator S kills the highest weight space $F(\nu)^{I,(a_1,a_2,a_0)} = F(\nu)_{\nu}$, where $\nu \stackrel{\text{def}}{=} (a_1 + p - 1, a_2, a_0)$.

Recall that if Q is an algebraic \overline{G} -module, $v_{\nu} \in Q_{\nu}$ and $\alpha \in \Phi$ is a root then (cf. [Jan03], II.1.19, (5) and (6))

$$u_{\alpha}(t)v_{\nu} = \sum_{i \in \mathbb{N}} t^{i} v_{\nu+i\alpha},$$

where $v_{\nu+i\alpha} \in Q_{\nu+i\alpha}$ (and $0^0 = 1$).

36
Applying this with $Q = F(\nu)$ we obtain

$$S(v_{\nu}) = \sum_{i,j,k \ge 0} \left(\sum_{x,y,z \in \mathbb{F}_p} x^{i+p-(a_2-a_0)} y^j z^{k+p-(a_1-a_0)} \right) v_{i,j,k}$$

where $v_{i,j,k} \in F(\nu)_{(a_0,a_2,a_1+p-1)+(i+j,-i+k,-j-k)}$. We deduce that if the inner sum is non-zero for a fixed triple $(i, j, k) \in \mathbb{N}^3$, then we necessarily have $i \ge a_2 - a_0 - 1$, $j \ge p-1$ and $k \ge a_1 - a_0 - 1$. But then $(a_0, a_2, a_1 + p - 1) + (i+j, -i+k, -j-k) \le (a_1 + p - 1, a_2, a_0)$ and the corresponding weight space in $F(\nu)$ is zero. \Box

Remark 3.1.5. By an immediate computation one checks that $S\left(\tau^{I,(a_1,a_2,a_0)}\right) \subseteq \tau^{\overline{T}(\mathbb{F}_p),(a_2-1,a_1,a_0+1)}$

Lemma 3.1.6. We have $S(\tau^{I,(a_1,a_2,a_0)}) \subseteq W$.

Proof. By Lemma 3.1.4 and Remark 3.1.5, it is enough to show that the weight spaces of $F(a_1, a_0, a_2 - p + 1)$, $F(a_0 + p - 2, a_2, a_1 + 1)$ and of $F(a_1 - 1, a_0, a_2 - p + 2)$ do not afford the $\overline{T}(\mathbb{F}_p)$ -character $F(a_2 - 1) \otimes F(a_1) \otimes F(a_0 + 1)$.

By using the dominance order \leq on $X^*(\overline{T})$ we see that the weights in $F(a_1, a_0, a_2 - p + 1)$ are of the form (α, β, γ) with $\alpha, \beta, \gamma \in \{a_2 - p + 1, \dots, a_1 - 1, a_1\}$. Hence $a_2 - p < \alpha < a_2 - 1$ and $\alpha \not\equiv a_2 - 1$ modulo (p-1). With a similar argument we see that any weight (α, β, γ) appearing in $F(a_0 + p - 2, a_2, a_1 + 1), F(a_1 - 1, a_0, a_2 - p + 2)$ satisfies $\beta \not\equiv a_1$ modulo (p-1).

Lemma 3.1.7. We have $S(\tau^{I,(a_1,a_2,a_0)}) \neq 0$.

Proof. Consider $\tau \hookrightarrow \sigma$. Then $\tau^{I,(a_1,a_2,a_0)}$ consists of functions $f \in \sigma$ such that

(i) $\operatorname{supp}(f) \subseteq \overline{B}(\mathbb{F}_p)\dot{s}_1\overline{B}(\mathbb{F}_p);$ (ii) f is constant on $\dot{s}_1\begin{pmatrix}1 & * & *\\ & 1 & *\\ & & 1\end{pmatrix}.$

Take any non-zero $f \in \tau^{I,(a_1,a_2,a_0)}$ and let $g \stackrel{\text{def}}{=} \begin{pmatrix} 1 & 1 \\ 1 & -1 \end{pmatrix}$. It then follows immediately from the definitions that

$$S(f)(g) = \sum_{x,y,z \in \mathbb{F}_p} x^{p-(a_2-a_0)} z^{p-(a_1-a_0)} f\left(\begin{array}{ccc} 1 & & \\ z-1 & & \\ y-z & x-1 & 1 \end{array}\right).$$

We deduce from (i) that the term inside the sum is non-zero iff $x = 1, z = y, z \neq 1$. Hence,

$$S(f)(g) = \sum_{z \in \mathbb{F}_p^{\times}} (z+1)^{p-(a_1-a_0)} f\begin{pmatrix} 1 & & \\ z & 1 & \\ & & 1 \end{pmatrix}$$
$$= (-1)^{a_2} \sum_{z \in \mathbb{F}_p^{\times}} (z+1)^{p-(a_1-a_0)} z^{-(a_2-a_1)} f(\dot{s}_1),$$

where the last equality follows from (i), (ii) via the relation

$$\begin{pmatrix} 1\\z&1\\&&1 \end{pmatrix} = \begin{pmatrix} -z^{-1}&1\\&z\\&&1 \end{pmatrix} \dot{s}_1 \begin{pmatrix} 1&z^{-1}\\&1\\&&1 \end{pmatrix}.$$

Note that the term inside the sum is a linear combination of monomials z^i with |i| . We easily deduce:

$$S(f)(g) = (-1)^{a_2 - 1} \binom{p - (a_1 - a_0)}{a_2 - a_1} f(\dot{s}_1) \neq 0.$$

Lemma 3.1.8. We have $S(V^{I,(a_1,a_2,a_0)}) \subseteq V^{I,(a_2-1,a_1,a_0+1)}$.

Proof. This is clear from Lemma 3.1.4 and Remark 3.1.5, as (just as in Lemma 3.1.6) we have $\operatorname{soc}(V)^{\overline{T}(\mathbb{F}_p),(a_2-1,a_1,a_0+1)} = \operatorname{soc}(V)^{\overline{U}(\mathbb{F}_p)}$.

Proof of Proposition 3.1.2. We start by proving (i). Recall that we defined $W \stackrel{\text{def}}{=} H^0(\lambda)$ with $\lambda \stackrel{\text{def}}{=} (a_0 + p - 1, a_1, a_2 - p + 1)$. By Lemma 3.1.8 it is enough to show that $S(V^{I,(a_1,a_2,a_0)}) \neq 0$ or, equivalently (cf. Lemma 3.1.6), that $S(\tau^{I,(a_1,a_2,a_0)}) \not\subseteq \text{soc}(W)$.

Let us define $X \stackrel{\text{def}}{=} \sum_{y \in \mathbb{F}_p} y^{p-2} \begin{pmatrix} 1 & y \\ & 1 \end{pmatrix}$. We note that X kills $S(\tau^{I,(a_1,a_2,a_0)})$

as the latter is easily checked to be fixed under $\begin{pmatrix} 1 & & \mathbb{F}_p \\ & & 1 \end{pmatrix}$; hence, to prove (i), it is enough to show that X acts injectively on $\operatorname{soc}(W)^{\overline{T}(\mathbb{F}_p),(a_2-1,a_1,a_0+1)}$.

As in Lemma 3.1.6 we see that $W^{\overline{T}(\mathbb{F}_p),(a_2-1,a_1,a_0+1)} = W_{\mu}$, where $\mu \stackrel{\text{def}}{=} (a_2 - \mu)^{-1} = W_{\mu}$, where $\mu \stackrel{\text{def}}{=} (a_2 - \mu)^{-1} = W_{\mu}$.

1, $a_1, a_0 + 1$). Let $\alpha \stackrel{\text{def}}{=} (1, 0, -1) \in \Phi$. By [Jan03], II.1.19, (5) and (6) we have for $v_\mu \in W_\mu$:

$$u_{\alpha}(t)v_{\mu} = \sum_{i\geq 0} t^{i} \left(X_{\alpha,i}v_{\mu} \right),$$

where $X_{\alpha,i} \in \text{Dist}(\overline{U}_{\alpha})$ and $X_{\alpha,i}v_{\mu} \in W_{\mu+i\alpha}$ (see [Jan03], II.1.12 for the definition of the operators $X_{\alpha,i}$). As $\mu + i\alpha \not\leq \lambda$ if $i \geq p$ we deduce that $X = -X_{\alpha,1}$ on W_{μ} .

To prove the proposition we will show that $X_{\alpha,1}: W_{\mu} \twoheadrightarrow W_{\mu+\alpha}$ is surjective, with a one-dimensional kernel not contained in $\operatorname{soc}(W)$.

Let \overline{M} denote the Levi subgroup $\begin{pmatrix} * & * & * \\ * & * & * \end{pmatrix} \subseteq \overline{G}$. Any irreducible constituent of the restriction $W|_{\overline{M}}$ has highest weight (α, β, γ) with $\alpha, \beta, \gamma \in \{a_2 - p + 1, \ldots, a_0 + p - 1\}$. As $a_0 + p > a_0 + p - 1$, the Linkage Principle [Jan03], II.6.17 and [Jan03], II.2.12(1) yield

$$W|_{\overline{M}} \cong F^{\oplus m} \oplus W',$$

where F is the irreducible \overline{M} -representation with highest weight μ and no irreducible constituent of W' has highest weight μ .

Claim: The operator $X_{\alpha,1}$ induces an isomorphism $W'_{\mu} \xrightarrow{\sim} W'_{\mu+\alpha}$.

Proof of the claim. By dévissage we may assume that W' is irreducible. Let $\nu \in X^*(\overline{T})$ be its highest weight, so in particular $\nu \neq \mu$. Then $W' \hookrightarrow H^0_{\overline{M}}(\nu)$, where $H^0_{\overline{M}}(\nu)$ denotes the dual Weyl module for \overline{M} of highest weight ν . As $\text{Dist}(\overline{M})$ preserves $W' \hookrightarrow H^0_{\overline{M}}(\nu)$ it is enough to show that $X_{\alpha,1} : H^0_{\overline{M}}(\nu)_{\mu} \to H^0_{\overline{M}}(\nu)_{\mu+\alpha}$ and $X_{-\alpha,1} : H^0_{\overline{M}}(\nu)_{\mu+\alpha} \to H^0_{\overline{M}}(\nu)_{\mu}$ are both injective.

and $X_{-\alpha,1}: H^0_{\overline{M}}(\nu)_{\mu+\alpha} \to H^0_{\overline{M}}(\nu)_{\mu}$ are both injective. Since $M^{\operatorname{der}} \cong \operatorname{SL}_2$ and $\begin{pmatrix} 1 & t \\ & 1 \end{pmatrix} X^i Y^j = X^i Y^j + tj X^{i+1} Y^{j-1} + \dots$, we see that $X_{\alpha,1}: H^0_{\overline{M}}(\nu)_{\mu} \to H^0_{\overline{M}}(\nu)_{\mu+\alpha}$ is injective provided that $\frac{1}{2} \left(\langle \nu, \alpha^{\vee} \rangle - \langle \mu, \alpha^{\vee} \rangle \right) \not\equiv 0$ modulo p. (This is because one has $H^0_{\overline{M}}(\nu)|_{\overline{M}^{\operatorname{der}}} \cong \operatorname{Sym}^{\langle \nu, \alpha^{\vee} \rangle}(\operatorname{Std})$ and $0 \neq X^{\frac{1}{2}(\langle \nu, \alpha^{\vee} \rangle + \langle \mu, \alpha^{\vee} \rangle)}Y^{\frac{1}{2}(\langle \nu, \alpha^{\vee} \rangle - \langle \mu, \alpha^{\vee} \rangle)} \in H^0_{\overline{M}}(\nu)_{\mu}$.) Similarly, $X_{-\alpha,1}: H^0_{\overline{M}}(\nu)_{\mu+\alpha} \to H^0_{\overline{M}}(\nu)_{\mu}$ is injective provided that $\frac{1}{2} \left(\langle \nu, \alpha^{\vee} \rangle + \langle \mu + \alpha, \alpha^{\vee} \rangle \right) \not\equiv 0$ modulo p. As α^{\vee} is dominant and $\nu \leq \lambda$ (both relative to \overline{M}) we have

$$\langle \nu, \alpha^{\vee} \rangle \leq \langle \lambda, \alpha^{\vee} \rangle = 2(p-1) - (a_2 - a_0).$$

Similarly, if $H^0_{\overline{M}}(\nu)_{\mu} \neq 0$ or $H^0_{\overline{M}}(\nu)_{\mu+\alpha} \neq 0$ then

$$\langle \nu, \alpha^{\vee} \rangle > \langle \mu, \alpha^{\vee} \rangle = a_2 - a_0 - 2$$

(here we used that $\nu \neq \mu$). Now we can deduce the above two congruences, proving the claim.

From the claim we deduce that the morphism $X_{\alpha,1}: W_{\mu} \to W_{\mu+\alpha}$ is surjective, with kernel of dimension m. The Kostant multiplicity formula shows that

$$m = \dim_{\mathbb{F}} W_{\mu} - \dim_{\mathbb{F}} W_{\mu+\alpha}$$

= $(p+1-(a_2-a_0)) - (p-(a_2-a_0)) = 1.$

Moreover, $F(\mu) \cong \operatorname{cosoc}(W)$ contains the irreducible representation of \overline{M} of highest weight μ . Hence $\operatorname{soc}(W) \subseteq W'$ and X is indeed injective on $\operatorname{soc}(W)_{\mu}$. This proves part (i) of the proposition.

For statement (ii) it is now enough to apply the automorphism
$$g \mapsto \begin{pmatrix} & & 1 \\ 1 & & & \end{pmatrix}$$
.
 ${}^{t}g^{-1} \cdot \begin{pmatrix} & & 1 \\ 1 & & & \end{pmatrix}$ to part (i) and relabel $(-a_0, -a_1, -a_2)$ as (a_2, a_1, a_0) .

Remark 3.1.9. Similarly, there exists a unique non-split extension of irreducible $\overline{G}(\mathbb{F}_p)$ -representations

$$0 \to F(a_2 - 1, a_1, a_0 + 1) \to V \to F(a_0 + p - 1, a_1, a_2 - p + 1) \to 0,$$

namely V is the Weyl module of highest weight $(a_0 + p - 1, a_1, a_2 - p + 1)$. The following operator induces an isomorphism $V^{I,(a_0,a_1,a_2)} \xrightarrow{\sim} V^{I,(a_2-1,a_1,a_0+1)}$ of one-dimensional vector spaces:

$$\sum_{x,y,z\in\mathbb{F}_p}\sum_{i=0}^{p-(a_2-a_0)}\frac{(-1)^i}{i+1}\binom{p-(a_1-a_0)}{p-(a_2-a_0)-i}(xz)^{p-1-i}y^{i+1}\binom{1}{2} \begin{pmatrix} 1 & x & y \\ & 1 & z \\ & & 1 \end{pmatrix}\dot{w}_0.$$

We skip the somewhat lengthy proof, since we will not need it.

We conclude this section with a simple but important lemma concerning the action of U_p -operators on $G(\mathbb{Q}_p)$ -representations over \mathbb{F} . We define

10)
$$U_{1} \stackrel{\text{def}}{=} \sum_{x,y \in \mathbb{Z}_{p}/p\mathbb{Z}_{p}} \begin{pmatrix} p^{-1} & & \\ & 1 & \\ & & 1 \end{pmatrix} \begin{pmatrix} 1 & & \\ x & 1 & \\ y & & 1 \end{pmatrix},$$
$$U_{2} \stackrel{\text{def}}{=} \sum_{x,y \in \mathbb{Z}_{p}/p\mathbb{Z}_{p}} \begin{pmatrix} p^{-1} & & \\ & p^{-1} & \\ & & 1 \end{pmatrix} \begin{pmatrix} 1 & & \\ x & y & 1 \end{pmatrix},$$

(3.1.10)

so that for any $G(\mathbb{Q}_p)$ -representation σ over \mathfrak{O}_E and any $(a_2, a_1, a_0) \in \mathbb{Z}^3$, both U_1 and U_2 preserve $\sigma^{I,(a_2,a_1,a_0)}$.

Lemma 3.1.11. Let $(a_2, a_1, a_0) \in \mathbb{Z}^3$ be a triple with $a_2 - a_1 > 0$, $a_1 - a_0 > 0$ and $a_2 - a_0 . Let <math>\overline{\tau} \stackrel{\text{def}}{=} \operatorname{Ind}_I^K (\omega^{a_1} \otimes \omega^{a_2} \otimes \omega^{a_0})$, and suppose that σ is a representation of $G(\mathbb{Q}_p)$ over \mathbb{F} .

Then $\operatorname{Hom}_{K}(\overline{\tau}, \sigma)[U_{i}] = \operatorname{Hom}_{K}(\overline{\tau}/M_{i}, \sigma)$ for $i \in \{1, 2\}$, where M_{1} (resp. M_{2}) is the minimal subrepresentation of $\overline{\tau}$ containing $F(a_{2}, a_{0}, a_{1} - p + 1)$ (resp. $F(a_{0} + p - 1, a_{1}, a_{2} - p + 1)$) as subquotient. Concretely, both M_{i} are indecomposable of length 2 with socle $F(a_{0} + p - 1, a_{2}, a_{1})$.

Proof. Note that $\overline{\tau}^{I,(a_1,a_2,a_0)}$ is spanned by a function f with supp(f) = I and f(1) = 1. Consider the case i = 2, and suppose $\varphi \in \operatorname{Hom}_{K}(\overline{\tau}, \sigma)$. Then $U_{2}\varphi = 0$ iff $U_2(\varphi(f)) = 0. \text{ We note that } \Pi^2 U_2 = U'_2, \text{ where } U'_2 \stackrel{\text{def}}{=} \sum_{x,y \in \mathbb{F}_p} \begin{pmatrix} 1 & x & y \\ & 1 \\ & & 1 \end{pmatrix} \begin{pmatrix} 1 & & 1 \\ & & 1 \\ & & 1 \end{pmatrix},$ sending $\rho^{I,(a_1,a_2,a_0)}$ to $\rho^{I,(a_0,a_1,a_2)}$ for any K-representation ρ over \mathbb{F} . We have that $U'_2 f \neq 0$, as $(U'_2 f) \begin{pmatrix} 1 \\ 1 \end{pmatrix} = 1$. Hence $\langle K \cdot U'_2 f \rangle_{\mathbb{F}}$ is a quotient of $\overline{\tau}' \stackrel{\text{def}}{=} \operatorname{Ind}_{I}^{K} \left(\omega^{a_{0}} \otimes \omega^{a_{1}} \otimes \omega^{a_{2}} \right)$ that injects into $\overline{\tau}$, so as the cosocle of $\overline{\tau}'$ is $F(a_{0} + \omega^{a_{1}})$ $p-1, a_1, a_2-p+1$ and $\overline{\tau}$ is multiplicity-free, we deduce that $\langle K \cdot U'_2 f \rangle_{\mathbb{F}} = M_2$. By [Le], Proposition 2.2.2, M_2 is uniserial of shape $F(a_0 + p - 1, a_2, a_1) - F(a_0 + p - 1, a_2, a_3)$ $p - 1, a_1, a_2 - p + 1$). We have that

$$U_2\varphi = 0 \Longleftrightarrow U_2'(\varphi(f)) = 0 \Longleftrightarrow \varphi(U_2'f) = 0 \Longleftrightarrow \langle K \cdot U_2'f \rangle_{\mathbb{F}} \subseteq \ker \varphi.$$

The proof for the case i = 1 is analogous.

Corollary 3.1.12. Let $(a_2, a_1, a_0) \in \mathbb{Z}^3$ be a triple with $a_2 - a_1 > 0$, $a_1 - a_0 > 0$ and $a_2 - a_0 . Suppose <math>\sigma$ is a representation of $G(\mathbb{Q}_p)$ over \mathbb{F} .

- (i) If $v \in \sigma^{I,(a_1,a_2,a_0)}$ then $U_2v = 0$ if and only if $\langle Kv \rangle_{\mathbb{F}}$ does not contain $\begin{array}{l}F(a_0+p-1,a_1,a_2-p+1) \ as \ subquotient.\\ (ii) \ If \ v \in \ \sigma^{I,(a_2,a_0,a_1)} \ then \ U_1v = 0 \ if \ and \ only \ if \ \langle Kv \rangle_{\mathbb{F}} \ does \ not \ contain\end{array}$
- $F(a_0 + p 1, a_1, a_2 p + 1)$ as subquotient.

Proof. (i) In the notation of the proof of Lemma 3.1.11, by Frobenius reciprocity there is a unique $\varphi \in \operatorname{Hom}_K(\overline{\tau}, \sigma)$ such that $v = \varphi(f)$. The claim follows, as $\langle Kv \rangle_{\mathbb{F}} = \operatorname{im} \varphi$ (since $\langle Kf \rangle_{\mathbb{F}} = \overline{\tau}$). The proof of (ii) is analogous.

3.2. Jacobi sums and characteristic zero principal series. In this section we consider certain group algebra operators with \mathcal{O}_E -coefficients and study their effect on principal series representations.

We fix once and for all a triple $(a_2, a_1, a_0) \in \mathbb{Z}^3$ verifying (2.1.6). We define the following elements in the group algebra $\mathcal{O}_E[\overline{G}(\mathbb{F}_p)]$, lifting the mod-p operators introduced in Section 3.1:

(3.2.1)
$$\widehat{S} \stackrel{\text{def}}{=} \sum_{x,y,z \in \mathbb{F}_p} \widetilde{x}^{p-(a_2-a_0)} \widetilde{z}^{p-(a_1-a_0)} \begin{pmatrix} 1 & x & y \\ & 1 & z \\ & 1 \end{pmatrix} \dot{w}_0,$$
$$\widehat{S}' \stackrel{\text{def}}{=} \sum_{x,y,z \in \mathbb{F}_p} \widetilde{x}^{p-(a_2-a_1)} \widetilde{z}^{p-(a_2-a_0)} \begin{pmatrix} 1 & x & y \\ & 1 & z \\ & & 1 \end{pmatrix} \dot{w}_0.$$

The following relation between the actions of \widehat{S} , \widehat{S}' on certain principal series representations will be crucial later. To lighten notation, we set $\Pi \stackrel{\text{def}}{=} \begin{pmatrix} 1 & 1 \\ n & 1 \end{pmatrix} \in$ $N_{G(\mathbb{Q}_p)}(I).$

Proposition 3.2.2. Let $(a_2, a_1, a_0) \in \mathbb{Z}^3$ be a triple verifying (2.1.6). Let $\pi_p \stackrel{\text{def}}{=}$ $\operatorname{Ind}_{B(\mathbb{Q}_p)}^{G(\mathbb{Q}_p)}(\chi_1 \otimes \chi_2 \otimes \chi_0)$ be a principal series representation, where the smooth characters $\chi_i : \mathbb{Q}_p^{\times} \to E^{\times}$ verify $\chi_i|_{\mathbb{Z}_p^{\times}} = \widetilde{\omega}^{a_i}$ for $i \in \{0, 1, 2\}$.

On the one-dimensional isotypical component $\pi_p^{I,(a_1,a_2,a_0)}$ we have

(3.2.3)
$$\widehat{S}' \circ \Pi = p\chi_1(p)\kappa \widehat{S},$$

where $\kappa \in \mathbb{Z}_p^{\times}$ verifies $\kappa \equiv (-1)^{a_1-a_0} \cdot \frac{a_2-a_1}{a_1-a_0} \mod p$ and is independent of the scalars $\chi_i(p)$.

Remark 3.2.4. Note that equation (3.2.3) makes sense, as both \widehat{S} and \widehat{S}' act on $\pi^{K(1)}$ for any smooth K-representation π , where $K(1) \stackrel{\text{def}}{=} \ker \left(K \to \overline{G}(\mathbb{F}_p)\right)$.

The proof of Proposition 3.2.2 relies on certain direct manipulations on Jacobi sums and will occupy the rest of this section. Let us pick a non-zero element \hat{v} in the one-dimensional isotypical component $\pi_p^{I,(a_1,a_2,a_0)}$. In particular, \hat{v} is K(1)-fixed. We note that $\operatorname{supp}(\hat{v}) = B(\mathbb{Q}_p)I$ and $\hat{v}(1) \neq 0$.

Lemma 3.2.5. We have

$$\Pi \cdot \widehat{v} = \chi_1(p) \sum_{\lambda, \mu \in \mathbb{F}_p} \begin{pmatrix} \lambda & 1 \\ \mu & 1 \\ 1 & 1 \end{pmatrix} \cdot \widehat{v}.$$

Proof. Noting that $\Pi U_1 = \sum_{\lambda,\mu\in\mathbb{F}_p} \begin{pmatrix} \lambda & 1 \\ \mu & 1 \end{pmatrix}$, it suffices to show that $U_1 \cdot \hat{v} = U_1(\mu,\mu) = U_1(\mu,\mu)$.

 $\chi_1(p)^{-1} \cdot \hat{v}$. As $U_1 \cdot \hat{v}$ lies in the one-dimensional space $\pi_p^{I,(a_1,a_2,a_0)}$ and $\hat{v}(1) \neq 0$, it is enough to observe that

$$(U_1\widehat{v})(1) = \sum_{\lambda,\mu\in\mathbb{Z}_p/p\mathbb{Z}_p}\widehat{v}\left(\begin{pmatrix}p^{-1} & & \\ & 1 & \\ & & 1\end{pmatrix}\begin{pmatrix}1 & & \\ & \mu & & 1\end{pmatrix}\right) = \chi_1(p)^{-1}\cdot\widehat{v}(1). \quad \Box$$

We now compute the action of the operator \hat{S}' on the element $\Pi \hat{v}$.

Lemma 3.2.6. We have

$$\widehat{S}'(\Pi \widehat{v}) = (-1)^{a_2 - a_0} \chi_1(p) \sum_{\lambda, \mu, x, y, z \in \mathbb{F}_p} \widetilde{\lambda}^{a_2 - a_1} \widetilde{\mu}^{p - 1 - (a_1 - a_0)} (\overbrace{x - \mu}^{p - (a_2 - a_1)} (\overbrace{z - \lambda}^{p - (a_2 - a_0)} \begin{pmatrix} 1 & x & y \\ 1 & 1 & z \\ & & 1 \end{pmatrix}} \dot{w}_0 \widehat{v}.$$

Proof. In order to ease notation, we write $a \stackrel{\text{def}}{=} a_2 - a_0$, $b \stackrel{\text{def}}{=} a_1 - a_0$. An immediate computation using Lemma 3.2.5 gives

(3.2.7)
$$\widehat{S}'(\Pi \widehat{v}) = \chi_1(p) \sum_{\lambda,\mu,x,y,z \in \mathbb{F}_p} \widetilde{x}^{p-(a-b)} \widetilde{z}^{p-a} \underbrace{\begin{pmatrix} 1+y\lambda+x\mu & y & x\\ z\lambda+\mu & z & 1\\ \lambda & \\ \vdots & \vdots \\ \lambda & \\ \vdots & \vdots \\ \vdots & \vdots$$

We now break up the sum (3.2.7) according to the values of $\lambda, \mu \in \mathbb{F}_p$. Case 1: we have $\lambda \mu = 0$. Suppose first that $\lambda = \mu = 0$. Then one has

$$A_{0,0} = \begin{pmatrix} 1 & z \\ 1 & z \\ 1 & 1 \end{pmatrix} \dot{s}_2 \begin{pmatrix} 1 & y & x \\ 1 & 1 \\ 1 & 1 \end{pmatrix}$$

and therefore, using that \hat{v} is I_1 -fixed, we obtain

$$\sum_{x,y,z\in\mathbb{F}_p} \widetilde{x}^{p-(a-b)} \widetilde{z}^{p-a} A_{0,0} \widehat{v} = \left(\sum_{z\in\mathbb{F}_p} \widetilde{z}^{p-a} \begin{pmatrix} 1 & & \\ & 1 & \\ & & 1 \end{pmatrix} \dot{s}_2 \widehat{v} \right) \left(\sum_{x,y\in\mathbb{F}_p} \widetilde{x}^{p-(a-b)} \right)$$
$$= 0.$$

In an analogous fashion, if $\lambda \mu \neq 0$ we have

$$A_{\lambda,0} = \begin{pmatrix} 1 & y + \lambda^{-1} \\ 1 & z \\ 1 & 1 \end{pmatrix} \dot{s}_2 \dot{s}_1 \begin{pmatrix} \lambda & 1 \\ -\lambda^{-1} & x \\ 1 & 1 \end{pmatrix},$$
$$A_{0,\mu} = \begin{pmatrix} 1 & \mu^{-1} + x & y - z(\mu^{-1} + x) \\ 1 & 1 & 1 \end{pmatrix} \dot{s}_1 \dot{s}_2 \begin{pmatrix} \mu & z & 1 \\ 1 & -\mu^{-1} \end{pmatrix},$$

so, using that \hat{v} is an eigenvector for the *I*-action, we obtain

$$\sum_{x,y,z\in\mathbb{F}_p}\sum_{\lambda\in\mathbb{F}_p}\widetilde{x}^{p-(a-b)}\widetilde{z}^{p-a}A_{\lambda,0}\widehat{v}=0,$$
$$\sum_{x,y,z\in\mathbb{F}_p}\sum_{\mu\in\mathbb{F}_p}\widetilde{x}^{p-(a-b)}\widetilde{z}^{p-a}A_{0,\mu}\widehat{v}=0$$

(in the first equation, the sum over x is zero; in the second equation the sum over z is zero, provided one first replaces y by $y' \stackrel{\text{def}}{=} y - z(\mu^{-1} + x))$.

Case 2: we have $\lambda \mu \neq 0$. As before, a direct computation gives

$$A_{\lambda,\mu} = \begin{pmatrix} 1 & \mu^{-1} + x & y + x\mu\lambda^{-1} + \lambda^{-1} \\ 1 & z + \mu\lambda^{-1} \\ 1 & 1 \end{pmatrix} \dot{w}_0 \begin{pmatrix} \lambda & 1 \\ -\mu\lambda^{-1} & 1 \\ & -\mu^{-1} \end{pmatrix}$$

and, recalling that \hat{v} lies in $\pi_p^{I,(a_1,a_2,a_0)}$, we obtain

$$\sum_{x,y,z\in\mathbb{F}_p}\sum_{\lambda,\mu\in\mathbb{F}_p^{\times}}\widetilde{x}^{p-(a-b)}\widetilde{z}^{p-a}A_{\lambda,\mu}\widehat{v} =$$

$$=\sum_{x,y,z\in\mathbb{F}_p}\sum_{\lambda,\mu\in\mathbb{F}_p^{\times}}(-1)^a\widetilde{\lambda}^{-(a-b)}\widetilde{\mu}^a\widetilde{x}^{p-(a-b)}\widetilde{z}^{p-a}\left(\begin{array}{ccc}1&\mu^{-1}+x&y+x\mu\lambda^{-1}+\lambda^{-1}\\1&z+\mu\lambda^{-1}\\1&1\end{array}\right)\dot{w}_0\widehat{v}$$

We obtain the desired result by putting the two cases together and using the change of variables $(x', y', z', \lambda', \mu') \stackrel{\text{def}}{=} (x + \mu^{-1}, y + x\mu\lambda^{-1} + \lambda^{-1}, z + \mu\lambda^{-1}, \mu\lambda^{-1}, \mu^{-1}).$

Proof of Proposition 3.2.2. Note that in the expression of Lemma 3.2.6 we may assume that $x \neq 0$ (resp. $z \neq 0$) since otherwise the sum over μ (resp. λ) vanishes. Using the change of variables $(\lambda', \mu') \stackrel{\text{def}}{=} (\lambda z^{-1}, \mu x^{-1})$ we obtain

$$\widehat{S}'(\Pi \widehat{v}) = (-1)^a \chi_1(p) \kappa_1 \kappa_2 \widehat{S}(\widehat{v}),$$

where $\kappa_1, \kappa_2 \in \mathbb{Z}_p$ are the following Jacobi sums:

$$\kappa_1 \stackrel{\text{def}}{=} \sum_{\mu \in \mathbb{F}_p} \widetilde{\mu}^{p-1-b} (\widetilde{1-\mu})^{p-(a-b)}, \qquad \kappa_2 \stackrel{\text{def}}{=} \sum_{\lambda \in \mathbb{F}_p} \widetilde{\lambda}^{a-b} (\widetilde{1-\lambda})^{p-a}.$$

By Stickelberger's theorem (see e.g. [BD14], Théorème 2.5.1) one has $\operatorname{ord}_p(\kappa_1) = 0$, $\operatorname{ord}_p(\kappa_2) = 1$ and

$$\kappa_1 \equiv \frac{(p-1-b)!(p-(a-b))!}{(p-a)!}, \qquad \frac{\kappa_2}{p} \equiv (-1)\frac{(a-b)!(p-a)!}{(p-b)!} \mod p.$$

Therefore

$$\widehat{S}'(\Pi \widehat{v}) = (-1)^a \chi_1(p) \kappa \widehat{S}(\widehat{v}),$$

where $\operatorname{ord}_p(\kappa) = 1$ and $\frac{\kappa}{p} \equiv -\frac{(p-(a-b))!(a-b)!}{p-b} \mod p$. As $(p-(a-b))!(a-b)! \equiv (-1)^{p-(a-b)}(p-1)!(a-b) \mod p$ we finally obtain

$$(-1)^a \frac{\kappa}{p} \equiv (-1)^b \frac{a-b}{b},$$

and the proof of Proposition 3.2.2 is now complete.

Lemma 3.2.8. Suppose that $\pi_p \stackrel{\text{def}}{=} \operatorname{Ind}_{B(\mathbb{Q}_p)}^{G(\mathbb{Q}_p)} (\chi_1 \otimes \chi_2 \otimes \chi_0)$ is a principal series representation, where the smooth characters $\chi_i : \mathbb{Q}_p^{\times} \to E^{\times}$ verify $\chi_i|_{\mathbb{Z}_p^{\times}} = \widetilde{\omega}^{a_i}$ for $i \in \{0, 1, 2\}$ and where a_0, a_1, a_2 are distinct modulo p - 1. On the one-dimensional isotypical component $\pi_p^{I,(a_1,a_2,a_0)}$ we have $U_1 = \chi_1(p)^{-1}$ and $U_2 = \chi_1(p)^{-1}\chi_2(p)^{-1}$.

Proof. For U_1 this was observed in the proof of Lemma 3.2.5. The U_2 -eigenvalue is computed similarly.

4. Local-global compatibility

In this section we establish most of our main results, namely Theorems A, D, and E of the introduction.

4.1. The space of automorphic forms on certain unitary groups. Let F/\mathbb{Q} be a CM field and F^+ its maximal totally real subfield. Assume that $F^+ \neq \mathbb{Q}$, and that p splits completely in F. We write c for the generator of $\operatorname{Gal}(F/F^+)$. For $w \nmid \infty$ (resp. $v \nmid \infty$) a place of F (resp. F^+) we denote by k_w (resp. k_v) the residue field of F_w (resp. F_v^+).

We let $G_{/F^+}$ be a reductive group, which is an outer form of GL₃ which splits over F. We assume that $G(F_v^+) \cong U_3(\mathbb{R})$ for all $v \mid \infty$. By the argument of [CHT08], §3.3, G admits a model \mathcal{G} over \mathcal{O}_{F^+} such that $\mathcal{G} \times \mathcal{O}_{F_v^+}$ is reductive for all places v of F^+ that split in F. For any such place v of F^+ and $w \mid v$ of F we get an isomorphism $\iota_w : G(F_v^+) \xrightarrow{\sim} \operatorname{GL}_3(F_w)$ which restricts moreover to an isomorphism $\iota_w : \mathcal{G}(\mathcal{O}_{F_v^+}) \xrightarrow{\sim} \operatorname{GL}_3(\mathcal{O}_{F_w})$.

Let $F_p^+ \stackrel{\text{def}}{=} F^+ \otimes_{\mathbb{Q}} \mathbb{Q}_p$ and $\mathcal{O}_{F^+,p} \stackrel{\text{def}}{=} \mathcal{O}_{F^+} \otimes_{\mathbb{Z}} \mathbb{Z}_p$. If W is an \mathcal{O}_E -module endowed with an action of $\mathcal{G}(\mathcal{O}_{F^+,p})$ and $U \leq G(\mathbb{A}_{F^+}^{\infty,p}) \times \mathcal{G}(\mathcal{O}_{F^+,p})$ is a compact open subgroup, the space of algebraic automorphic forms on G of level U and coefficients in W is defined as the following \mathcal{O}_E -module:

$$S(U,W) \stackrel{\text{def}}{=} \left\{ f : G(F^+) \backslash G(\mathbb{A}_{F^+}^{\infty}) \to W \, | \, f(gu) = u_p^{-1} f(g) \, \, \forall g \in G(\mathbb{A}_{F^+}^{\infty}), u \in U \right\}$$

(with the obvious notation $u = u^p u_p$ for the elements in U).

We recall that the level U is said to be sufficiently small if for all $t \in G(\mathbb{A}_{F^+}^{\infty})$, the finite group $t^{-1}G(F^+)t \cap U$ is of order prime to p. For a finite place v of F^+ we say that U is unramified at v if one has a decomposition $U = \mathcal{G}(\mathcal{O}_{F_v^+})U^v$ for some compact open subgroup $U^v \leq G(\mathbb{A}_{F^+}^{\infty,v})$.

Let \mathcal{P}_U denote the set consisting of finite places w of F such that $v \stackrel{\text{def}}{=} w|_{F^+}$ is split in $F, v \nmid p$ and U is unramified at v. If $\mathcal{P} \subseteq \mathcal{P}_U$ is a subset of finite complement that is closed under complex conjugation, we write $\mathbb{T}^{\mathcal{P}} = \mathcal{O}_E[T_w^{(i)} : w \in \mathcal{P}, i \in \{0, 1, 2, 3\}]$ for the abstract Hecke algebra on \mathcal{P} , where the Hecke operator $T_w^{(i)}$ acts on the space S(U, W) as the usual double coset operator

$$\iota_w^{-1} \left[\operatorname{GL}_3(\mathcal{O}_{F_w}) \begin{pmatrix} \varpi_w \operatorname{Id}_i \\ & \operatorname{Id}_{3-i} \end{pmatrix} \operatorname{GL}_3(\mathcal{O}_{F_w}) \right],$$

where ϖ_w denotes a uniformizer of F_w .

4.2. Serre weights. In this section we recall the notion of Serre weights, as well as define the set of Serre weights of a Galois representation $\overline{r}: G_F \to \mathrm{GL}_3(\mathbb{F})$.

Definition 4.2.1. A Serre weight for \mathcal{G} (or just Serre weight if \mathcal{G} is clear from the context) is an isomorphism class of a smooth, (absolutely) irreducible representation of $\mathcal{G}(\mathcal{O}_{F^+,p})$ over \mathbb{F} . If w|p is a place of F, a Serre weight at w is an isomorphism class of a smooth, (absolutely) irreducible representation of $\mathrm{GL}_3(\mathcal{O}_{F_w})$ over \mathbb{F} , or equivalently an isomorphism class of an (absolutely) irreducible representation of $\mathrm{GL}_3(k_w)$ over \mathbb{F} .

Let S_p^+ (resp. S_p) denote the set of places of F^+ (resp. F) that divide p. Suppose that for each $w \in S_p$ we are given a p-restricted triple $a_w = (a_{w,2}, a_{w,1}, a_{w,0}) \in \mathbb{Z}^3$ such that $a_{w,i} + a_{w^c,2-i} = 0$ for all $0 \le i \le 2$ and all $w \in S_p$. Let F_{a_w} denote the irreducible representation $F(a_{w,2}, a_{w,1}, a_{w,0})$ of $\operatorname{GL}_3(k_w)$ over \mathbb{F} defined in Section 3. Then $F_{a_v} \stackrel{\text{def}}{=} F_{a_w} \circ \iota_w$ is an irreducible representation of $\mathcal{G}(\mathcal{O}_{F_v^+})$ that is independent of the choice of place w dividing v, and we define the Serre weight $F_a \stackrel{\text{def}}{=} \bigotimes_{v|p} F_{a_v}$.

By [EGH13], Lemma 7.3.4 any Serre weight is of this form.

Definition 4.2.2. Let $\overline{r}: G_F \to \operatorname{GL}_3(\mathbb{F})$ be a continuous, absolutely irreducible Galois representation and let V be a Serre weight for \mathcal{G} . We say that \overline{r} is *automorphic of weight* V or that V is a *Serre weight of* \overline{r} if there exist a compact open subgroup U of $G(\mathbb{A}_F^{\infty, \mathcal{P}}) \times \mathcal{G}(\mathcal{O}_{F^+, \mathcal{P}})$ which is unramified at all places dividing pand a cofinite subset \mathcal{P} of \mathcal{P}_U such that \overline{r} is unramified at each place of \mathcal{P} and $S(U, V)_{\mathfrak{m}_{\overline{r}}} \neq 0$, where $\mathfrak{m}_{\overline{r}}$ is the maximal ideal of $\mathbb{T}^{\mathcal{P}}$ with residue field \mathbb{F} defined by the formula

$$\det\left(1-\overline{r}^{\vee}(\operatorname{Frob}_w)X\right) = \sum_{j=0}^3 (-1)^j (\mathbf{N}_{F/\mathbb{Q}}(w))^{\binom{j}{2}} (T_w^{(j)} \mod \mathfrak{m}_{\overline{r}}) X^j \quad \forall w \in \mathcal{P}$$

(and $\mathbf{N}_{F/\mathbb{Q}}(w)$ denotes the norm from F to \mathbb{Q} of the prime w). We write $W(\overline{r})$ for the set of all Serre weights of \overline{r} .

From now on we fix a place $w \in S_p$ and assume that \overline{r} is automorphic of weight F_a , where $a_{w'}$ lies in the lowest alcove, i.e. $a_{w',2} - a_{w',0} , for all <math>w' \in S_p \setminus \{w, w^c\}$.

Let $v \stackrel{\text{def}}{=} w|_{F^+}$ and $V' \stackrel{\text{def}}{=} \bigotimes_{v' \in S_p^+ \setminus \{v\}} F_{a_{v'}}$. Now the construction in [EGH13], §7.1.4 (replacing the coefficients $\overline{\mathbb{Z}}_p$ by \mathcal{O}_E) gives a finite free \mathcal{O}_E -module $W_{a_{v'}}$ with a linear action of $\mathcal{G}(\mathcal{O}_{F_{v'}^+})$ for any $v' \in S_p^+ \setminus \{v\}$, and we let $\widetilde{V}' \stackrel{\text{def}}{=} \bigotimes_{v' \in S_p^+ \setminus \{v\}} W_{a_{v'}}$. Note that since $a_{w'}$ lies in the lowest alcove for all w' we have $\widetilde{V}' \otimes_{\mathcal{O}_E} \mathbb{F} \cong V'$.

Definition 4.2.3. We define

 $W_w(\overline{r}) \stackrel{\text{def}}{=} \left\{ F_{a_w} : a_w \in \mathbb{Z}^3 \text{ is } p \text{-restricted and } (F_{a_w} \circ \iota_w) \otimes_{\mathbb{F}} V' \in W(\overline{r}) \right\}.$ Note that by construction we have $W_w(\overline{r}) \neq \emptyset$.

From now on we suppose that U is a compact open subgroup of $G(\mathbb{A}_{F}^{\infty,p}) \times \mathfrak{G}(\mathfrak{O}_{F^{+},p})$ that is sufficiently small, unramified at all places dividing p, and a subset $\mathcal{P} \subseteq \mathcal{P}_{U}$ as above such that $S(U, (V \circ \iota_{w}) \otimes_{\mathbb{F}} V')_{\mathfrak{m}_{\overline{r}}} \neq 0$ for some Serre weight V at w. (Such a group U exists, since we are free to shrink U further to ensure it is sufficiently small, cf. [EGH13], Remark 7.3.6.)

Having fixed V', for any representation V of $\operatorname{GL}_3(\mathcal{O}_{F_w}) = \operatorname{GL}_3(\mathbb{Z}_p)$ over \mathcal{O}_E , let

$$S(V) \stackrel{\text{def}}{=} S(U, (V \circ \iota_w) \otimes_{\mathcal{O}_E} V').$$

In particular, $W_w(\overline{r}) \supseteq \{$ Serre weights V at $w : S(V)_{\mathfrak{m}_{\overline{r}}} \neq 0 \}$ (and equality holds if we shrink U and \mathcal{P} sufficiently).

4.3. Potentially semistable lifts. In this section we discuss the relationship between Serre weights and potentially semistable lifts. We start by recalling some facts about Deligne–Lusztig representations, referring to [Her09], Section 4 for details.

For any $n \ge 1$ let $k_{w,n}$ denote the extension of k_w of degree n. We write T for a maximal torus of GL_{3/k_w} . We have an identification

(4.3.1)
$$T(k_w) \xrightarrow{\sim} \prod_{j=1}^{r} k_{w,n_j}^{\times}$$

where $1 \le n_j \le 3$ and $\sum_{j=1}^r n_j = 3$ (cf. [Her09], Lemma 4.7), which is well defined up to $\prod_{j=1}^r \text{Gal}(k_{w,n_j}/k_w)$.

For any homomorphism $\theta: T(k_w) \to \overline{\mathbb{Q}}_p^{\times}$ we have a Deligne-Lusztig representation R_T^{θ} of $\operatorname{GL}_3(k_w)$ over $\overline{\mathbb{Q}}_p$. Via the identification (4.3.1) we have $\theta = \bigotimes_{j=1}^r \theta_j$, where $\theta_j : k_{w,n_j}^{\times} \to \overline{\mathbb{Q}}_p^{\times}$. We say that θ is *primitive* if for each j the Gal $(k_{w,n_j}/k_w)$ conjugates are pairwise distinct. By letting $\Theta(\theta_j)$ be the cuspidal representation of $\operatorname{GL}_{n_i}(k_w)$ associated to the primitive character θ_i via [Her09], Lemma 4.7, we have

$$R_T^{\theta} \cong (-1)^{3-r} \operatorname{Ind}_{P_{\underline{n}}(k_w)}^{\operatorname{GL}_3(k_w)} (\otimes_j \Theta(\theta_j)),$$

where $P_{\underline{n}}$ is the standard parabolic subgroup containing the Levi $\prod_{j} \operatorname{GL}_{n_{j}}$.

Let $F_{w,n} \stackrel{\text{def}}{=} W(k_{w,n})[\frac{1}{p}]$ be the unramified extension of F_w of degree n. We consider θ_j as a character on $\mathcal{O}_{F_{w,n_j}}^{\times}$ by inflation and let $\operatorname{Art}_{F_{w,n}}$ be the isomorphism $F_{w,n}^{\times} \xrightarrow{\sim} W_{F_{w,n}}^{\mathrm{ab}}$ of local class field theory, normalized so that geometric Frobenius elements correspond to uniformizers. We define the inertial type $\operatorname{rec}(\theta): I_{F_w} \to \overline{\mathbb{Q}}_p^{\times}$ as follows:

- (i) $\operatorname{rec}(\theta) \stackrel{\text{def}}{=} \bigoplus_{j=1}^{3} \theta_{j} \circ \operatorname{Art}_{F_{w}}^{-1} \text{ if } \theta_{j} : k_{w}^{\times} \to \overline{\mathbb{Q}}_{p}^{\times} \text{ for all } j = 1, 2, 3;$ (ii) $\operatorname{rec}(\theta) \stackrel{\text{def}}{=} \theta_{1} \circ \operatorname{Art}_{F_{w}}^{-1} \oplus \bigoplus_{\sigma \in \operatorname{Gal}(k_{w,2}/k_{w})} \sigma\left(\theta_{2} \circ \operatorname{Art}_{F_{w,2}}^{-1}\right) \text{ if } \theta_{1} : k_{w}^{\times} \to \overline{\mathbb{Q}}_{p}^{\times} \text{ and}$
- $\begin{aligned} \theta_2 : k_{w,2}^{\times} \to \overline{\mathbb{Q}}_p^{\times} \text{ are primitive characters;} \\ (\text{iii)} \quad \operatorname{rec}(\theta) \stackrel{\text{def}}{=} \bigoplus_{\sigma \in \operatorname{Gal}(k_{w,3}/k_w)} \sigma\left(\theta_1 \circ \operatorname{Art}_{F_{w,3}}^{-1}\right) \text{ if } \theta_1 : k_{w,3}^{\times} \to \overline{\mathbb{Q}}_p^{\times} \text{ is a primitive} \end{aligned}$

Lemma 4.3.2. Let V_w be a Serre weight at w for the Galois representation \overline{r} : $G_F \to \operatorname{GL}_3(\mathbb{F})$ and assume that $V_w \otimes_{\mathbb{F}} \overline{\mathbb{F}}_p$ is a Jordan-Hölder constituent of the mod-p reduction of a Deligne-Lusztig representation R_T^{θ} of $GL_3(k_w)$, where T is a maximal torus of $\operatorname{GL}_{3/k_w}$ and $\theta: T(k_w) \to \overline{\mathbb{Q}}_p^{\times}$ is a primitive character. Then $\overline{r}|_{G_{F_w}}$ has a potentially semistable lift with Hodge–Tate weights $\{-2, -1, 0\}$ and inertial type $rec(\theta)$.

Proof. This is proved in [MP], Theorem 5.5 (cf. also [EGH13], Proposition 2.4.1 and Proposition 7.4.4 in cases (i), (iii)), except when $n_i = 1$ for all *i* and some of the characters θ_i are equal. The exceptional case follows in the same way, by using the lemma below, which is an extension of [EGH13], Proposition 2.4.1(ii).

(We recall the idea: by the Deligne–Serre lifting lemma we obtain a global lift r of the dual \overline{r}^{\vee} which is attached to an automorphic representation in trivial weight and dual type $(R_T^{\theta})^{\vee}$. Classical local-global compatibility together with the lemma below shows that $r^{\vee}|_{G_{F_w}}$ provides the required lift.)

Suppose $n \geq 1$ and let B_n denote the Borel subgroup of GL_n of upper-triangular matrices. Let rec_{F_w} denote the local Langlands correspondence for $\operatorname{GL}_n(F_w)$ over $\overline{\mathbb{Q}}_p$, as in [EGH13], §1.3.

Lemma 4.3.3. Suppose that π is an irreducible admissible representation of $\operatorname{GL}_n(F_w)$ over $\overline{\mathbb{Q}}_p$. Assume that $\pi|_{\operatorname{GL}_n(\mathbb{O}_{F_w})}$ contains a Jordan–Hölder factor of the $\operatorname{GL}_n(\mathbb{O}_{F_w})$ inflation of a principal series representation of the form

(4.3.4)
$$\operatorname{Ind}_{B_n(k_w)}^{\operatorname{GL}_n(k_w)}(\theta_1 \otimes \cdots \otimes \theta_n)$$

for some characters $\theta_i : k_w^{\times} \to \overline{\mathbb{Q}}_p^{\times}$. Then $\operatorname{rec}_{F_w}(\pi)|_{I_{F_w}} \cong \bigoplus_{i=1}^n \theta_i \circ \operatorname{Art}_{F_w}^{-1}$.

Proof. We indicate how the proof of [EGH13], Proposition 2.4.1(ii) needs to be adjusted. Recall the pair (I, ρ) described in that proof, which has the property that $\operatorname{Ind}_{I}^{\operatorname{GL}_{n}(\mathcal{O}_{F_{w}})}\rho$ is isomorphic to the inflation of the representation (4.3.4) to $\operatorname{GL}_{n}(\mathcal{O}_{F_{w}})$. Thus by assumption $\operatorname{Hom}_{I}(\rho, \pi) \neq 0$. By [Roc98], Theorem 7.7 we deduce that π is a subquotient of $\operatorname{Ind}_{B_{n}(F_{w})}^{\operatorname{GL}_{n}(F_{w})}\tilde{\theta}$ for some character $\tilde{\theta} = \tilde{\theta}_{1} \otimes \cdots \otimes \tilde{\theta}_{n}$ with $\tilde{\theta}_{i}|_{\mathcal{O}_{F_{w}}^{\times}} = \theta_{i}$ for all i. Now from [Hen02], §1.9 it follows that if π' is another irreducible admissible representation of $\operatorname{GL}_{n}(F_{w})$, then π, π' have the same supercuspidal support iff $\operatorname{rec}_{F_{w}}(\pi)|_{W_{F_{w}}} \cong \operatorname{rec}_{F_{w}}(\pi')|_{W_{F_{w}}}$. The claim then follows from [HT01], §VII.2 (pp. 251f).

4.4. Serre weights of \overline{r} in the maximally non-split case. We keep the notation and assumptions of Section 4.2. In particular, p splits completely in F, there is a distinguished place $w \in S_p$, and $\overline{r} : G_F \to \operatorname{GL}_3(\mathbb{F})$ is an automorphic, absolutely irreducible Galois representation. The aim of this section is to almost completely determine the Serre weights of \overline{r} at w when $\overline{r}|_{G_{F_w}}$ is maximally non-split and generic. The techniques of this section follow the strategy of [EGH13], §5, and have been further pursued in [MP] to cover other cases when \overline{r} is ordinary at w.

Theorem 4.4.1. Assume that $\overline{r}|_{G_{F_w}}$ is maximally non-split and generic, of the form

$$\overline{r}|_{G_{F_w}} \sim \begin{pmatrix} \omega^{a+1} \mathrm{nr}_{\mu_2} & * & * \\ & \omega^{b+1} \mathrm{nr}_{\mu_1} & * \\ & & \omega^{c+1} \mathrm{nr}_{\mu_0} \end{pmatrix}.$$

(i) If $\operatorname{FL}(\overline{r}|_{G_{F_m}}) \notin \{0,\infty\}$ we have

$$\{F(a-1,b,c+1)\} \subseteq W_w(\bar{r}) \subseteq \\ \subseteq \{F(a-1,b,c+1), F(c+p-1,b,a-p+1)\}.$$

(ii) If $\operatorname{FL}(\overline{r}|_{G_{F_w}}) = \infty$ we have

$$\{F(a-1,b,c+1), F(a,c,b-p+1)\} \subseteq W_w(\bar{r}) \subseteq \\ \subseteq \{F(a-1,b,c+1), F(c+p-1,b,a-p+1), F(a,c,b-p+1)\}.$$

(iii) If
$$\operatorname{FL}(\overline{r}|_{G_{F_w}}) = 0$$
 we have

$$\{F(a-1,b,c+1), F(b+p-1,a,c)\} \subseteq W_w(\overline{r}) \subseteq \subseteq \{F(a-1,b,c+1), F(c+p-1,b,a-p+1), F(b+p-1,a,c)\}.$$

Remark 4.4.2. In fact, the proof shows that the containments in (i)–(iii) hold if we replace $W_w(\bar{r})$ by {Serre weights V at $w : S(V)_{\mathfrak{m}_{\bar{r}}} \neq 0$ } (for any group U and subset $\mathcal{P} \subseteq \mathcal{P}_U$ considered in §4.2).

In the remainder of this section we will prove this theorem, except for showing the existence of the shadow weights F(a, c, b - p + 1), F(b + p - 1, a, c) in parts (ii) and (iii). We will complete the proof in Proposition 4.5.10. We start with the following preliminary lemma.

Lemma 4.4.3. Let $V \in W_w(\overline{r})$ be a Serre weight of \overline{r} at w. Then V is isomorphic to one of the weights in the following list:

$$\begin{split} F(b+p-1,a-1,c+1), \ F(a-1,c+1,b-p+1), \ F(c+p,b,a-p), \\ F(b+p-1,a,c), \ F(a,c,b-p+1), \\ F(c+p-1,b,a-p+1), \ F(a-1,b,c+1). \end{split}$$

Proof. Let us write $V \cong F(x, y, z)$ for a *p*-restricted weight (x, y, z). By Frobenius reciprocity the weight F(x, y, z) is a constituent of the principal series $\operatorname{Ind}_{\overline{B}(k_w)}^{\overline{G}(k_w)} (\omega^x \otimes \omega^y \otimes \omega^z)$. Hence, by Lemma 4.3.2, $\overline{r}|_{G_{F_w}}$ admits a potentially semistable lift of Hodge–Tate weights $\{-2, -1, 0\}$ and inertial type $\theta \stackrel{\text{def}}{=} \widetilde{\omega}^x \oplus \widetilde{\omega}^y \oplus \widetilde{\omega}^z$. Proposition 2.6.1 implies $\{x, y, z\} \equiv \{a, b, c\}$ or $\{x, y, z\} \equiv \{a - 1, b, c + 1\}$ as subsets of $\mathbb{Z}/(p-1)\mathbb{Z}$.

If $x - z \ge p - 2$, we conclude by genericity that V belongs to the following list:

$$F(b+p-1, a-1, c+1), F(a-1, c+1, b-p+1), F(c+p, b, a-p),$$

 $F(b+p-1, a, c), F(a, c, b-p+1), F(c+p-1, b, a-p+1).$

If x - z , then noting that <math>x + 1, y, z - 1 are distinct modulo p - 1 by above, the proof of [EGH13], Lemma 6.1.1, shows that V is also a constituent of the principal series $\operatorname{Ind}_{\overline{B}(k_w)}^{\overline{G}(k_w)} (\omega^{z-1} \otimes \omega^y \otimes \omega^{x+1})$, hence $\{x + 1, y, z - 1\} \equiv \{a, b, c\}$ or $\{x + 1, y, z - 1\} \equiv \{a - 1, b, c + 1\}$ as subsets of $\mathbb{Z}/(p - 1)\mathbb{Z}$. We deduce by genericity that $V \cong F(a - 1, b, c + 1)$.

Proof of Theorem 4.4.1 (modulo the existence of shadows). We first establish the upper bound on $W_w(\bar{r})$ (this is sometimes called "weight elimination"). Let $V \in W_w(\bar{r})$. Then V is one of the weights appearing in the list of Lemma 4.4.3. We now prove that the weights F(a-1,c+1,b-p+1), F(c+p,b,a-p) and F(b+p-1,a-1,c+1) cannot be Serre weights of \bar{r} at w. Indeed, by the proof of [Her09], Proposition 7.4, it is easily checked that $F(b+p-1,a-1,c+1) \otimes_{\mathbb{F}} \overline{\mathbb{F}}_p$ is an element of $JH(\overline{R}^{\theta}_T)$, where the maximal torus T verifies $T(k_w) \cong k_{w,2}^{\times} \times k_w^{\times}$ and $\theta \cong \tilde{\omega}_2^{a-1+pb} \otimes \tilde{\omega}^{c+1}$. If F(b+p-1,a-1,c+1) is a Serre weight of \bar{r} then by Lemma 4.3.2, $\bar{r}|_{G_{F_w}}$ has a potentially semistable lift of Hodge–Tate weights $\{-2, -1, 0\}$ and inertial type $\tilde{\omega}_2^{a-1+pb} \oplus \tilde{\omega}_2^{b+p(a-1)} \oplus \tilde{\omega}^{c+1}$, which is not possible by Proposition 2.6.3.

In a similar fashion, the weights $F(a-1,c+1,b-p+1)\otimes_{\mathbb{F}}\overline{\mathbb{F}}_p$, $F(c+p,b,a-p)\otimes_{\mathbb{F}}\overline{\mathbb{F}}_p$ appear in $\operatorname{JH}(\overline{R}_T^{\theta})$, where now $\theta \cong \widetilde{\omega}_2^{b+p(c+1)} \otimes \widetilde{\omega}^{a-1}$, providing a contradiction with Proposition 2.6.3.

Assume now that $\operatorname{FL}(\overline{r}|_{G_{F_w}}) \neq \infty$; we claim that the weight F(a, c, b - p + 1) cannot be a Serre weight of \overline{r} at w. Again, using [Her09], proof of Proposition 7.4, we see that $F(a, c, b - p + 1) \otimes_{\mathbb{F}} \overline{\mathbb{F}}_p$ is an element of $\operatorname{JH}(\overline{R}_T^{\theta})$, where now $\theta \cong \widetilde{\omega}_2^{a+1+p(b-1)} \otimes \widetilde{\omega}^c$. By Lemma 4.3.2, $\overline{r}|_{G_{F_w}}$ has a potentially semistable lift of Hodge–Tate weights $\{-2, -1, 0\}$ and inertial type $\widetilde{\omega}_2^{a+1+p(b-1)} \oplus \widetilde{\omega}_2^{b-1+p(a+1)} \oplus \widetilde{\omega}^c$, which implies $\operatorname{FL}(\overline{r}|_{G_{F_w}}) = \infty$ by Proposition 2.6.4. By duality, F(b + p - 1, a, c) cannot be a Serre weight of \overline{r} at w if $\operatorname{FL}(\overline{r}|_{G_{F_w}}) \neq 0$.

By weight cycling we will now deduce that $F(a-1,b,c+1) \in W_w(\bar{r})$. (We refer to the introduction of [EGH13] for the term "weight cycling" as well as its history.) Recall the (commuting) Hecke operators \overline{T}_1 , \overline{T}_2 at w defined in [EGH13], §4.2. They act on the finite-dimensional \mathbb{F} -vector space S(V) for any Serre weight V at w. If \overline{T}_i has a non-zero eigenvalue on $S(F(c+p-1,b,a-p+1))_{\mathfrak{m}_{\overline{r}}}$ for some i, we deduce exactly as in the proof of [EGH13], Corollary 4.5.4 that $\overline{r}|_{G_{F_w}}$ has a crystalline lift ρ over E, where ρ admits a subrepresentation of Hodge–Tate weight -(c+p+1) or a quotient of Hodge–Tate weight -(a-p+1). This implies that $\overline{r}|_{I_{F_w}}$ admits ω^{c+2} as subrepresentation or ω^a as quotient, contradicting our assumptions on $\overline{r}|_{G_{F_w}}$. Hence \overline{T}_i act nilpotently on $S(V)_{\mathfrak{m}_{\overline{r}}}$ for V = F(c+p-1,b,a-p+1). By a similar argument the same holds when V = F(b+p-1,a,c) or F(a,c,b-p+1).

Suppose that $F(a-1,b,c+1) \notin W_w(\bar{r})$. If $V \stackrel{\text{def}}{=} F(c+p-1,b,a-p+1) \in W_w(\bar{r})$, then $S(V)_{\mathfrak{m}_{\overline{r}}} \neq 0$ (for suitable U, \mathcal{P}). By the previous paragraph we may apply weight cycling ([EGH13], Proposition 6.1.3(ii)) to the smooth $\operatorname{GL}_3(F_w)$ -representation $\pi \stackrel{\text{def}}{=} S(U^v, V')_{\mathfrak{m}_{\overline{r}}}$ (defined just after (4.5.1)) and our weight V. Using the nilpotency of \overline{T}_1 (resp. \overline{T}_2) we see that $S(V'')_{\mathfrak{m}_{\overline{r}}} \neq 0$, hence $V'' \in W_w(\bar{r})$, for at least one weight V'' listed there. However, the genericity of $\overline{r}|_{G_{F_w}}$ together with our upper bound for $W_w(\bar{r})$ lets us rule out four out of five weights in each case and deduce that $F(a,c,b-p+1) \in W_w(\bar{r})$ (resp. $F(b+p-1,a,c) \in W_w(\bar{r})$). But this contradicts the upper bound on $W_w(\bar{r})$ we obtained above. Similarly, if $F(b+p-1,a,c) \in W_w(\bar{r})$, then weight cycling with \overline{T}_2 and genericity imply that $W_w(\bar{r})$ contains $F(a,c,b-p+1) \in W_w(\bar{r})$. We conclude that $F(a-1,b,c+1) \in W_w(\bar{r})$.

This ends the proof of Theorem 4.4.1 modulo the existence of shadow weights.

Corollary 4.4.4. In the setting of Theorem 4.4.1, \overline{r} has an automorphic lift $r : G_F \to \operatorname{GL}_3(\mathcal{O}_E)$ (after possibly enlarging E) such that $r|_{G_{F_w}}$ is crystalline and ordinary of Hodge–Tate weights $\{-a-1, -b-1, -c-1\}$.

Proof. Keeping our notation of the proof of Theorem 4.4.1, consider the Hecke operators \overline{T}_1 , \overline{T}_2 acting on $S(F(a-1,b,c+1))_{\mathfrak{m}_{\overline{r}}}$, which is non-zero by the proof above. If \overline{T}_1 fails to be invertible on this vector space, then one of the weights F(c+p,a-1,b), F(a-1,c+1,b-p+1) is contained in $W_w(\overline{r})$ by [EGH13], Proposition 6.1.3(i). This contradicts the upper bound of Theorem 4.4.1 by genericity, hence \overline{T}_1 is invertible on $S(F(a-1,b,c+1))_{\mathfrak{m}_{\overline{r}}}$. The same is true for \overline{T}_2 . Enlarging E if necessary we can pick simultaneous eigenvalues $\alpha_i \in \mathbb{F}^{\times}$ of \overline{T}_i (i = 1, 2) on this

space. Consider the double coset operators $T_w^{(i)}$ (i = 1, 2) defined as in Section 4.1 but for our fixed place w|p and let $\widetilde{T}_w^{(1)} \stackrel{\text{def}}{=} p^{-c-1}T_w^{(1)}$, $\widetilde{T}_w^{(2)} \stackrel{\text{def}}{=} p^{-b-c-1}T_w^{(2)}$. Let $\mathbb{T} \stackrel{\text{def}}{=} \mathbb{T}^{\mathcal{P}}[\widetilde{T}_w^{(1)}, \widetilde{T}_w^{(2)}]$ and consider the homomorphism $\theta : \mathbb{T} \to \mathbb{F}$ whose kernel contains $\mathfrak{m}_{\overline{r}}$ and which sends $\widetilde{T}_w^{(i)}$ to α_i (i = 1, 2). Redefining $a_w = (a - 1, b, c + 1)$, the construction in [EGH13], §4.1.1 (replacing the coefficients \mathbb{Z}_p by \mathcal{O}_E) gives a finite free \mathcal{O}_E -module W_{a_w} with a linear action of $\operatorname{GL}_3(\mathcal{O}_{F_w})$. As a_w lies in the lowest alcove we have $S(W_{a_w}) \otimes_{\mathcal{O}_E} \mathbb{F} \cong S(F(a - 1, b, c + 1))$. The commutative \mathcal{O}_E -algebra \mathbb{T} acts naturally on both sides, with $\widetilde{T}_w^{(i)}$ acting as \overline{T}_i on the right-hand side (see [EGH13], §4.4). Using the Deligne–Serre lifting lemma and enlarging E if necessary we can lift the Hecke eigenvalues θ to $S(W_{a_w})$ and obtain an automorphic lift $r : G_F \to \operatorname{GL}_3(\mathcal{O}_E)$ of \overline{r} such that $r|_{G_{F_w}}$ is crystalline of Hodge–Tate weights $\{-a - 1, -b - 1, -c - 1\}$. It is moreover ordinary, i.e. its Hodge and Newton polygons coincide, using that $\alpha_i \neq 0$ for i = 1, 2 (cf. the proof of [EGH13], Corollary 4.5.4).

4.5. Local-global compatibility. We keep the notation and assumptions of Section 4.2. Suppose that \bar{r} is as in Theorem 4.4.1. We recall the operators $S, S' \in \mathbb{F}[GL_3(\mathbb{F}_p)]$ obtained from (3.1.1) by setting $(a_2, a_1, a_0) = (-c, -b, -a)$:

(4.5.1)
$$S \stackrel{\text{def}}{=} \sum_{x,y,z \in \mathbb{F}_p} x^{p-(a-c)} z^{p-(a-b)} \begin{pmatrix} 1 & x & y \\ & 1 & z \\ & 1 \end{pmatrix} \dot{w}_0,$$
$$S' \stackrel{\text{def}}{=} \sum_{x,y,z \in \mathbb{F}_p} x^{p-(b-c)} z^{p-(a-c)} \begin{pmatrix} 1 & x & y \\ & 1 & z \\ & & 1 \end{pmatrix} \dot{w}_0.$$

As the compact open subgroup U we fixed is unramified at v, we can write $U = \mathcal{G}(\mathcal{O}_{F_v^+})U^v$ for some compact open subgroup $U^v \leq G(\mathbb{A}_{F^+}^{\infty,v})$. If W is an \mathcal{O}_{E^+} module endowed with an action of $\prod_{v' \in S_p^+ \setminus \{v\}} \mathcal{G}(\mathcal{O}_{F_{v'}^+})$ we define $S(U^v, W) \stackrel{\text{def}}{=} \lim_{v \to U_v} S(U^v \cdot U_v, W)$, where the limit runs over all compact open subgroups $U_v \leq \mathcal{G}(\mathcal{O}_{F_v^+})$. Then $S(U^v, W)$ has a smooth action of $G(F_v^+)$ and hence via ι_w of $\operatorname{GL}_3(F_w) = \operatorname{GL}_3(\mathbb{Q}_p)$. We also define $\widetilde{S}(U^v, W) \stackrel{\text{def}}{=} \varprojlim_s S(U^v, W/\varpi_E^s)$, which has a linear action of $\operatorname{GL}_3(\mathbb{Q}_p)$. Note that $\widetilde{S}(U^v, W)$ is isomorphic to the p-adic completion of $S(U^v, W)$, provided W is finitely generated.

We recall that in Section 3 we defined an Iwahori subgroup I of $K = \operatorname{GL}_3(\mathbb{Z}_p)$ and the notation $V^{I,(a_2,a_1,a_0)}$ for any representation V of K over \mathcal{O}_E and any triple $(a_2,a_1,a_0) \in \mathbb{Z}^3$. We note that from the definitions it follows that $\widetilde{S}(U^v, \widetilde{V}')^{I,(-b,-c,-a)} \cong$ $S(U^v, \widetilde{V}')^{I,(-b,-c,-a)}$, a finite free \mathcal{O}_E -module. Recall furthermore that $\Pi = \begin{pmatrix} 1 \\ 0 \end{pmatrix}$.

Theorem 4.5.2. We make the following assumptions:

(i) $\overline{r}|_{G_{F_{w}}}$ is maximally non-split and generic, of the form

$$\overline{r}|_{G_{F_w}} \sim \begin{pmatrix} \omega^{a+1} \mathrm{nr}_{\mu_2} & * & * \\ & \omega^{b+1} \mathrm{nr}_{\mu_1} & * \\ & & \omega^{c+1} \mathrm{nr}_{\mu_0} \end{pmatrix}$$

- (ii) $\operatorname{FL}(\overline{r}|_{G_{F_w}}) \notin \{0, \infty\}.$
- (iii) The \mathcal{O}_E -dual of $S(U^v, \widetilde{V}')^{I,(-b,-c,-a)}_{\mathfrak{m}_{\overline{\tau}}}$ is free over \mathbb{T} , where \mathbb{T} denotes the \mathcal{O}_E -subalgebra of End $\left(S(U^v, \widetilde{V}')^{I,(-b,-c,-a)}_{\mathfrak{m}_{\overline{\tau}}}\right)$ generated by $\mathbb{T}^{\mathcal{P}}$, U_1 , and U_2 .

Then we have the equality

(4.5.3)
$$S' \circ \Pi = (-1)^{a-b} \cdot \frac{b-c}{a-b} \cdot \operatorname{FL}(\overline{r}|_{G_{F_w}}) \cdot S$$

of maps

$$S(U^{v},V')[\mathfrak{m}_{\overline{r}}]^{I,(-b,-c,-a)}[U_{1},U_{2}] \to S(U^{v},V')[\mathfrak{m}_{\overline{r}}]^{I,(-c-1,-b,-a+1)}.$$

Moreover, these maps are injective with non-zero domain. In particular, $\operatorname{FL}(\overline{r}|_{G_{F_w}})$ is determined by the smooth $\operatorname{GL}_3(\mathbb{Q}_p)$ -representation $S(U^v, V')[\mathfrak{m}_{\overline{r}}]$.

Remark 4.5.4. The proof shows that for any non-zero $v \in S(U^v, V')[\mathfrak{m}_{\overline{r}}]^{I,(-b,-c,-a)}[U_1, U_2]$, the K-subrepresentation generated by v and Πv is of the form

$$F(-c-1,-b,-a+1) - \left(F(-b+p-1,-c,-a) \oplus F(-c,-a,-b-p+1)\right)$$

with socle F(-c-1, -b, -a+1). Together with Proposition 3.1.2 this explains why there exists a constant $x \in \mathbb{F}^{\times}$ such that $S' \circ \Pi = xS$ on $S(U^v, V')[\mathfrak{m}_{\overline{r}}]^{I,(-b,-c,-a)}[U_1, U_2]$.

Proof. Let $\eta: I \to \mathcal{O}_E^{\times}$ denote the character $\widetilde{\omega}^b \otimes \widetilde{\omega}^c \otimes \widetilde{\omega}^a$ and set

$$M \stackrel{\mathrm{def}}{=} S(U^v, \widetilde{V}')^{I, (-b, -c, -a)}_{\mathfrak{m}_{\overline{r}}} \cong S\big(U^v \cdot I, \eta \otimes \widetilde{V}'\big)_{\mathfrak{m}_{\overline{r}}}$$

We remark that $M \neq 0$, as $S(F(a-1,b,c+1))_{\mathfrak{m}_{\overline{r}}} \neq 0$ and F(a-1,b,c+1) is a Jordan–Hölder factor of $\overline{\operatorname{Ind}_{I}^{K}\eta}$, and also that M is a finite free \mathcal{O}_{E} -module. For any \mathcal{O}_{E} -algebra A we let $M_{A} \stackrel{\text{def}}{=} M \otimes_{\mathcal{O}_{E}} A$ and $\mathbb{T}_{A} \stackrel{\text{def}}{=} \mathbb{T} \otimes_{\mathcal{O}_{E}} A$.

Picking a \mathbb{Q}_p -linear embedding $E \hookrightarrow \overline{\mathbb{Q}}_p$, as well as an isomorphism $i : \overline{\mathbb{Q}}_p \xrightarrow{\sim} \mathbb{C}$, we see that

(4.5.5)
$$M_{\overline{\mathbb{Q}}_p} \cong \bigoplus_{\pi} m(\pi) \cdot \pi_v^{I,(-b,-c,-a)} \otimes (\pi^{\infty,v})^{U^v},$$

where the sum runs over irreducible representations $\pi \cong \pi_{\infty} \otimes \pi_v \otimes \pi^{\infty,v}$ of $G(\mathbb{A}_{F^+})$ such that $\pi \otimes_i \mathbb{C}$ is a cuspidal automorphic representation of multiplicity $m(\pi) \in \mathbb{Z}_{>0}$ with $\pi_{\infty} \otimes_i \mathbb{C}$ determined by the algebraic representation $(\tilde{V}')^{\vee}$ and with Galois representation r_{π} lifting \bar{r}^{\vee} (cf. [EGH13], Lemma 7.1.6). For any π contributing to (4.5.5) we have

- (a) $\pi_v \cong \operatorname{Ind}_{B(\mathbb{Q}_p)}^{G(\mathbb{Q}_p)} \left(\psi_b \left| \cdot \right|^2 \otimes \psi_c \left| \cdot \right| \otimes \psi_a \right)$ for some smooth characters $\psi_i : \mathbb{Q}_p^{\times} \to \overline{\mathbb{Q}}_p^{\times}$ (depending on π) with $\psi_i|_{\pi^{\times}} = \widetilde{\omega}^{-i}$ for $i \in \{a, b, c\}$:
- $$\begin{split} \overline{\mathbb{Q}}_p^{\times} & (\text{depending on } \pi) \text{ with } \psi_i|_{\mathbb{Z}_p^{\times}} = \widetilde{\omega}^{-i} \text{ for } i \in \{a, b, c\}; \\ \text{(b)} & r_{\pi}^{\vee}|_{G_{F_w}} \text{ is potentially crystalline with Hodge-Tate weights } \{-2, -1, 0\} \text{ and } \\ & \mathrm{WD}(r_{\pi}^{\vee}|_{G_{F_w}})^{\mathrm{F-ss}} \cong \psi_b^{-1} \oplus \psi_c^{-1} \oplus \psi_a^{-1}. \end{split}$$

Here, part (a) follows from [EGH13], Propositions 2.4.1 and 7.4.4 and (b) follows from classical local-global compatibility. In particular from (b) we deduce that the φ -eigenvalue on $\mathbb{D}_{\mathrm{st}}^{\mathbb{Q}_p,2}(r_{\pi}^{\vee}|_{G_{Fw}})^{I_{Fw}=\widetilde{\omega}^b}$ equals $p^2\psi_b(p)^{-1}$, so from Theorem 2.5.1 and assumptions (i), (ii) it follows that

(4.5.6)
$$\operatorname{ord}_p(\psi_b(p)) = 1 \quad \text{and} \quad \operatorname{FL}(\overline{r}|_{G_{F_w}}) = \left[\frac{\psi_b(p)}{p}\right].$$

We also deduce from Corollary 2.4.11 that $\operatorname{ord}_p(p^2\psi_i(p)^{-1}) \in (0,2)$ for $i \in \{a, b, c\}$, with sum equal to 3. In particular, the eigenvalue of U_1 (resp. U_2) on $\pi_v^{I,(-b,-c,-a)}$, which equals $p^2\psi_b(p)^{-1}$ (resp. $p^3\psi_b(p)^{-1}\psi_c(p)^{-1}$) by Lemma 3.2.8, has positive valuation. (Note for later reference that this is true even if $\operatorname{FL}(\overline{r}|_{G_{Fw}}) \in \{0,\infty\}$.) Note that the image \mathbb{T}_a of $\mathbb{T}^{\mathcal{P}}$ in $\operatorname{End}(M)$ is local with maximal ideal the image of $\mathfrak{m}_{\overline{r}}$. As $\mathbb{T}_a \subseteq \mathbb{T}$ is a finite ring extension, $\mathfrak{m}_{\overline{r}} \subseteq \operatorname{rad}(\mathbb{T})$. Since all eigenvalues of U_i on $M_{\overline{\mathbb{Q}}_p}$ have positive valuation, we deduce that $U_i \in \operatorname{rad}(\mathbb{T})$ (i = 1, 2). It then follows that \mathbb{T} is local with maximal ideal \mathfrak{m} generated by $\mathfrak{m}_{\overline{r}}$, U_1 , and U_2 . As $\mathbb{T}_{\overline{\mathbb{Q}}_p}$ acts faithfully and semisimply on $M_{\overline{\mathbb{Q}}_p}$ (by (4.5.5)), we see that $\mathbb{T}_{\overline{\mathbb{Q}}_p}$ is semisimple, hence reduced. In particular, \mathbb{T} is reduced.

As U is sufficiently small, our assumptions as well as the statement we want to prove is insensitive to a finite base change $E \to E'$. Hence, by passing to a finite extension of E, we may assume that $\mathbb{T}_E \cong E^r$ for some r > 0.

We have $M_E = \bigoplus_{\mathfrak{p}} M_E[\mathfrak{p}_E]$, where the sum runs over the minimal primes of \mathbb{T} and $\mathfrak{p}_E \stackrel{\text{def}}{=} \mathfrak{p}\mathbb{T}_E$. Note that, by the above, $\mathbb{T}_E/\mathfrak{p}_E = E$ for any such \mathfrak{p} . Then $M_E[\mathfrak{p}_E] \otimes_E \overline{\mathbb{Q}}_p$ is a direct summand of (4.5.5), where π runs over a subset of the automorphic representations in (4.5.5), and we claim that each π_v occurring in this direct summand is the same. To see this, note that r_{π} is determined by \mathfrak{p} , by using Cebotarev density and classical local-global compatibility at the places in \mathcal{P} . (Note that classical local-global compatibility, which is known only up to Frobenius-semisimplification, determines $\operatorname{tr}(r_{\pi}|_{G_{F_{w'}}})$ for $w' \in \mathcal{P}$ and that $\operatorname{tr}(r_{\pi})$ determines r_{π} .) Then we deduce the claim from classical local-global compatibility at w.

By Proposition 3.2.2 we have

(4.5.7)
$$\widehat{S}' \circ \Pi = \frac{\psi_b(p)}{p} \kappa \, \widehat{S}$$

on $M_E[\mathfrak{p}_E]$, where $\kappa \in \mathbb{Z}_p^{\times}$ is such that $\kappa \equiv (-1)^{a-b} \cdot \frac{b-c}{a-b} \mod p$ and where $\widehat{S}, \ \widehat{S}' \in \mathcal{O}_E[\mathrm{GL}_3(\mathbb{F}_p)]$ are obtained from (3.2.1) by setting $(a_2, a_1, a_0) = (-c, -b, -a)$.

Let $M^{d} \stackrel{\text{def}}{=} \operatorname{Hom}_{\mathcal{O}_{E}}(M, \mathcal{O}_{E})$, which is finite free over \mathbb{T} by assumption (iii). Fix any minimal prime \mathfrak{p} of \mathbb{T} . As $\mathbb{T}_{E}/\mathfrak{p}_{E} = E$ we have $\mathbb{T}/\mathfrak{p} = \mathcal{O}_{E}$, so in particular M^{d}/\mathfrak{p} is a finite free \mathcal{O}_{E} -module. The identity (4.5.7) holds also on $M_{E}^{d}/\mathfrak{p}_{E} \cong$ $\operatorname{Hom}_{E}(M_{E}[\mathfrak{p}_{E}], E)$ and on its submodule M^{d}/\mathfrak{p} . By (4.5.7) and (4.5.6), and since $\mathfrak{p} + \varpi_{E}\mathbb{T} = \mathfrak{m}$, we deduce the identity (4.5.3) on $(M^{d}/\mathfrak{p}) \otimes_{\mathcal{O}_{E}} \mathbb{F} \cong M^{d}/\mathfrak{m}$, hence also on $M_{\mathbb{F}}[\mathfrak{m}] \cong \operatorname{Hom}_{\mathbb{F}}(M^{d}/\mathfrak{m}, \mathbb{F})$. Now observe that $M_{\mathbb{F}} \cong S(U^{v}, V')_{\mathfrak{m}_{\mathcal{F}}}^{I,(-b,-c,-a)}$ and hence $M_{\mathbb{F}}[\mathfrak{m}] \cong S(U^{v}, V')[\mathfrak{m}_{\mathcal{F}}]^{I,(-b,-c,-a)}[U_{1}, U_{2}].$

We claim that $U_i\Pi(M_{\mathbb{F}}[\mathfrak{m}]) = 0$ for i = 1, 2. Let $N \stackrel{\text{def}}{=} S(U^v, \widetilde{V}')^{I,(-c,-a,-b)}_{\mathfrak{m}_{\mathbb{F}}}$ and let $\mathbb{T}' \subseteq \operatorname{End}(N)$ be the \mathcal{O}_E -subalgebra generated by $\mathbb{T}^{\mathcal{P}}$, U_1 , and U_2 . By above, $\Pi(M[\mathfrak{p}]) \subseteq N[\mathfrak{p}']$ for the minimal prime \mathfrak{p}' of \mathbb{T}' lying over the same prime of $\mathbb{T}^{\mathcal{P}}$ as \mathfrak{p} such that moreover the image of U_i in $\mathbb{T}'/\mathfrak{p}' = \mathcal{O}_E$ equals the eigenvalue of U_i on $\pi_v^{I,(-c,-a,-b)}$. (In fact, $\Pi(M[\mathfrak{p}]) = N[\mathfrak{p}']$, by reversing the argument.) From the above, the natural injection $M[\mathfrak{p}] \otimes_{\mathcal{O}_E} \mathbb{F} \hookrightarrow M_{\mathbb{F}}[\mathfrak{m}]$ is an isomorphism, so $\Pi(M_{\mathbb{F}}[\mathfrak{m}])$ is killed by the unique maximal ideal \mathfrak{m}' of \mathbb{T}' , which contains U_1, U_2 by the same argument as for \mathbb{T} . This proves the claim. (In fact, if $\operatorname{Hom}_{\mathcal{O}_E}(N, \mathcal{O}_E)$ is moreover free as \mathbb{T}' -module, which is true under the hypotheses of Theorem 5.1.1, then we even get that Π induces an isomorphism $M_{\mathbb{F}}[\mathfrak{m}] \xrightarrow{\sim} N_{\mathbb{F}}[\mathfrak{m}']$.)

Now by Theorem 4.4.1(i) and [EGH13], Lemma 7.4.3 any irreducible K-subrepresentation of $S(U^v, V')_{\mathfrak{m}_{\overline{\tau}}}$ is isomorphic to F(-c-1, -b, -a+1) or F(-a+p-1, -b, -c-p+1). Thus by Corollary 3.1.12(i), if $v \in M_{\mathbb{F}}[\mathfrak{m}]$ is non-zero, then $\langle Kv \rangle_{\mathbb{F}}$ is the unique quotient of $\operatorname{Ind}_I^K (\omega^{-b} \otimes \omega^{-c} \otimes \omega^{-a})$ with socle F(-c-1, -b, -a+1). By [Le], Proposition 2.2.2 we see that $\langle Kv \rangle_{\mathbb{F}}$ is the uniserial length 2 representation of shape F(-c-1, -b, -a+1)—F(-b+p-1, -c, -a). A similar argument shows that $\langle K \Pi v \rangle_{\mathbb{F}}$ is the uniserial length 2 representation of shape F(-c-1, -b, -a + b)1)—F(-c, -a, -b - p + 1). In particular, $Sv \neq 0$ by Proposition 3.1.2.

Remark 4.5.8. Theorem 4.5.2 holds equally well if we replace \mathbb{T} in the statement and proof by the subalgebra $\mathbb{T}_2 \subseteq \mathbb{T}$ generated by $\mathbb{T}^{\mathcal{P}}$ and U_2 . Assumption (iii) will be verified in Section 5 both for \mathbb{T}_2 and for \mathbb{T} , under suitable hypotheses. The same comments apply for the subalgebra $\mathbb{T}_a \subseteq \mathbb{T}$, provided $W_w(\bar{r}) = \{F(a-1, b, c+1)\}$ (see Remark 5.3.5).

Remark 4.5.9. Theorem 4.5.2 holds if we replace assumption (iii) by either of the following two statements:

(iii')
$$\dim_{\mathbb{F}} S(U^{v}, V')[\mathfrak{m}_{\overline{r}}]^{I, (-b, -c, -a)}[U_{1}, U_{2}] = 1;$$

(iii'')
$$\dim_{\mathbb{F}} S(U^v, V')[\mathfrak{m}_{\overline{r}}]^{I,(-b,-c,-a)} = 1$$

It suffices to note that $(iii'') \Rightarrow (iii') \Rightarrow (iii)$. The first implication is obvious. For the second, we have $\dim_{\mathbb{F}} M^{\mathrm{d}}/\mathfrak{m} = \dim_{\mathbb{F}} M_{\mathbb{F}}[\mathfrak{m}] = 1$, in the notation of the proof. As in the proof, we may assume that $\mathbb{T}/\mathfrak{m} = \mathbb{F}$. By Nakayama's lemma, we have a surjective \mathbb{T} -linear map $\mathbb{T} \to M^d$, which has to be an isomorphism, as M^d is a faithful T-module. (Note that assumption (iii'') even implies that $\mathbb{T}_a = \mathbb{T}$.)

We now establish the existence of shadow weights in Theorem 4.4.1, thereby completing the proof of that theorem.

Proposition 4.5.10. Suppose \overline{r} satisfies assumption (i) of Theorem 4.5.2.

- $\begin{array}{ll} \text{(i)} & \textit{If} \ \mathrm{FL}(\overline{r}|_{G_{F_w}}) = \infty \ \textit{then} \ F(a,c,b-p+1) \in W_w(\overline{r}).\\ \text{(ii)} & \textit{If} \ \mathrm{FL}(\overline{r}|_{G_{F_w}}) = 0 \ \textit{then} \ F(b+p-1,a,c) \in W_w(\overline{r}). \end{array}$

Remark 4.5.11. Suppose that \overline{r} is as in Proposition 4.5.10(i). Then the proof below shows that for some non-zero $v \in S(U^v, V')[\mathfrak{m}_{\overline{r}}]^{I,(-b,-c,-a)}[U_1, U_2]$, the Ksubrepresentation generated by v and Πv is of the form

$$F(-b+p-1, -c, -a) \oplus (F(-c-1, -b, -a+1) - F(-c, -a, -b-p+1)).$$

In particular, Sv = 0, $S' \Pi v \neq 0$. Similar remarks apply in case $FL(\bar{r}|_{G_{F_w}}) = 0$.

Proof. (i) We follow the proof of Theorem 4.5.2, noting that with our current assumption only three things change: (a) we get $\operatorname{ord}_p(\psi_b(p)) < 1$ instead of (4.5.6), (b) the natural injection $M[\mathfrak{p}] \otimes_{\mathcal{O}_E} \mathbb{F} \hookrightarrow M_{\mathbb{F}}[\mathfrak{m}]$ need not be an isomorphism, and (c) we only know that $\langle Kv \rangle_{\mathbb{F}}$ (for non-zero $v \in M_{\mathbb{F}}[\mathfrak{m}]$) is a quotient of the uniserial representation of shape F(-c-1, -b, -a+1)—F(-b+p-1, -c, -a). From (a) we deduce that $\widehat{S} = \frac{p}{\psi_b(p)} \kappa^{-1} \widehat{S}' \circ \Pi$ on $M_E[\mathfrak{p}_E]$, so Sv = 0 for some non-zero $v \in M_{\mathbb{F}}[\mathfrak{m}]$ (but maybe not all, see (b)). Then (c) together with Proposition 3.1.2 implies that $(Kv)_{\mathbb{F}} \cong F(-b+p-1,-c,-a)$ for such v. In particular, $F(a,c,b-p+1) \in W_w(\overline{r})$. Part (ii) is analogous.

5. Freeness over the Hecke Algebra

In this section, we prove Theorem 5.1.1, which states that the dual

$$\operatorname{Hom}_{\mathcal{O}_E}(S(U^v, \widetilde{V}')^{I, (-b, -c, -a)}_{\mathfrak{m}_{\overline{v}}}, \mathfrak{O}_E)$$

of the space of automorphic forms is free over a Hecke algebra under certain conditions on U^v and \overline{r} (and \widetilde{V}' and $\mathfrak{m}_{\overline{r}}$ are as defined in Section 4.2).

5.1. The setup. As before, let F/\mathbb{Q} be a CM field in which p splits completely, and let F^+ be its maximal totally real subfield. Assume moreover:

(i) F/F^+ is unramified at all finite places.

Fix a place w|p of F, and let $v \stackrel{\text{def}}{=} w|_{F^+}$. Let $\overline{r} : G_F \to \text{GL}_3(\mathbb{F})$ be a Galois representation with $\overline{r}|_{G_{F_w}}$ maximally non-split and generic as in Theorem 4.5.2(i), satisfying the following additional properties:

- (ii) \overline{r} is unramified at all finite places not dividing p;
- (iii) \overline{r} is Fontaine–Laffaille and regular at all places dividing p;
- (iv) \overline{r} has image containing $\operatorname{GL}_3(k)$ for some $k \subseteq \mathbb{F}$ with #k > 9;
- (v) $\overline{F}^{\ker \operatorname{ad} \overline{r}}$ does not contain $F(\zeta_p)$.

Note that condition (iv), which is stronger than the usual condition of adequacy (see Definition 2.3 of [Tho12]), allows us to choose a finite place v_1 of F^+ which is prime to p satisfying the following properties (see Section 2.3 of [CEG⁺16]):

- $\circ v_1$ splits in F as $v_1 = w_1 w_1^c$;
- v_1 does not split completely in $F(\zeta_p)$;
- $\overline{r}(\operatorname{Frob}_{w_1})$ has distinct \mathbb{F} -rational eigenvalues, no two of which have ratio $(\mathbf{N}v_1)^{\pm 1}$.

We choose a unitary group $G_{/F^+}$ and a model $\mathcal{G}_{/\mathcal{O}_{F^+}}$ as in §4.1. We note that G is automatically quasi-split, hence unramified by (i), at all finite places, as we are in odd rank. (See the proof of [BC09], Lemma 6.2.4.) Let $U^v = \prod_{i \neq i} U_{v'} \leq G(\mathbb{A}_{F^+}^{\infty,v})$

be a compact open subgroup satisfying the following properties:

- (vi) $U_{v'} = \mathcal{G}(\mathcal{O}_{F_{v'}^+})$ for all places v' which split in F other than v_1 and those dividing p;
- (vii) U_{v_1} is the preimage of the upper-triangular matrices under the map

$$\mathfrak{G}(\mathfrak{O}_{F_{v_1}^+}) \to \mathfrak{G}(k_{v_1}) \xrightarrow{\sim}_{\iota_{w_1}} \mathrm{GL}_3(k_{w_1});$$

(viii) $U_{v'}$ is a hyperspecial maximal compact open subgroup of $G(F_{v'}^+)$ if v' is inert in F.

The choice of U_{v_1} implies that $U^v U_v$ is sufficiently small in the sense of Section 4.1 for any compact open subgroup U_v of $G(F_v^+)$.

As in Section 4.2 suppose that for each $w' \in S_p \setminus \{w, w^c\}$ we are given a *p*-restricted triple $a_{w'} = (a_{w',2}, a_{w',1}, a_{w',0}) \in \mathbb{Z}^3$ such that $a_{w',i} + a_{w'^c,2-i} = 0$ for all $0 \leq i \leq 2$ and all $w' \in S_p \setminus \{w, w^c\}$. Suppose moreover that $a_{w',2} - a_{w',0} , for all <math>w' \in S_p \setminus \{w, w^c\}$, a slightly stronger condition than before. Recall that we defined $V' = \bigotimes_{v' \in S_p^+ \setminus \{v\}} F_{a_{v'}}$ and $\widetilde{V}' = \bigotimes_{v' \in S_p^+ \setminus \{v\}} W_{a_{v'}}$. Let \mathcal{P} denote the set of finite places w' of F that split over F^+ and do not divide

Let \mathcal{P} denote the set of finite places w' of F that split over F^+ and do not divide p or v_1 , and define the maximal ideal $\mathfrak{m}_{\overline{r}}$ of $\mathbb{T}^{\mathcal{P}}$ as in Section 4.2. We make the additional automorphy assumption:

(ix) $S(U^v, V')_{\mathfrak{m}_{\overline{r}}}$ is non-zero.

By Theorem 4.4.1 this implies $S(U^v \mathcal{G}(\mathcal{O}_{F_v^+}), V' \otimes F(a-1, b, c+1))_{\mathfrak{m}_{\overline{\tau}}} \neq 0$ and hence also assumptions (ii)–(iii) above. Just as in the proof of Theorem 4.5.2 it also implies that $S(U^v, \widetilde{V}')_{\mathfrak{m}_{\overline{\tau}}}^{I,(-b,-c,-a)} \neq 0$.

Let \mathbb{T}_a (resp. \mathbb{T}) denote the \mathcal{O}_E -subalgebra of

$$\operatorname{End}(S(U^{v},\widetilde{V}')^{I,(-b,-c,-a)}_{\mathfrak{m}_{\overline{r}}}) \cong \operatorname{End}((S(U^{v},\widetilde{V}')^{I,(-b,-c,-a)}_{\mathfrak{m}_{\overline{r}}})^{d})$$

generated by $\mathbb{T}^{\mathcal{P}}$ (resp. $\mathbb{T}^{\mathcal{P}}$, U_1 , and U_2). Here the subscript "a" stands for the "anemic" Hecke algebra, and the superscript "d" denotes the Schikhof dual (see Section 1.8 of [CEG⁺16]). Note that \mathbb{T}_a and \mathbb{T} are local \mathcal{O}_E -algebras (see the proof of Theorem 4.5.2).

Theorem 5.1.1. Let \bar{r} be as in Theorem 4.5.2(i) with $\operatorname{FL}(\bar{r}|_{G_{F_w}}) \neq \infty$. Assume (i)–(ix) in the setup above. Then

$$(S(U^v,\widetilde{V}')^{I,(-b,-c,-a)}_{\mathfrak{m}_{\overline{r}}})^{\mathrm{d}}$$

is free over \mathbb{T} .

Because of the following lemma, we may assume in the proof of Theorem 5.1.1 that E is as large as we like. (In fact, for the remainder of this section it will suffice to assume that E is large enough such that $\mathbb{T}_a[1/p] \cong E^r$ for some r and that $\mathbb{T}/\mathfrak{m}_{\mathbb{T}} = \mathbb{F}$.)

Lemma 5.1.2. Suppose that M is a finite free \mathcal{O}_E -module and that A is a local \mathcal{O}_E -algebra acting faithfully on M with $A/\mathfrak{m}_A = \mathbb{F}$. If E'/E is a finite extension and $M \otimes_{\mathcal{O}_E} \mathcal{O}_{E'}$ is free over $A \otimes_{\mathcal{O}_E} \mathcal{O}_{E'}$, then M is free over A.

Proof. As $A/\mathfrak{m}_A = \mathbb{F}$ we deduce that $A \otimes_{\mathfrak{O}_E} \mathfrak{O}_{E'}$ is local with residue field $\mathfrak{O}_{E'}/\mathfrak{m}_{E'}$. Pick any surjection $f: A^d \twoheadrightarrow M$ with d minimal. It is easy to see that $f \otimes_{\mathfrak{O}_E} \mathfrak{O}_{E'}$ is an isomorphism, hence so is f.

5.2. The Taylor–Wiles method. Let S be the set of places $S_p^+ \cup \{v_1\}$. For each place v' in S, fix a place \tilde{v}' of F lying over v' and let \tilde{S} be the set of these places \tilde{v}' . We will assume that $\tilde{v} = w$. For places w' in F, let $R_{w'}^{\Box}$ be the universal \mathcal{O}_E -lifting ring of $\bar{r}|_{G_{F_{w'}}}^{\vee}$. For $w' \in S_p \setminus \{w, w^c\}$ let $\psi_{w'} = a_{w'} \in \mathbb{Z}^3$ and let $R_{w'}^{\Box, \psi_{w'}}$ be the framed crystalline deformation ring for $\bar{r}|_{G_{F_{w'}}}^{\vee}$ of Hodge–Tate weights $\psi_{w'} + (2, 1, 0)$. Let δ_{F/F^+} denote the quadratic character of F/F^+ . Consider the deformation problem

$$\mathfrak{S} \stackrel{\text{def}}{=} (F/F^+, S, \widetilde{S}, \mathfrak{O}_E, \overline{r}^{\vee}, \varepsilon^{-2} \delta_{F/F^+}, \{R^{\square}_{\widetilde{v}_1}\} \cup \{R^{\square, \psi_{\widetilde{v}'}}_{\widetilde{v}'}\}_{v'|p, v' \neq v} \cup \{R^{\square}_{\widetilde{v}}\})$$

in the terminology of [CHT08, §2.3]. There is a universal deformation ring $R_{\mathbb{S}}^{\text{univ}}$ and a universal S-framed deformation ring $R_{\mathbb{S}}^{\Box_S}$ in the sense of [CHT08, §2.2]. (We work with deformations of \overline{r}^{\vee} to be consistent with [CEG⁺16]. Note that $\overline{r}^{\vee}(\text{Frob}_w)$ is used in our definition of $\mathfrak{m}_{\overline{r}}$ in §4.2.)

Let

$$R^{\mathrm{loc}} \stackrel{\mathrm{def}}{=} \widehat{\otimes}_{v'|p,v' \neq v} R^{\Box,\psi_{\widetilde{v}'}}_{\widetilde{v}'} \widehat{\otimes} R^{\Box}_{\widetilde{v}} \widehat{\otimes} R^{\Box}_{\widetilde{v}_1}$$

where all completed tensor products are taken over \mathcal{O}_E . Choose an integer $q \geq 3[F^+:\mathbb{Q}]$ as in Section 2.6 of [CEG⁺16]. We introduce the local ring

$$R_{\infty} \stackrel{\text{def}}{=} R^{\text{loc}}[[x_1, \dots, x_{q-3[F^+:\mathbb{Q}]}]],$$

over which [CEG⁺16] constructs a patched module of automorphic forms. Let $\tau \stackrel{\text{def}}{=} \operatorname{Ind}_{I}^{K}(\widetilde{\omega}^{-b} \otimes \widetilde{\omega}^{-c} \otimes \widetilde{\omega}^{-a}) \otimes_{\mathbb{Z}_{p}} \mathcal{O}_{E}$ be the natural \mathcal{O}_{E} -lattice in a principal series type over E. In the following we will identify $G(F_{v}^{+})$ with $\operatorname{GL}_{3}(\mathbb{Q}_{p})$ and $\mathcal{G}(\mathcal{O}_{F_{v}^{+}})$ with $K = \operatorname{GL}_{3}(\mathbb{Z}_{p})$ via ι_{w} . We collect some results from [CEG⁺16].

54

Theorem 5.2.1. There exist a map

$$S_{\infty} \stackrel{\text{def}}{=} \mathcal{O}_E[[z_1, \dots, z_{9\#S}, y_1, \dots, y_q]] \to R_{\infty}$$

and an $R_{\infty}[[K]]$ -module M_{∞} together with a compatible $\operatorname{GL}_3(\mathbb{Q}_p)$ -action, satisfying the following properties. For a finitely generated \mathfrak{O}_E -module W with continuous K-action, let $M_{\infty}(W)$ denote $\operatorname{Hom}_K^{\operatorname{cont}}(W, M_{\infty}^{\vee})^{\vee}$, where \cdot^{\vee} denotes the Pontryagin dual.

- (i) M_∞ is a finitely generated projective S_∞[[K]]-module. In particular, if W is p-torsion free, then M_∞(W) is a finite free S_∞-module.
- (ii) Suppose that W[1/p] is a locally algebraic type (as defined in [CEG⁺16], §4). Assume that M_∞(W) is non-zero and let Spec R_∞(W) be its support in Spec R_∞. Then M_∞(W) is a maximal Cohen–Macaulay R_∞(W)-module and M_∞(W)[1/p] is a projective R_∞(W)[1/p]-module.
- (iii) Let $\mathfrak{a} \stackrel{\text{def}}{=} (z_1, \ldots, z_{9\#S}, y_1, \ldots, y_q)$ be the augmentation ideal of S_{∞} . There is a natural $\operatorname{GL}_3(\mathbb{Q}_p)$ -equivariant identification

$$M_{\infty}/\mathfrak{a})^{\mathrm{d}} \cong \widetilde{S}(U^{v}, \widetilde{V}')_{\mathfrak{m}_{\overline{r}}},$$

which induces a $GL_3(\mathbb{Q}_p)$ -equivariant identification

$$(M_{\infty}(\tau)/\mathfrak{a})^{\mathrm{d}} \cong S(U^{v}, \widetilde{V}')^{I, (-b, -c, -a)}_{\mathfrak{m}_{\overline{\tau}}}.$$

Furthermore, there is a surjection $R_{\infty} \twoheadrightarrow \mathbb{T}_a$ so that the latter identification is R_{∞} -equivariant.

(iv) Let \wp_r denote the prime ideal of R_∞ corresponding to an \mathcal{O}_E -point of Spec \mathbb{T}_a and by abuse, the corresponding prime ideal of $\mathbb{T}^{\mathfrak{P}}$. By abuse, let $\mathfrak{m}_{\overline{r}}$ also denote the maximal ideal of R_∞ . Then

$$(M_{\infty}(\tau)/\wp_r)^{\mathrm{d}} \cong S(U^v, \widetilde{V}')[\wp_r]^{I, (-b, -c, -a)}$$

and

$$(M_{\infty}(\tau)/\mathfrak{m}_{\overline{r}})^{\vee} \cong S(U^v, V')[\mathfrak{m}_{\overline{r}}]^{I,(-b,-c,-a)}$$

Note that if W is p-torsion free, we have $M_{\infty}(W) \cong \operatorname{Hom}_{K}^{\operatorname{cont}}(W, M_{\infty}^{d})^{d}$ by Remark 4.15 of [CEG⁺16].

Proof. The construction of M_{∞} is in Section 2.8 of [CEG⁺16], except that we allow the Hodge–Tate weights $\psi_{\tilde{v}'}$ to depend on v', and we do not include Hecke operators at v_1 in our Hecke algebras. Without these Hecke operators, the rank of $M_{\infty}(W)[1/p]$ as a projective $R_{\infty}(W)[1/p]$ -module in part (ii) is greater than one. Otherwise, the necessary modifications are minor and straightforward. The map

$$S_{\infty} = \mathcal{O}_E[[z_1, \dots, z_{9\#S}, y_1, \dots, y_q]] \to R_{\infty}$$

and the module M_{∞} are defined in Section 2.8 of [CEG⁺16]. Part (i) follows from Proposition 2.10 and the proof of Lemma 4.18(1) of [CEG⁺16]. Part (ii) follows from Lemma 4.18(1) of [CEG⁺16].

The first identification in (iii) follows from Corollary 2.11 of [CEG⁺16]. For the second identification, we have

$$(M_{\infty}(\tau)/\mathfrak{a})^{\mathrm{d}} \cong \mathrm{Hom}_{K}^{\mathrm{cont}}(\tau, M_{\infty}^{\mathrm{d}})[\mathfrak{a}] \cong \mathrm{Hom}_{K}^{\mathrm{cont}}(\tau, (M_{\infty}/\mathfrak{a})^{\mathrm{d}})$$
$$\cong ((M_{\infty}/\mathfrak{a})^{\mathrm{d}})^{I,(-b,-c,-a)},$$

where the first isomorphism follows from Remark 4.15 of [CEG⁺16] and the final isomorphism follows from Frobenius reciprocity.

We now define the map $R_{\infty} \twoheadrightarrow \mathbb{T}_a$. The surjection $R_{\infty} \twoheadrightarrow \mathbb{T}_a$ comes from the construction in Section 2.8 of $[CEG^+16]$, as we now explain. We will freely use the notation of Section 2.8 of $[CEG^+16]$ with the caveat that we exclude the Hecke operators at v_1 from our Hecke algebras (which allows us to deduce that the maps from R_{∞} to our Hecke algebras are surjections). The patching argument produces compatible tuples $(\phi, M^{\Box}, \psi, \alpha)$ of all levels $N \ge 1$. Since the intersection $\bigcap_N \mathfrak{d}_N$ is 0, by completeness, the maps $\phi : R_{\infty} \twoheadrightarrow R_{\mathbb{S}}^{\mathrm{univ}}/\mathfrak{d}_N$ induce a surjection $R_{\infty} \twoheadrightarrow R_{S}^{\text{univ}}$. Furthermore, in the paragraph before Corollary 2.11 of [CEG⁺16], a map $R^{\text{univ}}_{\mathbb{S}} \to \mathbb{T}^{S_p}_{\xi,\tau}(U^{\mathfrak{p}}, \mathbb{O})_{\mathfrak{m}}$ is described as the inverse limit of maps $R^{\text{univ}}_{\mathbb{S}} \to$ $\mathbb{T}^{S_p}_{\xi,\tau}(U_m, \mathcal{O})_{\mathfrak{m}}$ (or equivalently of maps $R^{\text{univ}}_{\mathfrak{S}} \to \mathbb{T}^{S_p}_{\xi,\tau}(U_{2N}, \mathcal{O}/\varpi^N)_{\mathfrak{m}}$) at finite level. Using the explicit generators given by the surjection $\mathbb{T}^{S_p,\text{univ}} \to \mathbb{T}^{S_p}_{\mathcal{E},\tau}(U^{\mathfrak{p}}, \mathbb{O})_{\mathfrak{m}}$ (again for us with operators at v_1 excluded) and Proposition 3.4.4(2) of [CHT08], each of these maps at finite level is surjective. By completeness, the map $R_{\rm S}^{\rm univ} \rightarrow$ $\mathbb{T}^{S_p}_{\mathcal{E},\tau}(U^{\mathfrak{p}},\mathbb{O})_{\mathfrak{m}}$ is surjective. We conclude that the composition $R_{\infty} \to R^{\mathrm{univ}}_{\mathcal{S}} \to$ $\mathbb{T}_{\mathcal{E},\tau}^{S_p}(U^{\mathfrak{p}},\mathbb{O})_{\mathfrak{m}}$ is surjective. Composing with the natural map $\mathbb{T}_{\mathcal{E},\tau}^{S_p}(U^{\mathfrak{p}},\mathbb{O})_{\mathfrak{m}} \twoheadrightarrow \mathbb{T}_a$ gives a surjective homomorphism $R_{\infty} \twoheadrightarrow \mathbb{T}_a$.

The proof of Corollary 2.11 of $[\text{CEG}^+16]$ shows that the first identification of (iii) is R_{∞} -equivariant, with R_{∞} acting via the map $R_{\mathbb{S}}^{\text{univ}} \to \mathbb{T}_{\xi,\tau}^{S_p}(U^{\mathfrak{p}}, \mathbb{O})_{\mathfrak{m}}$ on the righthand side, since at each finite level N, the map $\psi: M^{\Box}/\mathfrak{a} \xrightarrow{\sim} \tilde{S}_{\xi,\tau}(U_{2N}, \mathbb{O}/\varpi^N)_{\mathfrak{m}}$ is R_{∞} -equivariant, with R_{∞} acting via the map $R_{\mathbb{S}}^{\text{univ}} \to \mathbb{T}_{\xi,\tau}^{S_p}(U_{2N}, \mathbb{O}/\varpi^N)_{\mathfrak{m}}$ on the right-hand side. This shows that the second identification is also R_{∞} -equivariant, with R_{∞} acting on the right-hand side via the map $R_{\infty} \twoheadrightarrow \mathbb{T}_a$ constructed above. Part (iv) follows from part (iii), noting that $\mathfrak{a} \subseteq \ker(R_{\infty} \to \mathbb{T}_a)$ by part (iii). \Box

Let R be the R_{∞} -subalgebra of $\operatorname{End}_{R_{\infty}}(M_{\infty}(\tau))$ generated by U_1 and U_2 .

Lemma 5.2.2. The ring R is local with maximal ideal $(\mathfrak{m}_{\overline{r}}, U_1, U_2)$.

Here, by abuse, $\mathfrak{m}_{\overline{r}}$ denotes the image of the maximal ideal of R_{∞} in R.

Proof. As $M_{\infty}(\tau)$ is a finitely-generated $R_{\infty}(\tau)$ -module, we deduce that the ring extension $R_{\infty}(\tau) \subseteq R$ is finite, hence $\mathfrak{m}_{\overline{r}} \subseteq \operatorname{rad}(R)$. In particular, $\mathfrak{a} \subseteq \operatorname{rad}(R)$, hence also $\sqrt{\mathfrak{a}R} \subseteq \operatorname{rad}(R)$. By Theorem 5.2.1(iii), the action of R_{∞}/\mathfrak{a} on $M_{\infty}(\tau)/\mathfrak{a}$ factors through \mathbb{T}_a , hence the action of R/\mathfrak{a} on $M_{\infty}(\tau)/\mathfrak{a}$ factors through \mathbb{T} , giving rise to a surjective homomorphism $R/\mathfrak{a} \twoheadrightarrow \mathbb{T}$. As the action of R on $M_{\infty}(\tau)$ is faithful, the R/\mathfrak{a} -module $M_{\infty}(\tau)/\mathfrak{a}$ has full support by [Tay08, Lemma 2.2] (i.e., is nearly faithful in the terminology of [Tay08]). Since \mathbb{T} is reduced, we get an induced isomorphism $(R/\mathfrak{a})_{\mathrm{red}} \xrightarrow{\sim} \mathbb{T}$. It follows that R is local, and $U_i \in \mathfrak{m}_R$, as $U_i \in \mathfrak{m}_{\mathbb{T}}$ by the proof of Theorem 4.5.2 (i = 1, 2). It is then easy to see that $\mathfrak{m}_R = (\mathfrak{m}_{\overline{r}}, U_1, U_2)$.

We now show that Theorem 5.1.1 follows from the following theorem.

Theorem 5.2.3. Let \bar{r} be as in Theorem 4.5.2(i) with $\operatorname{FL}(\bar{r}|_{G_{F_w}}) \neq \infty$. The module $M_{\infty}(\tau)$ is finite free over R.

Proof of Theorem 5.1.1. As $M_{\infty}(\tau)$ is finite free over R by Theorem 5.2.3, $M_{\infty}(\tau)/\mathfrak{a}$ is finite free over R/\mathfrak{a} . Hence the natural surjection $R/\mathfrak{a} \twoheadrightarrow \mathbb{T}$ considered in the proof of Lemma 5.2.2 is an isomorphism. We conclude by Theorem 5.2.1(iii).

5.3. The Diamond–Fujiwara trick. In this section, we prove Theorem 5.2.3 using the method of Diamond and Fujiwara. Let $B \subseteq \operatorname{GL}_{3/\mathbb{Z}_p}$ be the algebraic subgroup of upper-triangular matrices. For a dominant character λ of B, let $W_{\mathfrak{O}_E}(\lambda)$ be the algebraic induced module $(\operatorname{Ind}_B^{\operatorname{GL}_3} w_0 \lambda)^{\operatorname{alg}} \otimes_{\mathbb{Z}_p} \mathfrak{O}_E$, considered as representation of $K = \operatorname{GL}_3(\mathbb{Z}_p)$. (This is the characteristic 0 version of the module defined in Section 3.) Let $W \stackrel{\text{def}}{=} W_{\mathfrak{O}_E}(-c-1, -b, -a+1)$.

Proposition 5.3.1. The module $M_{\infty}(W)$ is non-zero and finite free over its support in R_{∞} .

Proof. Note that $F(-c-1, -b, -a+1) \cong W \otimes_{\mathcal{O}_E} \mathbb{F}$ by Proposition 3.18(ii) of [Her09]. Hence $M_{\infty}(W) \neq 0$ by Theorem 4.4.1, Theorem 5.2.1(iii), as well as Proposition 5.1.1 in [Le]. Let $\psi_{\widetilde{v}} \stackrel{\text{def}}{=} (a-1, b, c+1)$ and

$$R_{\infty}(W)' \stackrel{\text{def}}{=} \widehat{\otimes}_{v'|p} R_{\widetilde{v}'}^{\Box,\psi_{\widetilde{v}'}} \widehat{\otimes} R_{\widetilde{v}_1}^{\Box}[[x_1,\ldots,x_{q-3[F^+:\mathbb{Q}]}]],$$

where $R_{\widetilde{v}}^{\Box,\psi_{\widetilde{v}}}$ is the framed crystalline deformation ring for $\overline{r}|_{G_{F_{\widetilde{v}}}}^{\lor}$ of Hodge–Tate weights $\psi_{\widetilde{v}} + (2,1,0)$. Then $M_{\infty}(W)$ is supported on Spec $R_{\infty}(W)'$ by Lemma 4.17(1) of [CEG⁺16]. (Note the dual in the definition of "Hodge–Tate weights prescribed by σ_{alg} " in Section 4 of [CEG⁺16].) By the proof of Lemma 4.18 of [CEG⁺16], $M_{\infty}(W)$ is a maximal Cohen–Macaulay $R_{\infty}(W)'$ -module. By the choice of $v_1, R_{\widetilde{v}_1}^{\Box}$ is formally smooth over \mathcal{O}_E , see Lemma 2.5 of [CEG⁺16]. By Lemma 2.4.1 of [CHT08], $R_{\widetilde{v}'}^{\Box,\psi_{\widetilde{v}'}}$ is formally smooth over \mathcal{O}_E for all v'|p. As $R_{\infty}(W)'$ is a regular local ring, $M_{\infty}(W)$ is a projective and hence free $R_{\infty}(W)'$ -module by the Auslander–Buchsbaum–Serre theorem and the Auslander–Buchsbaum formula. \Box

Let n be the (free) rank of $M_{\infty}(W)$ over its support.

Lemma 5.3.2. Let \overline{r} be as in Theorem 4.5.2(i) with $\operatorname{FL}(\overline{r}|_{G_{F_w}}) \neq \infty$. The *R*-module $M_{\infty}(\tau)$ is generated by *n* elements.

Proof. Let $\overline{\tau} \stackrel{\text{def}}{=} \tau \otimes_{\mathcal{O}_E} \mathbb{F}$. By Nakayama's lemma, it suffices to show that $M_{\infty}(\overline{\tau})/(\mathfrak{m}_{\overline{\tau}}, U_1, U_2)$ has dimension n. Consider

$$(M_{\infty}(\overline{\tau})/(\mathfrak{m}_{\overline{\tau}}, U_1, U_2))^{\vee} \cong \operatorname{Hom}_K(\overline{\tau}, M_{\infty}^{\vee})[\mathfrak{m}_{\overline{\tau}}, U_1, U_2]$$
$$\cong \operatorname{Hom}_K(\overline{\tau}, M_{\infty}^{\vee}[\mathfrak{m}_{\overline{\tau}}])[U_1, U_2]$$
$$\cong \operatorname{Hom}_K(\overline{\tau}/M, M_{\infty}^{\vee})[\mathfrak{m}_{\overline{\tau}}]$$
$$\cong (M_{\infty}(\overline{\tau}/M)/\mathfrak{m}_{\overline{\tau}})^{\vee},$$

where the third isomorphism holds by Lemma 3.1.11, and where M is the minimal subrepresentation of $\overline{\tau}$ containing F(-c, -a, -b-p+1) and F(-a+p-1, -b, -c-p+1) as Jordan–Hölder factors. By the isomorphism $M_{\infty}(\overline{\tau})/(\mathfrak{m}_{\overline{\tau}}, U_1, U_2) \cong M_{\infty}(\overline{\tau}/M)/\mathfrak{m}_{\overline{\tau}}$, it suffices to show that $M_{\infty}(\overline{\tau}/M)/\mathfrak{m}_{\overline{\tau}}$ has dimension n. By Proposition 2.2.2 of [Le] and Theorem 4.4.1, M is of length 3 and $(\overline{\tau}/M)^{\vee}$ contains no element of $W_w(\overline{\tau})$ other than F(a-1,b,c+1). Hence by Proposition 5.1.1 of [Le], the natural inclusion $M_{\infty}(F(-c-1,-b,-a+1)) \hookrightarrow M_{\infty}(\overline{\tau}/M)$ is an isomorphism. We conclude that the maps

$$M_{\infty}(W)/\mathfrak{m}_{\overline{r}} \to M_{\infty}(F(-c-1,-b,-a+1))/\mathfrak{m}_{\overline{r}} \to M_{\infty}(\overline{\tau}/M)/\mathfrak{m}_{\overline{r}}$$

are isomorphisms of n-dimensional \mathbb{F} -vector spaces, by definition of n.

Proof of Theorem 5.2.3. By Lemma 5.3.2, we can and do fix a surjection $\mathbb{R}^n \twoheadrightarrow M_{\infty}(\tau)$ with kernel P. This induces an exact sequence

(5.3.3)
$$0 \to P[1/p] \to (R[1/p])^n \to M_{\infty}(\tau)[1/p] \to 0.$$

We now freely use the notation of Section 4 of [CEG⁺16]. First, we note that $R_{\infty}(\tau)[1/p] \rightarrow R[1/p]$ is an isomorphism by Theorem 4.1 and Lemma 4.17(2) of [CEG⁺16]. Note that while the construction of M_{∞} is slightly different in this paper, the proof of Lemma 4.17(2) still applies verbatim.

By Proposition 2.4.7 of [EGH13] (and its easy converse) $\tau \otimes_{\mathbb{O}_E} \overline{\mathbb{Q}}_p$ is an irreducible smooth representation of K that is associated to the inertial type $\tilde{\omega}^{-b} \oplus \tilde{\omega}^{-c} \oplus \tilde{\omega}^{-a}$ by the inertial local Langlands correspondence ([CEG⁺16], Theorem 3.7). Hence by Theorem 5.2.1(ii), $M_{\infty}(\tau)[1/p]$ is a projective $R_{\infty}(\tau)[1/p]$ -module. We claim that it is of constant rank n. The rank is clearly no larger than n by (5.3.3). Thus it suffices to show that each irreducible component of $\operatorname{Spec}(R_{\infty}(\tau)[1/p])$, or equivalently of $\operatorname{Spec} R_{\infty}(\tau)$, contains an E-point over which the fiber of $M_{\infty}(\tau)[1/p]$ has E-dimension at least n.

First, we claim that each irreducible component Z of $\operatorname{Spec} R_{\infty}(\tau)$ has an Epoint in the closed subscheme $\operatorname{Spec} \mathbb{T}_a$. Indeed as $M_{\infty}(\tau)$ is finite free over S_{∞} by Theorem 5.2.1(i), $R_{\infty}(\tau)$ is a subring of a matrix ring over S_{∞} and hence a finite torsion-free S_{∞} -module. From this, we see that $R_{\infty}(\tau)$ is Cohen–Macaulay and in particular equidimensional. As $Z \to \operatorname{Spec} S_{\infty}$ is a finite map between irreducible schemes of the same dimension, it is surjective. In particular, the base change $Z \times_{\operatorname{Spec} S_{\infty}} \operatorname{Spec}(S_{\infty}/\mathfrak{a}) \subseteq \operatorname{Spec} R_{\infty}(\tau)/\mathfrak{a}$ has a non-empty generic fibre. Just as in the proof of Lemma 5.2.2, we have an isomorphism $(R_{\infty}(\tau)/\mathfrak{a})_{\operatorname{red}} \xrightarrow{\sim} \mathbb{T}_a$. Since E is sufficiently large, $\operatorname{Spec}(\mathbb{T}_a[1/p])$ is a union of copies of $\operatorname{Spec} E$. We conclude that the underlying reduced subscheme of the generic fiber of $Z \times_{\operatorname{Spec} S_{\infty}} \operatorname{Spec}(S_{\infty}/\mathfrak{a})$ is a non-empty union of copies of $\operatorname{Spec} E$ in $\operatorname{Spec} \mathbb{T}_a$.

Second, we claim that for each \mathcal{O}_E -point of Spec \mathbb{T}_a with corresponding prime ideal \wp_r , $M_{\infty}(\tau)/\wp_r$ has generic rank at least n. Let M be the minimal subrepresentation of $\overline{\tau}$ containing F(-c-1, -b, -a+1) as a Jordan–Hölder factor. By Proposition 2.2.2 of [Le] and Theorem 4.4.1, $(\overline{\tau}/M)^{\vee}$ contains no element of $W_w(\overline{r})$. By Proposition 5.1.1 of [Le], the natural map $M_{\infty}(M) \to M_{\infty}(\overline{\tau})$ is an isomorphism. The natural surjection $M_{\infty}(M) \twoheadrightarrow M_{\infty}(F(-c-1, -b, -a+1))$ gives a surjection $M_{\infty}(\overline{\tau}) \twoheadrightarrow M_{\infty}(F(-c-1, -b, -a+1))$, and therefore a surjection $M_{\infty}(\overline{\tau})/\mathfrak{m}_{\overline{\tau}} \twoheadrightarrow M_{\infty}(F(-c-1, -b, -a+1))/\mathfrak{m}_{\overline{\tau}}$. Since $M_{\infty}(F(-c-1, -b, -a+1))/\mathfrak{m}_{\overline{\tau}}$ has dimension n by definition of n, $M_{\infty}(\overline{\tau})/\mathfrak{m}_{\overline{\tau}}$ has dimension at least n. For any \mathcal{O}_E -point of Spec \mathbb{T}_a with corresponding prime ideal \wp_r , $M_{\infty}(\tau)/\wp_r$ is p-torsion free by Theorem 5.2.1(iv) and so has generic rank at least n.

As $M_{\infty}(\tau)[1/p]$ is projective of rank n over $R[1/p] \cong R_{\infty}(\tau)[1/p]$, we conclude by (5.3.3) that P[1/p] is projective of rank 0 and hence 0. As R is p-torsion free, P is 0.

Let R_2 be the R_∞ -subalgebra of $\operatorname{End}_{R_\infty}(M_\infty(\tau))$ generated by U_2 , and let \mathbb{T}_2 be the \mathcal{O}_E -subalgebra of $\operatorname{End}(S(U^v, \widetilde{V}')_{\mathfrak{m}_{\overline{\tau}}}^{I,(-b,-c,-a)})$ generated by $\mathbb{T}^{\mathcal{P}}$ and U_2 .

Theorem 5.3.4. Let \overline{r} be as in Theorem 4.5.2(*i*)-(*ii*). We have $R_2 = R$ and $\mathbb{T}_2 = \mathbb{T}$, *i.e.* $U_1 \in R_2$ and $U_1 \in \mathbb{T}_2$. Moreover, $S(U^v, V')[\mathfrak{m}_{\overline{r}}]^{I,(-b,-c,-a)}[U_2] = S(U^v, V')[\mathfrak{m}_{\overline{r}}]^{I,(-b,-c,-a)}[U_1, U_2].$

Proof. The argument for Theorem 5.2.3 goes through to show that the module $M_{\infty}(\tau)$ is also finite free over R_2 , and the free rank is the same as for R. (The main thing that changes in the proof is that the analogue of M in the proof of Lemma 5.3.2 now has length 2, and we need that $\operatorname{FL}(\overline{r}|_{G_{F_w}}) \notin \{0,\infty\}$ in order to deduce that $(\overline{\tau}/M)^{\vee}$ contains no element of $W_w(\overline{r})$.) Now $R_2 \subseteq R$ is a finite ring extension, hence $\mathfrak{m}_R \cap R_2 = \mathfrak{m}_{R_2}$. As E is sufficiently large, the residue fields of R_2 and R are both equal to \mathbb{F} . It follows that the surjection $M_{\infty}(\tau)/\mathfrak{m}_{R_2} \twoheadrightarrow M_{\infty}(\tau)/\mathfrak{m}_R$ is an isomorphism. Therefore any R_2 -basis of $M_{\infty}(\tau)$ is also an R-basis, so $R_2 = R$. The other claims follow.

Remark 5.3.5. If $W_w(\bar{r}) = \{F(a-1, b, c+1)\}$, then we even deduce (by the same argument) that $R_\infty(\tau) = R$, $\mathbb{T}_a = \mathbb{T}$, and that U_1, U_2 annihilate $S(U^v, V')[\mathfrak{m}_{\bar{r}}]^{I,(-b,-c,-a)}$.

Remark 5.3.6. The analogues of all the above results in this section hold for the "dual" lattice $\tau' \stackrel{\text{def}}{=} \operatorname{Ind}_{I}^{K} (\widetilde{\omega}^{-c} \otimes \widetilde{\omega}^{-a} \otimes \widetilde{\omega}^{-b}) \otimes_{\mathbb{Z}_{p}} \mathcal{O}_{E}$, provided one interchanges U_{1} with U_{2} and the condition $\operatorname{FL}(\overline{r}|_{G_{F_{w}}}) \neq \infty$ with $\operatorname{FL}(\overline{r}|_{G_{F_{w}}}) \neq 0$.

We now show that Theorem 4.5.2 applies to Galois representations of any possible invariant outside 0, ∞ .

Theorem 5.3.7. Suppose that $\overline{\rho}: G_{\mathbb{Q}_p} \to \mathrm{GL}_3(\mathbb{F})$ is upper-triangular, maximally non-split, and generic. Then, after possibly replacing \mathbb{F} by a finite extension, there exist a CM field F, a Galois representation $\overline{r}: G_F \to \mathrm{GL}_3(\mathbb{F})$, a place w|p of F, groups $G_{/F^+}$ and $\mathcal{G}_{/\mathcal{O}_{F^+}}$, and a compact open subgroup U^v (where $v = w|_{F^+}$) satisfying all hypotheses of the setup in Section 5.1 such that $\overline{r}|_{G_{Fw}} \cong \overline{\rho}$.

In particular, if $FL(\overline{\rho}) \notin \{0, \infty\}$, Theorem 4.5.2 applies to \overline{r} .

Proof. We suppose that $\overline{\rho}$ is as in (2.1.1) with $(a_2, a_1, a_0) = (a, b, c)$. We note that $\overline{\rho}$ satisfies the two hypotheses of [CEG⁺16], §2.1: $p \nmid 6$ by genericity, and $\overline{\rho}$ has a potentially diagonalizable lift of regular weight: for example, by Corollary 4.4.4 or by [GG12], Lemma 3.1.5 there exists a crystalline lift with distinct Hodge–Tate weights in the Fontaine–Laffaille range. Therefore the procedure of [CEG⁺16], §2.1, building on [EG14], §A, yields a "suitable globalization" $\overline{r} : G_F \to \text{GL}_3(\mathbb{F})$ (after possibly replacing \mathbb{F}) for a CM field F with F/F^+ unramified at all finite places. For each place $v' \in S_p^+$ there exists a place \tilde{v}' of F lying over v' with $F_{\tilde{v}'} = \mathbb{Q}_p$ and $\overline{r}|_{G_{F_{\tilde{v}'}}} \cong \overline{\rho}$, which is Fontaine–Laffaille and regular. We see that hypotheses (ii)–(v) in §5.1 are satisfied.

Choose a unitary group G with integral model \mathcal{G} as in §5.1. Fix any $v \in S_p^+$, let $w \stackrel{\text{def}}{=} \tilde{v}$, and define $\tilde{V}' = \bigotimes_{v' \in S_p^+ \setminus \{v\}} W_{a_{v'}}$ with $a_{\tilde{v}'} = (a - 1, b, c + 1)$ for any $v' \in S_p^+ \setminus \{v\}$. There exists a compact open subgroup $U = \prod_{v'} U_{v'} \leq G(\mathbb{A}_{F^+}^\infty)$ with $U_{v'} = \mathcal{G}(\mathcal{O}_{F_{v'}^+})$ (resp. $U_{v'}$ hyperspecial) for all places v' which split (resp. are inert) in F such that $S(U^v, \tilde{V}')_{\mathfrak{m}_{\overline{r}}} \neq 0$, where \mathcal{P} and $\mathfrak{m}_{\overline{r}}$ are defined as in §5.1. (Note that a priori the proof of [EG14], Corollary A.7 does not give us any information about the type (λ, τ) in [EG14], Definition 5.3.1. However, we can get the desired type by choosing appropriate lifts locally at places dividing p in applying [BLGGT14], Theorem 4.3.1 in the proof of [EG14], Lemma A.5.) By choosing a place v_1 and redefining U_{v_1} as in §5.1, we see that hypothesis (ix) in §5.1 is satisfied. \Box

A. Some integral *p*-adic Hodge theory

The first aim of this appendix is to collect some results in integral *p*-adic Hodge theory. Several of these are slight generalizations of known results in the literature, incorporating coefficients and descent data. The second aim of the appendix is to prove some lemmas and propositions of Sections 2.2 and 2.3.

A.1. On certain categories of torsion modules. The goal of this section is to recall the definition of various categories of *p*-torsion modules with additional structures (Breuil modules, Kisin modules, ϕ -modules) and the relations between them.

Recall that $K_0 = W(k)[\frac{1}{p}]$ and that $K = K_0(\varpi)$, where $\varpi = \sqrt[e]{-p}$ and $e \ge 1$ divides $p^{[k:\mathbb{F}_p]} - 1$. Also recall our definitions of $\overline{S}_k = k[u]/u^{ep}$ and $S_{W(k)}$ in Section 2.2. We will always suppose that $0 \le r \le p-2$. For the purpose of establishing Proposition 2.2.1 we will consider general e, but sometimes we will specialize to e = 1. We hope that this will not cause any confusion.

If \mathcal{A} is any additive category we let \mathbb{F} - \mathcal{A} denote the \mathbb{F} -linear additive category whose objects consist of pairs (A, i), where $A \in \mathcal{A}$ and $\mathbb{F} \to \text{End}(A)$ is a ring homomorphism (the morphisms are the obvious ones). We will often use below, without further comment, that a functor $\mathcal{A} \to \mathcal{B}$ of additive categories gives rise to a functor \mathbb{F} - $\mathcal{A} \to \mathbb{F}$ - \mathcal{B} .

We consider the general setting of §2.2.1. In particular, we write BrMod^r to denote the category of Breuil modules over \overline{S}_k and BrMod^r_0 for the category of Breuil modules without monodromy (also called *quasi-Breuil modules*), defined in the evident way. Note that these categories are respectively denoted by $\underline{\widetilde{M}}^r, \underline{\widetilde{M}}^r_0$ in [Car06], [Bre99b].

In [Bre99a], §2.2.1 (cf. also [Car06], §2.1) the authors introduce certain categories of modules, noted as $\underline{\mathcal{M}}^r$, $\underline{\mathcal{M}}^r_0$. We actually only consider their full subcategories formed by *p*-torsion objects, though we keep the same notation. The objects of these subcategories are finite free modules over $S_{W(k)}/pS_{W(k)}$. We have the following commutative diagram:

$$\underbrace{\underline{\mathcal{M}}^{r} \xrightarrow{\mathcal{T}} \operatorname{BrMod}^{r}}_{\underbrace{\bigcup_{r=1}^{r} \underbrace{\longrightarrow}_{T_{0}}} \operatorname{BrMod}_{0}^{r}$$

where the functors \mathcal{T} , \mathcal{T}_0 inducing equivalences of categories are defined in [Bre98], §2.2.2 when e = 1 and [Car06], §2.3 in general, and the vertical arrows are forgetful functors. (Strictly speaking, this diagram and the ones that follow only commute up to equivalence.)

Choose a sequence $(p_n)_n \in \overline{\mathbb{Q}}_p^{\mathbb{N}}$ verifying $p_{n+1}^p = (-1)^{p-1}p_n$ for all n and $p_0 \stackrel{\text{def}}{=} -p$, so $N_{K_0(p_{n+1})/K_0(p_n)}(p_{n+1}) = p_n$. Let $k((\underline{p}))$ be the field of norms associated to the extension $(K_0)_{\infty}/K_0$, where $(K_0)_{\infty} \stackrel{\text{def}}{=} \bigcup_{n \in \mathbb{N}} K_0(p_n)$, and let $k[[\underline{p}]]$ be its ring of integers. In particular, \underline{p} is identified with the sequence $(p_n)_n$. We write $\mathfrak{Mod}_{k[[\underline{p}]]}^r$ for the category of $(\phi, k[[p]])$ -modules of height $\leq r$ defined in [Bre99b], §2.3.

We have an equivalence of categories:

$$M_{\mathfrak{S}}: \mathfrak{Mod}_{k[[\underline{p}]]}^{r} \xrightarrow{\sim} \mathrm{BrMod}_{0}^{r}$$

(cf. [Bre99b], §4, where the functor is noted by Θ_r).

We finally define the category $\mathfrak{Mod}_{k((\underline{p}))}$ of étale $(\phi, k((\underline{p})))$ -modules as the category of finite-dimensional $k((\underline{p}))$ -vector spaces \mathfrak{D} endowed with a semilinear map $\phi: \mathfrak{D} \to \mathfrak{D}$ (with respect to the Frobenius on $k((\underline{p}))$) and inducing an isomorphism $\phi^*\mathfrak{D} \to \mathfrak{D}$ (with obvious morphisms between objects).

By work of Fontaine [Fon90], we have an exact anti-equivalence

$$\mathbb{F}\text{-}\mathfrak{Mod}_{k((\underline{p}))} \xrightarrow{\sim} \operatorname{Rep}_{\mathbb{F}}(G_{(K_0)_{\infty}})$$
$$\mathfrak{D} \longmapsto \operatorname{Hom}(\mathfrak{D}, k((\underline{p}))^{\operatorname{sep}}).$$

When e = 1 we have the fully faithful embedding $\mathcal{F}_r : \mathcal{FL}^{[0,r]} \longrightarrow \operatorname{BrMod}_0^r$ of [Bre99b], §5 (via the equivalence \mathcal{T}_0 above). By composition we obtain a fully faithful functor

$$\mathbb{F}\text{-}\mathcal{FL}^{[0,r]} \xrightarrow{\mathcal{F}_r} \mathbb{F}\text{-}\mathrm{BrMod}_0^r \xleftarrow{M_{\mathfrak{S}}}{\sim} \mathbb{F}\text{-}\mathfrak{Mod}_{k[[p]]}^r \longrightarrow \mathbb{F}\text{-}\mathfrak{Mod}_{k((\underline{p}))},$$

where the last functor in the sequence is localization at \underline{p} (which is fully faithful by [Bre99b], Proposition 2.3.7).

We are now going to define the analogous categories and functors with descent data. Recall that $\varpi = \sqrt[6]{-p} \in K$. There is a unique sequence $(\varpi_n)_n \in \overline{\mathbb{Q}}_p^{\mathbb{N}}$ such that $\varpi_n^e = p_n, \ \varpi_{n+1}^p = (-1)^{p-1} \varpi_n$ for all n, and $\varpi_0 \stackrel{\text{def}}{=} \varpi$. In particular, $N_{K(\varpi_{n+1})/K(\varpi_n)}(\varpi_{n+1}) = \varpi_n$. By letting $K_{\infty} \stackrel{\text{def}}{=} \bigcup_{n \in \mathbb{N}} K(\varpi_n)$ we have a canonical isomorphism $\operatorname{Gal}(K_{\infty}/(K_0)_{\infty}) \xrightarrow{\sim} \operatorname{Gal}(K/K_0)$ which lets us identify ω_{ϖ} with a character of $\operatorname{Gal}(K_{\infty}/(K_0)_{\infty})$.

As $\underline{\varpi}^e = \underline{p}$ and e divides $|k^{\times}|$, the field of norms $k((\underline{\varpi}))$ associated to K_{∞}/K is a cyclic extension of $k((\underline{p}))$ of degree e. Note that $\operatorname{Gal}(K_{\infty}/(K_0)_{\infty}) \cong \operatorname{Gal}(k((\underline{\varpi}))/k((\underline{p})))$ acts on $k((\underline{\varpi}))$; concretely this action is determined by $g \cdot \underline{\varpi} = \omega_{\overline{\varpi}}(g)\underline{\varpi}$ for $g \in \operatorname{Gal}(K_{\infty}/(K_0)_{\infty})$.

We write $\mathfrak{Mod}_{k[[\underline{\varpi}]]}^{r}$ for the category of $(\phi, k[[\underline{\varpi}]])$ -modules of height $\leq r$, described in [Car11], §2.1 (where it is denoted by $\mathrm{Mod}_{/\tilde{\mathfrak{S}}}^{\phi}$). By [Car11], Théorème 2.1.2 (building on Breuil [Bre99b], Théorème 4.1.1) we have an exact equivalence of categories

$$M_{\mathfrak{S}}: \mathfrak{Mod}_{k[[\varpi]]}^{r} \xrightarrow{\sim} \operatorname{BrMod}_{0}^{r}.$$

We define in the evident way the category $\mathfrak{Mod}_{k[[\underline{\varpi}]], \mathrm{dd}}^{r}$ consisting of $(\phi, k[[\underline{\varpi}]])$ modules endowed with a descent data from K to K_0 , i.e. a semilinear action of $\operatorname{Gal}(K_{\infty}/(K_0)_{\infty})$ that commutes with ϕ (the morphisms now being required to be compatible with the descent datum).

Lemma A.1.1. There exists an exact covariant functor

$$M_{\mathfrak{S}}: \mathbb{F}\operatorname{-\mathfrak{Mod}}^{r}_{k[[\varpi]], \operatorname{dd}} \to \mathbb{F}\operatorname{-BrMod}^{r}_{\operatorname{dd}, 0}$$

which establishes an equivalence of categories.

Here, the category \mathbb{F} -BrMod^r_{dd,0} has the same definition as \mathbb{F} -BrMod^r_{dd}, except that we drop monodromy.

Proof. We may assume that $\mathbb{F} = \mathbb{F}_p$. By above we only need to check compatibility with the descent data. For this we follow the strategy outlined in [EGH13], proof of Proposition 3.2.6. More precisely, given an object $\mathfrak{M} \in \mathfrak{Mod}_{k[[\underline{\varpi}]], \mathrm{dd}}^r$ and an element $g \in \mathrm{Gal}(K/K_0)$ we define its twist $\mathfrak{M}^{(g)}$ as $k[[\underline{\varpi}]] \otimes_{g,k[[\underline{\varpi}]]} \mathfrak{M}$ with ϕ acting diagonally. It is easily checked that $\mathfrak{M}^{(g)} \in \mathfrak{Mod}_{k[[\underline{\varpi}]]}^r$ and that $\widehat{g} : \mathfrak{M} \to \mathfrak{M}$ induces a morphism $\widehat{g} : \mathfrak{M}^{(g)} \to \mathfrak{M}$ in $\mathfrak{Mod}_{k[[\underline{\varpi}]]}^r$; moreover, $\widehat{gh} = \widehat{g} \circ \widehat{h}^{(g)}$ for any $g, h \in \mathrm{Gal}(K/K_0)$. It is formal to verify that the datum of a family of morphisms $\widehat{g} : \mathfrak{M}^{(g)} \to \mathfrak{M}$ verifying the cocycle relation is equivalent to a descent datum on \mathfrak{M} from K to K_0 .

Recall now that the functor $M_{\mathfrak{S}}:\mathfrak{Mod}_{k[[\underline{\varpi}]]}^{r} \to \operatorname{BrMod}_{0}^{r}$ is defined as the base change induced by the morphism $k[[\underline{\varpi}]] \twoheadrightarrow \overline{S}_{k} \xrightarrow{\varphi} \overline{S}_{k}$, where the first map sends $\sum \lambda_{i} \underline{\varpi}^{i}$ to $\sum \lambda_{i} u^{i}$. Therefore $M_{\mathfrak{S}}(\mathfrak{M}^{(g)}) \cong (M_{\mathfrak{S}}(\mathfrak{M}))^{(g)}$, where the twist in the right-hand side is the one defined in [EGH13], proof of Proposition 3.2.6.

By functoriality, the quasi-Breuil module $\mathcal{M} \stackrel{\text{def}}{=} M_{\mathfrak{S}}(\mathfrak{M})$ is thus endowed with a family of morphism $\mathcal{M}^{(g)} \to \mathcal{M}$ verifying the above cocycle relation, i.e. \mathcal{M} is endowed with descent data from K to K_0 . We conclude that the composite functor $\mathfrak{Mod}_{k[[\underline{\varpi}]], \mathrm{dd}}^r \to \mathfrak{Mod}_{k[[\underline{\varpi}]]}^r \xrightarrow{\sim} \mathrm{BrMod}_0^r$ factors though $\mathrm{BrMod}_{\mathrm{dd},0}^r \to \mathrm{BrMod}_0^r$. Its exactness and the induced equivalence follow as a formal consequence of the analogous properties of $M_{\mathfrak{S}}$, cf. [Car11], Théorème 2.1.2.

We can finally introduce the category $\mathfrak{Mod}_{k((\underline{\varpi})), \mathrm{dd}}$ of étale $(\phi, k((\underline{\varpi})))$ -modules with descent data: an object \mathfrak{D} is defined in the analogous way as for the category $\mathfrak{Mod}_{k((\underline{p}))}$, but we moreover require that \mathfrak{D} is endowed with a semilinear action of $\mathrm{Gal}(K/K_0)$ that commutes with ϕ .

We can now define a functor by composition:



A.2. Some basic lemmas on Breuil modules.

Lemma A.2.1. Let \mathbb{N} be an \overline{S} -submodule of a Breuil module $\mathbb{M} \in \mathbb{F}$ -BrMod^r. Then \mathbb{N} is an \overline{S}_k -direct summand of \mathbb{M} if and only if \mathbb{N} is free as \overline{S} -module.

Proof. By Baer's criterion the noetherian ring \overline{S}_k is self-injective; in particular an \overline{S}_k -module is injective if and only if it is projective. As \overline{S}_k is local, \mathbb{N} is an \overline{S}_k -direct summand of \mathbb{M} if and only if \mathbb{N} is free as \overline{S}_k -module. This is equivalent to \mathbb{N} being free as \overline{S} -module by [EGH13], Remark 3.2.1.

Lemma A.2.2. If $\mathcal{M} \in \operatorname{BrMod}^r$, then $\operatorname{Fil}^r \mathcal{M}/u^e \operatorname{Fil}^r \mathcal{M}$ is a free $k[u]/u^e$ -module of rank $\operatorname{rank}_{\overline{S}_k} \mathcal{M}$ and the map

(A.2.3)
$$\overline{S}_k \otimes_{\varphi, k[u]/u^e} \operatorname{Fil}^r \mathcal{M}/u^e \operatorname{Fil}^r \mathcal{M} \to \mathcal{M}$$

given by $1 \otimes \varphi_r$ is an isomorphism.

Proof. (Cf. [Bre98], Lemme 2.2.1.2 when e = 1.) As \mathcal{M} is finite free over \overline{S}_k and $u^{er}\mathcal{M} \subseteq \operatorname{Fil}^r \mathcal{M}$ the first claim follows. Clearly the map (A.2.3) is well defined and surjective. It is an isomorphism for dimension reasons.

Corollary A.2.4. If $\mathfrak{M} \in \underline{\mathfrak{M}}^r$, let

$$\operatorname{Fil}^{r+1} \mathfrak{M} \stackrel{\text{def}}{=} (\operatorname{Fil}^1 S_{W(k)}) (\operatorname{Fil}^r \mathfrak{M}) + (\operatorname{Fil}^{r+1} S_{W(k)}) \mathfrak{M}.$$

Then the natural map

 $\operatorname{Fil}^{r} \mathcal{M} / \operatorname{Fil}^{r+1} \mathcal{M} \to \operatorname{Fil}^{r} \mathcal{T}(\mathcal{M}) / u^{e} \operatorname{Fil}^{r} \mathcal{T}(\mathcal{M})$

is an isomorphism of $S_{W(k)}/(p, \operatorname{Fil}^1 S_{W(k)}) \cong k[u]/u^e$ -modules, and the map

(A.2.5)
$$S_{W(k)}/p \otimes_{\varphi, k[u]/u^e} \operatorname{Fil}^r \mathcal{M}/\operatorname{Fil}^{r+1} \mathcal{M} \to \mathcal{M}$$

given by $1 \otimes \varphi_r$ is an isomorphism.

Proof. The first claim follows, as $\ker(\mathcal{M} \to \mathcal{T}(\mathcal{M})) = (\operatorname{Fil}^p S_{W(k)})\mathcal{M} \subseteq \operatorname{Fil}^{r+1} \mathcal{M}$. Thus by Lemma A.2.2 both sides of (A.2.5) are free $S_{W(k)}/p$ -modules of the same rank. As $S_{W(k)}$ is local, it suffices to check that (A.2.5) is an isomorphism after the base change $S_{W(k)}/p \twoheadrightarrow \overline{S}_k$, and hence we are done by Lemma A.2.2.

A.3. Functors towards Galois representations. The aim of this section is to briefly recall the constructions of various functors towards Galois representations defined on the categories of §A.1.

The functors towards Galois representations are defined via certain period rings introduced in [Bre99a], §2.2.2 (cf. also [Car06], §2.2 and 2.3). We have a natural surjective morphism $S_{W(k)} \to \overline{S}_k$, as well as $S_{W(k)}$ -algebras \widehat{A}_{st} , A_{cris} , which are endowed with an action of G_K . We have the following commutative diagram:

(A.3.1)
$$\widehat{A}_{st}/p \longrightarrow \widehat{A}_{st} \otimes_{S_{W(k)}} \overline{S}_{k} \stackrel{\text{def}}{=} \widehat{A} \\ \downarrow \qquad \qquad \downarrow \\ A_{cris}/p \longrightarrow A_{cris} \otimes_{S_{W(k)}} \overline{S}_{k} \stackrel{\text{def}}{=} \widehat{A}_{0}$$

which becomes equivariant by endowing the rings¹ \widehat{A} , \widehat{A}_0 with the actions of G_K , $G_{K_{\infty}}$ inherited from \widehat{A}_{st}/p , A_{cris}/p , and \overline{S}_k . The choice of $\underline{\varpi}$ lets us extend these actions to G_{K_0} (respectively $G_{(K_0)_{\infty}}$ in the case of \widehat{A}_0), as explained in [EGH13], §3.1. The horizontal maps in the diagram are then G_{K_0} -equivariant, while the vertical maps are $G_{(K_0)_{\infty}}$ -equivariant.

Let us consider the contravariant functors

$$\underline{\mathcal{M}}^{r} \to \operatorname{Rep}_{\mathbb{F}_{p}}(G_{K}) \qquad \underline{\mathcal{M}}^{r}_{0} \to \operatorname{Rep}_{\mathbb{F}_{p}}(G_{K_{\infty}}) \\
\mathcal{M} \mapsto \operatorname{Hom}(\mathcal{M}, \widehat{A}_{\mathrm{st}}/p) \qquad \mathcal{M}_{0} \mapsto \operatorname{Hom}(\mathcal{M}_{0}, A_{\mathrm{cris}}/p)$$

(where homomorphisms, here and in the following, respect all structures) and their analogues on the categories BrMod^r , BrMod^r_0 , defined by replacing the rings $\widehat{A}_{\mathrm{st}}/p$, A_{cris}/p with \widehat{A} , \widehat{A}_0 respectively. (The functor $\operatorname{Hom}(-, \widehat{A})$ on the category BrMod^r is often denoted by T^*_{st} . This is compatible with the notation T^*_{st} in §2.2 for the functor on the category $\mathbb{F}\operatorname{BrMod}^r_{\mathrm{dd}}$.)

These functors have been extensively studied in [Bre99a], §2.3.1, [Bre99b], §3.2 and §4, [Car06], §2.2 and §2.3 and the morphisms between the various period rings

¹We remark that the rings \widehat{A} , \widehat{A}_0 , A_{cris}/p admit alternative descriptions. We refer the reader to [Car06], §2.2 and 2.3, where it is shown (Lemme 2.3.2) that $A_{\text{cris}}/p \cong R^{DP}$, R^{DP} being a certain period ring introduced in [Bre99b], §3.2, and where the ring \widehat{A} is explicitly described.

(A.3.1) induce natural transformation between the functors toward Galois representations. The following result provides us with a precise description of the relations between the categories and functors introduced above.

Proposition A.3.2. The natural maps (A.3.1) induce a commutative diagram



where the unlabeled morphisms are forgetful functors.

Proof. We may assume that $\mathbb{F} = \mathbb{F}_p$. The left square is commutative by above, and the external square is commutative by [Bre99a], Lemme 2.3.1.1.

We now indicate why the upper and lower triangles are commutative, completing the proof. For the upper triangle this follows from [Car06], Proposition 2.3.4. However, the second paragraph of the published proof needs to be fixed as follows. (We thank Xavier Caruso for the argument.) Given $\overline{\psi} : \mathcal{T}(\mathcal{M}) \to \widehat{A}$, we get an induced map $\overline{\psi} : \operatorname{Fil}^r \mathcal{T}(\mathcal{M}) \to \operatorname{Fil}^r \widehat{A}$. Also, as φ_r kills the kernel of the map $\widehat{A}_{\mathrm{st}}/p \to \widehat{A}$, we get an induced map $\overline{\varphi}_r : \operatorname{Fil}^r \widehat{A} \to \widehat{A}_{\mathrm{st}}/p$. The composite $\overline{\varphi}_r \circ \overline{\psi}$ induces a map $\operatorname{Fil}^r \mathcal{T}(\mathcal{M})/u^e \operatorname{Fil}^r \mathcal{T}(\mathcal{M}) \to \widehat{A}_{\mathrm{st}}/p$ that is linear with respect to the homomorphism $\varphi : k[u]/u^e \to S_{W(k)}/p$. Extending scalars, and using the isomorphisms of Corollary A.2.4, we get an $S_{W(k)}$ -linear map $\psi : \mathcal{M} \to \widehat{A}_{\mathrm{st}}/p$. It is now straightforward to check that ψ lifts $\overline{\psi}$ and that ψ is compatible with all structures. (We note that the results of [Car06] that we cite do not depend on the assumption er made there.)

The lower triangle is commutative by an analogous argument.

Recall the rings $k[[\underline{p}]]$, $k[[\underline{\varpi}]]$ introduced in §A.1. The theory of the field of norms [Win83] lets us endow the ring $k[[\underline{\varpi}]]^s \cong k[[\underline{p}]]^s$ with an action of $G_{(K_0)_{\infty}}$ in such a way that the natural morphism

(A.3.3)
$$k[[p]]^s \to A_{cris}/p$$

becomes $G_{(K_0)_{\infty}}$ -equivariant (cf. [Bre99b], §4.2, where $k[[\underline{p}]]^s$ is noted by \mathfrak{X}^s ; note that we need to take the composite of $\iota: \mathfrak{X}^s \to R$ in *loc. cit.* with the natural map $R \to R^{DP} \cong A_{\text{cris}}/p$).

We have an exact, contravariant functor $\text{Hom}(-, k[[\underline{p}]]^s)$ toward Galois representations (cf. [Bre99b], §2.3, [CL09], §2.1).

Proposition A.3.4. The morphism (A.3.3) and the equivalence $M_{\mathfrak{S}}$ induce a commutative diagram



Proof. By Proposition A.3.2 this follows from [Bre99b], Proposition 4.2.1 (which is stated for e = 1 but whose proof generalizes line by line to the case e > 1).

We consider finally the category of $(\phi, k((\underline{\omega})))$ -modules. By a classical result of Fontaine [Fon90], we have an anti-equivalence

$$\mathfrak{Mod}_{k((\underline{\varpi}))} \xrightarrow{\sim} \operatorname{Rep}_{\mathbb{F}_p}(G_{K_{\infty}})$$
$$\mathfrak{D} \longmapsto \operatorname{Hom}(\mathfrak{D}, k((\underline{p}))^s)$$

and it is an exercise to prove that the composite functor

$$\mathfrak{Mod}_{k((\underline{\varpi})), \mathrm{dd}} \to \mathfrak{Mod}_{k((\underline{\varpi}))} \to \mathrm{Rep}_{\mathbb{F}_p}(G_{K_{\infty}})$$

factors naturally through the restriction $\operatorname{Rep}_{\mathbb{F}_p}(G_{(K_0)_{\infty}}) \to \operatorname{Rep}_{\mathbb{F}_p}(G_{K_{\infty}})$. Indeed, for any $g \in G_{(K_0)_{\infty}}, f \in \operatorname{Hom}(\mathfrak{D}, k((p))^s)$ we define the element

(A.3.5)
$$g \cdot f \stackrel{\text{def}}{=} g \circ f \circ \overline{\overline{g}}^{-1},$$

where $\widehat{\overline{g}}$ denotes the endomorphism of \mathfrak{D} associated to $\overline{g} \in \operatorname{Gal}(K_{\infty}/(K_0)_{\infty})$ via the descent data. It is easy to see that the assignment (A.3.5) is well-defined and endows the former Hom-space with a continuous $G_{(K_0)_{\infty}}$ -action, thus providing us with the claimed factorization of functors.

Lemma A.3.6. The base change along $k((\underline{p})) \hookrightarrow k((\underline{\varpi}))$ induces a commutative diagram of equivalences of categories



Proof. We assume that $\mathbb{F} = \mathbb{F}_p$. For commutativity one just observes that for any $(\phi, k((p)))$ -module \mathfrak{D}_0 , the natural isomorphism

$$\operatorname{Hom}_{(\phi,k((\underline{p})))}(\mathfrak{D}_0,k((\underline{p}))^s) \xrightarrow{\sim} \operatorname{Hom}_{(\phi,k((\underline{\omega})))}(\mathfrak{D}_0 \otimes_{k((\underline{p}))} k((\underline{\omega})),k((\underline{p}))^s)$$

is $G_{(K_0)_{\infty}}$ -equivariant once we endow the right-hand side with the $G_{(K_0)_{\infty}}$ -action induced by the descent data on $\mathfrak{D}_0 \otimes_{k((\underline{p}))} k((\underline{\varpi}))$. The vertical arrow is an equivalence by Galois descent for vector spaces. \Box As in the case of $(\phi, k((\underline{\varpi})))$ -modules, the contravariant functor $\mathfrak{Mod}^r_{k[[\underline{\varpi}]], \mathrm{dd}} \to \mathfrak{Mod}^r_{k[[\underline{\varpi}]]} \to \operatorname{Rep}_{\mathbb{F}_p}(G_{K_{\infty}})$ factors through the restriction $\operatorname{Rep}_{\mathbb{F}_p}(G_{(K_0)_{\infty}}) \to \operatorname{Rep}_{\mathbb{F}_p}(G_{K_{\infty}})$ and we have the following lemma.

Lemma A.3.7. The functor $\operatorname{BrMod}^r_{\mathrm{dd}} \to \operatorname{BrMod}^r_{\mathrm{dd},0} \xrightarrow{\sim} \mathfrak{Mod}^r_{k[[\underline{\varpi}]], \mathrm{dd}}$ fits into a commutative diagram:

$$\begin{array}{c|c} \mathbb{F}\operatorname{-BrMod}^{r}_{\mathrm{dd}} \longrightarrow \mathbb{F}\operatorname{-\mathfrak{Mod}}^{r}_{k[[\underline{\varpi}]], \, \mathrm{dd}} \\ & & \downarrow \\ \Gamma^{*}_{\mathrm{st}} = \operatorname{Hom}(-, \widehat{A}) \downarrow & & \downarrow \\ & & \downarrow \\ \operatorname{Rep}_{\mathbb{F}}(G_{K_{0}}) \xrightarrow{\operatorname{res}} \operatorname{Rep}_{\mathbb{F}}(G_{(K_{0})_{\infty}}). \end{array}$$

Proof. As usual we prove the result when $\mathbb{F} = \mathbb{F}_p$. Combining Propositions A.3.2 and A.3.4 we have a commutative diagram (the arrows being the evident ones):

where the natural functorial isomorphism on the right side

(A.3.8)
$$\operatorname{Hom}(M_{\mathfrak{S}}(-), \widehat{A}_0) \xrightarrow{\sim} \operatorname{Hom}(-, k[[p]]^s)$$

is induced by the morphisms between the period rings described by (A.3.1), (A.3.3). By using twists as in the proof of Lemma A.1.1 and the fact that the morphisms between period rings are $G_{(K_0)_{\infty}}$ -equivariant, it follows that (A.3.8) is compatible with the $G_{(K_0)_{\infty}}$ -action induced by the descent data assignment (A.3.5). An analogous argument applies to the left side of the diagram.

In a similar fashion, we obtain the following.

Lemma A.3.9. The localization functor \mathbb{F} - $\mathfrak{Mod}^r_{k[[\underline{\varpi}]], \mathrm{dd}} \to \mathbb{F}$ - $\mathfrak{Mod}_{k((\underline{\varpi})), \mathrm{dd}}$ induces a commutative diagram



Proof. Without descent data this is [Bre99b], Lemme 2.3.3 (the latter being stated for e = 1 but its proof generalizes). The statement with descent data follows now as in the proof of Lemma A.3.7 (noting that $k[[\underline{p}]]^s \to k((\underline{p}))^s$ is obviously $G_{(K_0)_{\infty}}$ -equivariant).

66

A.4. On dualities. We recall that we have a notion of duality on the category \mathbb{F} -BrMod^r_{dd}; more precisely, there is an involutive contravariant functor

$$\begin{aligned} \mathbb{F}\text{-}\mathrm{BrMod}^r_{\mathrm{dd}} \to \mathbb{F}\text{-}\mathrm{BrMod}^r_{\mathrm{dd}} \\ \mathcal{M} \mapsto \mathcal{M}^* \end{aligned}$$

such that $T_{st}^r(\mathcal{M}) \cong T_{st}^*(\mathcal{M}^*)$. For details, see [EGH13], Definition 3.2.8 (building on work of Caruso [Car05], Chapitre V).

A.5. Proofs of some results in Section 2.2.

Proof of Proposition 2.2.1. As usual we may assume that $\mathbb{F} = \mathbb{F}_p$. By Lemmas A.3.6, A.3.7, and A.3.9 the commutativity of the upper square and of the right triangle in the statement of Proposition 2.2.1 is clear. Concerning the lower square, we consider the fully faithful functor $\mathcal{F}_{p-2} : \mathcal{FL}^{[0,p-2]} \to \operatorname{BrMod}^{p-2}$ defined in [Bre98], §2.4, which extends the functor of the same name considered in §A.1. On the other hand, we have the functor $M_{k(\underline{p})} : \operatorname{BrMod}^{p-2} \to \mathfrak{Mod}_{k(\underline{p})}$ obtained by specializing the functor $M_{k(\underline{m})}$ of §A.1 to the case when e = 1. The diagram



commutes: for the triangle, see [Bre98], Proposition 3.2.1.1, whereas the square is the specialization of the upper square of the statement of Proposition 2.2.1 to the case when e = 1. The composite morphism $\mathcal{FL}^{[0,p-2]} \to \operatorname{Rep}_{\mathbb{F}_p}(G_{(K_0)_{\infty}})$ is fully faithful, as the functor \mathcal{F} in §A.1 has this property. \Box

Proof of Lemma 2.2.4. Lemma A.2.2 shows that φ_r induces a $\operatorname{Gal}(K/K_0)$ -equivariant and $\varphi \otimes 1$ -semilinear isomorphism $\operatorname{Fil}^r \mathcal{M}/u \operatorname{Fil}^r \mathcal{M} \xrightarrow{\sim} \mathcal{M}/u \mathcal{M}$. We conclude by Nakayama's lemma and the semisimplicity of $(k \otimes_{\mathbb{F}_p} \mathbb{F})[\operatorname{Gal}(K/K_0)]$. \Box

Proof of Lemma 2.2.6. By passing to the new basis $\underline{e} \cdot A$ we may assume, without loss of generality, that A = 1. As $u^{er} \mathcal{M} \subseteq \operatorname{Fil}^r \mathcal{M}$ we deduce that $u^{er} = VV'$ in $\mathcal{M}_n(\overline{S})$ for some $V' \in \mathcal{M}_n(\overline{S})$. Letting $\widehat{V} \in \mathcal{M}_n(k \otimes_{\mathbb{F}_p} \mathbb{F}[[\underline{\varpi}]])$ denote any lift of V such that $\widehat{V}_{ij} \in (k \otimes_{\mathbb{F}_p} \mathbb{F}[[\underline{\varpi}]])_{\omega_{\overline{\omega}}^{p^{-1}a_j-a_i}}$ we see that $\underline{\varpi}^{er} = \widehat{V}\widehat{W} = \widehat{W}\widehat{V}$ for some $\widehat{W} \in \mathcal{M}_n(k \otimes_{\mathbb{F}_p} \mathbb{F}[[\underline{\varpi}]])$. Let \underline{e}^* denote the basis of \mathcal{M}^* that is dual to \underline{e} , and let $\underline{f}^* \stackrel{\text{def}}{=} \underline{e}^* \cdot {}^t W$, where $W \in \mathcal{M}_n(\overline{S})$ denotes the reduction of W. One easily computes that $f_i^* \in \operatorname{Fil}^r \mathcal{M}$ and $\varphi_r(f_i^*) = e_i^*$ for all i. As φ_r induces an isomorphism $\operatorname{Fil}^r \mathcal{M}^*/u \operatorname{Fil}^r \mathcal{M}^* \xrightarrow{\sim} \mathcal{M}^*/u \mathcal{M}^*$ (cf. the proof of Lemma 2.2.4), it follows that \underline{f}^* generates $\operatorname{Fil}^r \mathcal{M}$ as \overline{S} -module. Also, $\widehat{g}e_i^* = (\omega_{\overline{\omega}}^{-a_i}(g) \otimes 1)e_i^*$. Define $\mathfrak{M} \in \mathbb{F}\operatorname{-}\mathfrak{Mool}_k^r[[\underline{\omega}]], \mathrm{dof}$ rank n by giving it basis $\underline{\mathfrak{e}}$ and defining the maps

Define $\mathfrak{M} \in \mathbb{F}\operatorname{-}\mathfrak{Mod}_{k[[\underline{\varpi}]], \operatorname{dd}}^{r}$ of rank *n* by giving it basis $\underline{\mathfrak{e}}$ and defining the maps $\phi, \, \widehat{g}$ by $\operatorname{Mat}_{\underline{\mathfrak{e}}}(\phi) = {}^{t}\widehat{V}$ and $\widehat{g}\mathfrak{e}_{i} = (\omega_{\overline{\varpi}}^{-p^{-1}a_{i}}(g) \otimes 1)\mathfrak{e}_{i}$. (It is easy to check that \mathfrak{M} is of height $\leq r$, and the condition on \widehat{V}_{ij} above implies that $\widehat{g} \circ \phi = \phi \circ \widehat{g}$.)

We can compute $\mathcal{M}' \stackrel{\text{def}}{=} M_{\mathfrak{S}}(\mathfrak{M})$ without coefficients and descent data by the recipe in [CL09], §2.1, so $\mathcal{M}' \cong \overline{S}_k \otimes_{\varphi, k[[\underline{\varpi}]]} \mathfrak{M}$. It has \overline{S} -basis $\underline{e}' \stackrel{\text{def}}{=} 1 \otimes_{\varphi} \underline{\mathfrak{c}}$. By functoriality, \mathcal{M}' has an \mathbb{F} -action with $\lambda \in \mathbb{F}$ acting as $1 \otimes \lambda$. By the twisting argument of the proof of Lemma A.1.1, \mathcal{M}' has descent data with $g \in \text{Gal}(K/K_0)$

acting as $g \otimes \widehat{g}$. Letting $\underline{f}' \stackrel{\text{def}}{=} \underline{e}' \cdot {}^t W$, it is easy to check that $f'_i \in \operatorname{Fil}^r \mathcal{M}'$ and $\varphi_r(f'_i) = e'_i$ for all *i*. In particular, as above, we deduce that \underline{f}' generates $\operatorname{Fil}^r \mathcal{M}'$ as \overline{S} -module. Also, $\widehat{g}e'_i = (\omega_{\varpi}^{-a_i}(g) \otimes 1)e'_i$. Hence $\mathcal{M}' \cong \mathcal{M}^*$ in \mathbb{F} -BrMod $^r_{\mathrm{dd},0}$. \Box

Proof of Lemma 2.2.7. In the following e = 1, so $\overline{S}_k = k[u]/u^p$. Note that $\mathbb{N} \stackrel{\text{def}}{=} \mathcal{F}_{p-2}(M) \in \mathbb{F}\text{-BrMod}_0^{p-2}$ has \overline{S} -basis \underline{e} (or more precisely $1 \otimes \underline{e} \in \overline{S}_k \otimes_k M$). A computation shows that $\operatorname{Fil}^{p-2} \mathbb{N}$ has generating set \underline{f} with $\operatorname{Mat}_{\underline{e},\underline{f}}(\operatorname{Fil}^{p-2} \mathbb{N}) = \operatorname{Diag}(u^{p-2-m_0}, \ldots, u^{p-2-m_{n-1}})$ and $\operatorname{Mat}_{\underline{e},f}(\varphi_{p-2}) = F$.

We now apply Lemma 2.2.6 with e = 1 and $\mathcal{M} = \mathcal{N}^*$, noting that $M_{k((\underline{p}))}(\mathcal{N}) = \mathcal{F}(M)$ in this case.

Proof of Lemma 2.2.8. This is elementary. Indeed, by (2.2.9), one has

$$(V + u^{e(r+1)}M)B = AV$$

for some element $M \in M_n(\overline{S})$. By considering descent data, we see that $\underline{f}_1 \stackrel{\text{def}}{=} \underline{e} \cdot (V + u^{e(r+1)}M)$ is a framed generating family of Fil^r \mathcal{M} and one has

$$\operatorname{Mat}_{\underline{e},f_{+}}(\varphi_{r}) = \operatorname{Mat}_{\underline{e},f}(\varphi_{r}) = A.$$

Note that $\underline{f}' = \underline{e}' \cdot V' = \underline{f}_1 \cdot B$. As $B \in \operatorname{GL}_n(\overline{S})$ and by considering descent data, we see that \underline{f}' is a framed system of generators for $\operatorname{Fil}^r \mathcal{M}$. The last statement follows from $\varphi_r(\underline{f}') = \varphi_r(\underline{f}_1) \cdot \varphi(B) = \underline{e} \cdot A\varphi(B) = \underline{e}' \cdot \varphi(B)$.

A.6. **Proofs of some results in Section 2.3.** Recall that in Definition 2.3.1 we defined the notion of a Breuil submodule.

Proof of Lemma 2.3.2. We note that \mathbb{N} is a finite free \overline{S} -module by Lemma A.2.1. Let $\operatorname{Fil}^r \mathbb{N} \stackrel{\text{def}}{=} \operatorname{Fil}^r \mathfrak{M} \cap \mathbb{N}$, so $u^{er} \mathbb{N} \subseteq \operatorname{Fil}^r \mathbb{N}$. The map $\operatorname{Fil}^r \mathbb{N}/u^e \operatorname{Fil}^r \mathbb{N} \to \operatorname{Fil}^r \mathfrak{M}/u^e \operatorname{Fil}^r \mathfrak{M}$ is injective (as \mathbb{N} is an \overline{S}_k -direct summand). It is in fact a split injection of $k[u]/u^e$ modules by Lemma A.2.2, as $k[u]/u^e$ is self-injective. Thus, using again the same lemma, we have the following commutative diagram:

It follows that the left vertical map is injective, hence surjective by dimension reasons. It is now obvious that \mathcal{N} inherits from \mathcal{M} the structure of a Breuil module with descent data.

By defining Fil^r $(\mathcal{M}/\mathcal{N}) \stackrel{\text{def}}{=} (\text{Fil}^r \mathcal{M} + \mathcal{N})/\mathcal{N} \cong \text{Fil}^r \mathcal{M}/\text{Fil}^r \mathcal{N}$ we see that the Frobenius φ_r on Fil^r \mathcal{M} naturally induces a semilinear morphism $\varphi_r : \text{Fil}^r (\mathcal{M}/\mathcal{N}) \to \mathcal{M}/\mathcal{N}$ and it is immediate that the triple $(\mathcal{M}/\mathcal{N}, \text{Fil}^r (\mathcal{M}/\mathcal{N}), \varphi_r)$ inherits the structure of a Breuil module with descent data.

By construction, the complex $0 \to \mathbb{N} \to \mathbb{M} \to \mathbb{M}/\mathbb{N} \to 0$ is an exact sequence in \mathbb{F} -BrMod^r_{dd}. The last statement in the lemma is obvious.

Recall that given an exact sequence

$$0 \to \mathcal{M}_1 \xrightarrow{J} \mathcal{M} \xrightarrow{g} \mathcal{M}_2 \to 0$$

in \mathbb{F} -BrMod^r_{dd}, the morphisms f and g are respectively said to be an *admissible* monomorphism and an *admissible epimorphism*.

Lemma A.6.1. Let $f : \mathcal{M} \to \mathcal{N}$ be a morphism in \mathbb{F} -BrMod^r_{dd}. Then:

- (i) f is an admissible epimorphism if and only if f induces S̄-linear surjections M → N and Fil^r M → Fil^r N;
- (ii) f is an admissible monomorphism if and only if f induces an S-linear injection M → N and Fil^r M = f⁻¹(Fil^r N).

Proof. The "only if" part, both in (i) and (ii), is obvious. Let us assume $f : \mathcal{M} \to \mathcal{N}$ induces \overline{S} -linear surjections $\mathcal{M} \to \mathcal{N}$ and $\operatorname{Fil}^r \mathcal{M} \to \operatorname{Fil}^r \mathcal{N}$. Then $\operatorname{ker}(f)$ is clearly a Breuil submodule of \mathcal{M} ; by Lemma 2.3.2 it is an object in \mathbb{F} -BrMod^r_{dd} and the complex $0 \to \operatorname{ker}(f) \to \mathcal{M} \to \mathcal{N} \to 0$ is an exact sequence in \mathbb{F} -BrMod^r_{dd}.

Assume now that f induces an \overline{S} -linear injection $\mathcal{M} \hookrightarrow \mathcal{N}$ and that $\operatorname{Fil}^r \mathcal{M} = f^{-1}(\operatorname{Fil}^r \mathcal{N})$. Then, using Lemma A.2.1 one sees that $f(\mathcal{M})$ is a Breuil submodule of \mathcal{N} . As the map $f : \mathcal{M} \to f(\mathcal{M})$ is an isomorphism in $\mathbb{F}\operatorname{-BrMod}^r_{\mathrm{dd}}$, Lemma 2.3.2 implies that f is an admissible monomorphism, as required.

Recall from §A.4 that we have a duality on the category \mathbb{F} -BrMod^r_{dd}.

Lemma A.6.2. The functor $\mathcal{M} \mapsto \mathcal{M}^*$ preserves exact sequences in \mathbb{F} -BrMod^{*r*}_{dd}.

Proof. By definition of $\operatorname{Fil}^r(\mathfrak{M}^*)$ and the self-injectivity of \overline{S}/u^{er} (cf. Lemma A.2.1) one has an exact sequence of \overline{S} -modules

$$0 \to \operatorname{Fil}^{r}(\mathcal{M}^{*}) \to \mathcal{M}^{*} \to \operatorname{Hom}_{\overline{S}/u^{er}}(\operatorname{Fil}^{r} \mathcal{M}/u^{er} \mathcal{M}, \overline{S}/u^{er}) \to 0,$$

which is functorial in \mathcal{M} . In order to prove the lemma it is enough to prove that the functor $\mathcal{M} \mapsto \operatorname{Fil}^r(\mathcal{M}^*)$ is exact (in \overline{S} -modules) which is in turn equivalent, thanks to the exact sequence above and the self-injectivity of \overline{S}/u^{er} , to the exactness of the functor $\mathcal{M} \mapsto \operatorname{Fil}^r \mathcal{M}/u^{er} \mathcal{M}$ (in \overline{S} -modules). But the exactness of $\mathcal{M} \mapsto \operatorname{Fil}^r \mathcal{M}/u^{er} \mathcal{M}$ is an easy consequence of the snake lemma.

We are now finally in a position to prove Propositions 2.3.4 and 2.3.5.

Proof of Proposition 2.3.4. Clearly \mathbb{F} -BrMod^r_{dd} is an additive category and the class of exact sequences is closed under isomorphism. Moreover, given an exact sequence

$$0 \to \mathcal{M}_1 \xrightarrow{f} \mathcal{M} \xrightarrow{g} \mathcal{M}_2 \to 0$$

in \mathbb{F} -BrMod^r_{dd} it is clear that f is the categorical kernel of g. By duality, g is the categorical cokernel of f.

We shall now check the axioms Ex0, Ex1, Ex2, $Ex2^{op}$ of [Kel90], Appendix A.

- (Ex0): It is obvious as $0 \rightarrow 0$ is an admissible epimorphism.
- (Ex1): By Lemma A.6.1 a composition of admissible epimorphisms is an admissible epimorphism.
- (Ex2): Let $\alpha : \mathcal{M} \to \mathcal{M}_2$ be an admissible epimorphism and $\beta : \mathcal{N} \to \mathcal{M}_2$ a morphism in \mathbb{F} -BrMod^r_{dd}. By Lemma A.6.1 the map $\mathcal{M} \oplus \mathcal{N} \to \mathcal{M}_2$, $(x, y) \mapsto \alpha(x) \beta(y)$ is an admissible epimorphism, so its kernel \mathcal{P} is a Breuil submodule of $\mathcal{M} \oplus \mathcal{N}$. The same lemma shows moreover that the natural map $\mathcal{P} \to \mathcal{N}$ is an admissible epimorphism. It is then clear that \mathcal{P} is a categorical pullback of (α, β) .
- (Ex2^{op}): It follows from (Ex2) via the duality functor on \mathbb{F} -BrMod^r_{dd}.

It remains to prove the exactness of the functor T_{st}^r . Let

(A.6.3)
$$0 \to \mathcal{M}_1 \xrightarrow{f} \mathcal{M} \xrightarrow{g} \mathcal{M}_2 \to 0$$

be an exact sequence in \mathbb{F} -BrMod^{*t*}_{dd}. Then, by forgetting monodromy, descent data and coefficients, we deduce that the exact sequence (A.6.3) arises from a sequence $0 \to \mathfrak{M}_1 \to \mathfrak{M} \to \mathfrak{M}_2 \to 0$ in $\mathfrak{Mod}^r_{k[[\underline{\varpi}]]}$ via the equivalence $M_{\mathfrak{S}}$. By [CL09], Proposition 2.3.2 the latter sequence is exact. By Proposition A.3.4 and the exactness of the functor $\operatorname{Hom}(-, k[[\underline{p}]]^s)$ we deduce that $0 \to \operatorname{T}^r_{\mathrm{st}}(\mathfrak{M}_1) \to \operatorname{T}^r_{\mathrm{st}}(\mathfrak{M}) \to$ $\operatorname{T}^r_{\mathrm{st}}(\mathfrak{M}_2) \to 0$ is an exact sequence in $\operatorname{Rep}_{\mathbb{F}_p}(G_{K_{\infty}})$.

Proof of Proposition 2.3.5. Let $T' \subseteq T_{st}^r(\mathcal{M}) \cong T_{st}^*(\mathcal{M}^*)$. By the proof of [EGH13], Proposition 3.2.6 we get a unique (up to isomorphism) surjective homomorphism $f: \mathcal{M}^* \to \mathcal{N}'$ in F-BrMod^{*r*}_{dd} such that $\operatorname{im}(T_{st}^*(f)) = T'$. We claim that f is an admissible epimorphism. We recall from [EGH13], proof of Proposition 3.2.6 and [Car11], proof of Proposition 2.2.5 that f is obtained from a surjective homomorphism $\mathfrak{f}: \mathfrak{M}^* \to \mathfrak{N}'$ in $\mathfrak{Mod}_{k[[\varpi]]}^r$ via the equivalence $M_{\mathfrak{S}}$, and that we have an exact sequence in $\mathfrak{Mod}_{k[[\varpi]]}^r$ given by $0 \to \ker(\mathfrak{f}) \to \mathfrak{M}^* \to \mathfrak{N}' \to 0$. By the exactness of $M_{\mathfrak{S}}$ we deduce that f is indeed an admissible epimorphism. By Lemma A.6.2 the image of $(\mathcal{N}')^*$ inside \mathcal{M} is the desired Breuil submodule in \mathcal{M} mapping to T' via Θ . It is unique by construction.

We still need to check that Θ is order preserving. Suppose that \mathcal{M}_2 , \mathcal{M}_1 are Breuil submodules of \mathcal{M} such that $\mathcal{M}_2 \subseteq \mathcal{M}_1$. Then by Lemma 2.3.3 we have $\Theta(\mathcal{M}_2) \subseteq \Theta(\mathcal{M}_1)$. Moreover we have a natural isomorphism $\Theta(\mathcal{M}_1)/\Theta(\mathcal{M}_2) \cong T_{st}^r(\mathcal{M}_1/\mathcal{M}_2)$ by the exactness of T_{st}^r . Conversely, given $T_2 \subseteq T_1 \subseteq T_{st}^r(\mathcal{M})$ then, by the above, we have $T_1 = \operatorname{im}(T_{st}^r(f_1))$ for some admissible monomorphism $f_1 : \mathcal{M}_1 \hookrightarrow \mathcal{M}$ and, similarly, $T_2 = \operatorname{im}(T_{st}^r(f_2))$ for some admissible monomorphism $f_2 : \mathcal{M}_2 \hookrightarrow \mathcal{M}_1$. Then $f_1 \circ f_2$ is an admissible monomorphism and, by uniqueness, we obtain finally

$$\Theta^{-1}(T_2) = \operatorname{im}(f_1 \circ f_2) \subseteq \operatorname{im}(f_1) = \Theta^{-1}(T_1).$$

References

- [And87] Henning Haahr Andersen, Extensions of simple modules for finite Chevalley groups, J. Algebra 111 (1987), no. 2, 388–403. MR 916175 (89b:20089)
- [BC09] Joël Bellaïche and Gaëtan Chenevier, Families of Galois representations and Selmer groups, Astérisque (2009), no. 324, xii+314. MR 2656025 (2011m:11105)
- [BD14] Christophe Breuil and Fred Diamond, Formes modulaires de Hilbert modulo p et valeurs d'extensions entre caractères galoisiens, Ann. Sci. Éc. Norm. Supér. (4) 47 (2014), no. 5, 905–974. MR 3294620
- [BH15] Christophe Breuil and Florian Herzig, Ordinary representations of $G(\mathbb{Q}_p)$ and fundamental algebraic representations, Duke Math. J. **164** (2015), no. 7, 1271–1352. MR 3347316
- [BLGG12] Thomas Barnet-Lamb, Toby Gee, and David Geraghty, Congruences between Hilbert modular forms: constructing ordinary lifts, Duke Math. J. 161 (2012), no. 8, 1521– 1580. MR 2931274
- [BLGGT14] Thomas Barnet-Lamb, Toby Gee, David Geraghty, and Richard Taylor, Potential automorphy and change of weight, Ann. of Math. (2) 179 (2014), no. 2, 501–609. MR 3152941
- [BP12] Christophe Breuil and Vytautas Paškūnas, Towards a modulo p Langlands correspondence for GL₂, Mem. Amer. Math. Soc. **216** (2012), no. 1016, vi+114. MR 2931521
- [Bre97] C. Breuil, Représentations p-adiques semi-stables et transversalité de Griffiths, Math. Ann. 307 (1997), no. 2, 191–224. MR 1428871 (98b:14016)
- [Bre98] _____, Construction de représentations p-adiques semi-stables, Ann. Sci. École Norm. Sup. (4) 31 (1998), no. 3, 281–327. MR 1621389 (99k:14034)

[Bre99a] , Représentations semi-stables et modules fortement divisibles, Invent. Math. 136 (1999), no. 1, 89-122. MR 1681105 (2000c:14024) [Bre99b] Christophe Breuil, Une application de corps des normes, Compositio Math. 117 (1999), no. 2, 189-203. MR 1695849 (2000f:11157) [Bre03] C. Breuil, Sur quelques représentations modulaires et p-adiques de $GL_2(\mathbf{Q}_p)$. I, Compositio Math. 138 (2003), no. 2, 165–188. MR 2018825 (2004k:11062) [Bre11] _, Diagrammes de Diamond et (ϕ, Γ) -modules, Israel J. Math. **182** (2011), 349-382. MR 2783977 (2012f:22026) [Car05]Xavier Caruso, Conjecture de l'inertie modérée de Serre, Ph.D. thesis, Univ. Paris 13, available at http://tel.archives-ouvertes.fr/tel-00011202, 2005. [Car06], Représentations semi-stables de torsion dans le case er , J. ReineAngew. Math. 594 (2006), 35-92. MR 2248152 (2007d:11061) [Car11] X. Caruso, F_p -représentations semi-stables, Ann. Inst. Fourier (Grenoble) **61** (2011), no. 4, 1683-1747 (2012). MR 2951749 [CDT99] Brian Conrad, Fred Diamond, and Richard Taylor, Modularity of certain potentially Barsotti-Tate Galois representations, J. Amer. Math. Soc. 12 (1999), no. 2, 521–567. MR 1639612 (99i:11037) $[CEG^+16]$ Ana Caraiani, Matthew Emerton, Toby Gee, David Geraghty, Vytautas Paškūnas, and Sug Woo Shin, Patching and the p-adic local Langlands correspondence, Camb. J. Math. 4 (2016), no. 2, 197-287. MR 3529394 [CHT08] Laurent Clozel, Michael Harris, and Richard Taylor, Automorphy for some l-adic lifts of automorphic mod l Galois representations, Publ. Math. Inst. Hautes Études Sci. (2008), no. 108, 1-181. MR 2470687 (2010j:11082) [CL09] Xavier Caruso and Tong Liu, Quasi-semi-stable representations, Bull. Soc. Math. France 137 (2009), no. 2, 185-223. MR 2543474 (2011c:11086) [Col10] P. Colmez, Représentations de $GL_2(\mathbf{Q}_p)$ et (ϕ, Γ) -modules, Astérisque (2010), no. 330, 281-509. MR 2642409 (2011j:11224) [Dia97] Fred Diamond, The Taylor-Wiles construction and multiplicity one, Invent. Math. 128 (1997), no. 2, 379-391. MR 1440309 (98c:11047) [EG14] Matthew Emerton and Toby Gee, A geometric perspective on the Breuil-Mézard conjecture, J. Inst. Math. Jussieu 13 (2014), no. 1, 183-223. MR 3134019 [EGH13] M. Emerton, T. Gee, and F. Herzig, Weight cycling and Serre-type conjectures for unitary groups, Duke Math. J. 162 (2013), no. 9, 1649-1722. MR 3079258 [EGS15] Matthew Emerton, Toby Gee, and David Savitt, Lattices in the cohomology of Shimura curves, Invent. Math. 200 (2015), no. 1, 1-96. MR 3323575 M. Emerton, Local-global compatibility in the p-adic langlands programm for $\operatorname{GL}_{2/\mathbb{Q}}$, [Eme] http://www.math.uchicago.edu/~emerton/pdffiles/lg.pdf, preprint, 2011. [FL82] Jean-Marc Fontaine and Guy Laffaille, Construction de représentations p-adiques, Ann. Sci. École Norm. Sup. (4) 15 (1982), no. 4, 547-608 (1983). MR 707328 (85c:14028) [Fon90] Jean-Marc Fontaine, Représentations p-adiques des corps locaux. I, The Grothendieck Festschrift, Vol. II, Progr. Math., vol. 87, Birkhäuser Boston, Boston, MA, 1990, pp. 249-309. MR 1106901 (92i:11125) [GG12] Toby Gee and David Geraghty, Companion forms for unitary and symplectic groups, Duke Math. J. 161 (2012), no. 2, 247-303. MR 2876931 [GHS] Toby Gee, Florian Herzig, and David Savitt, General Serre weight conjectures, arXiv:1509.02527v2, 2016. [GL14] Hui Gao and Tong Liu, A note on potential diagonalizability of crystalline representations, Math. Ann. 360 (2014), no. 1-2, 481-487. MR 3263170 [Hen02] Guy Henniart, Une caractérisation de la correspondance de Langlands locale pour GL(n), Bull. Soc. Math. France **130** (2002), no. 4, 587–602. MR 1947454 (2004d:22013) Florian Herzig, The weight in a Serre-type conjecture for tame n-dimensional Galois [Her09] representations, Duke Math. J. 149 (2009), no. 1, 37-116. MR 2541127 (2010f:11083) [Her11] F. Herzig, The classification of irreducible admissible mod p representations of a

- [HT01] Michael Harris and Richard Taylor, The geometry and cohomology of some simple Shimura varieties, Annals of Mathematics Studies, vol. 151, Princeton University Press, Princeton, NJ, 2001, With an appendix by Vladimir G. Berkovich. MR 1876802 (2002m:11050)
- [Hum06] James E. Humphreys, Modular representations of finite groups of Lie type, London Mathematical Society Lecture Note Series, vol. 326, Cambridge University Press, Cambridge, 2006. MR MR2199819 (2007f:20023)
- [Jan03] Jens Carsten Jantzen, Representations of algebraic groups, second ed., Mathematical Surveys and Monographs, vol. 107, American Mathematical Society, Providence, RI, 2003. MR 2015057 (2004h:20061)
- [Kel90] Bernhard Keller, Chain complexes and stable categories, Manuscripta Math. 67 (1990), no. 4, 379-417. MR 1052551 (91h:18006)
- [Le] D. Le, Lattices in the cohomology of U(3) arithmetic manifolds, arXiv:1507.04766, 2015.
- [Liu08] T. Liu, On lattices in semi-stable representations: a proof of a conjecture of Breuil, Compos. Math. 144 (2008), no. 1, 61-88. MR 2388556 (2009c:14087)
- [LLHLM] Daniel Le, Bao V. Le Hung, Brandon Levin, and Stefano Morra, Potentially crystalline deformation rings and Serre weight conjectures (Shapes and Shadows), arXiv:1512.06380, 2015.
- [MP]S. Morra and C. Park, Serre weights for three dimensional ordinary Galois representations, preprint, 2014.
- [Pas07] V. Paskunas, On the restriction of representations of $GL_2(F)$ to a Borel subgroup, Compos. Math. 143 (2007), no. 6, 1533-1544. MR 2371380 (2009a:22013)
- [Roc98] Alan Roche, Types and Hecke algebras for principal series representations of split reductive p-adic groups, Ann. Sci. École Norm. Sup. (4) 31 (1998), no. 3, 361-413. MR 1621409 (99d:22028)
- Richard Taylor, Automorphy for some l-adic lifts of automorphic mod l Galois rep-[Tay08] resentations. II, Publ. Math. Inst. Hautes Études Sci. (2008), no. 108, 183–239. MR 2470688 (2010j:11085)
- [Tho12] Jack Thorne, On the automorphy of l-adic Galois representations with small residual image, J. Inst. Math. Jussieu 11 (2012), no. 4, 855-920, With an appendix by Robert Guralnick, Florian Herzig, Richard Taylor and Thorne. MR 2979825
- [Win83] Jean-Pierre Wintenberger, Le corps des normes de certaines extensions infinies de corps locaux; applications, Ann. Sci. École Norm. Sup. (4) 16 (1983), no. 1, 59–89. MR 719763 (85e:11098)

DEPARTMENT OF MATHEMATICS, UNIVERSITY OF TORONTO, 40 ST. GEORGE STREET, TORONTO, ON M5S 2E4, CANADA

E-mail address: herzig@math.toronto.edu

DEPARTMENT OF MATHEMATICS, UNIVERSITY OF TORONTO, 40 ST. GEORGE STREET, TORONTO, ON M5S 2E4, CANADA

E-mail address: le@math.toronto.edu

INSTITUT MONTPELLIÉRAIN ALEXANDER GROTHENDIECK, UNIVERSITÉ DE MONTPELLIER, CASE COURRIER 051, PLACE EUGÈNE BATAILLON, 34095 MONTPELLIER CEDEX, FRANCE

E-mail address: stefano.morra@umontpellier.fr